Court File No. CV-13-10383-00CL

ONTARIO SUPERIOR COURT OF JUSTICE COMMERCIAL LIST

IN THE MATTER OF THE COMPANIES' CREDITORS ARRANGEMENT ACT, R.S.C. 1985, c. C-36, AS AMENDED

AND IN THE MATTER OF A PLAN OF COMPROMISE OR ARRANGEMENT OF JAGUAR MINING INC.

Applicant

MOTION RECORD (Re: Stay Extension) (returnable February 27, 2014)

February 27, 2014

Norton Rose Fulbright Canada LLP

Royal Bank Plaza, South Tower Suite 3800 200 Bay Street, P.O. Box 84 Toronto, Ontario M5J 2Z4 CANADA

Tony Reyes LS#: 28218V Tel: 416.216.4825 Email: tony.reyes@nortonrosefulbright.com

Evan Cobb LS#: 55787N Tel: 416.216.1929 Email: evan.cobb@nortonrosefulbright.com

Fax: 416.216.3930

Lawyers for the Applicant, Jaguar Mining Inc.

Court File No. CV-13-10383-00CL

natur balan Bara di Bara da Baratan Batan Batan Batan Batan da Kabupa

ONTARIO SUPERIOR COURT OF JUSTICE COMMERCIAL LIST

동네는 동안은 동안을 했다.

2 S S I -

11.1.1.1.1.1.1.1.

IN THE MATTER OF THE COMPANIES' CREDITORS ARRANGEMENT ACT, R.S.C. 1985, c. C-36, AS AMENDED

AND IN THE MATTER OF A PLAN OF COMPROMISE OR ARRANGEMENT OF JAGUAR MINING INC.

Applicant

INDEX

Tab:	Document:	Page Number:
1	Notice of Motion returnable February 27, 2014	1-10
2	Affidavit of T. Douglas Willock sworn on February 27, 2014 along with Exhibits	11-156
3	Draft form of Order	157-159

Tab 1

n fa fara an in 15 fan 'n die nationale ar staar gebruik gebruik.

searchean an the search of a second second second

ALTER CALLED STORE CONTRACTOR

Court File No. CV-13-10383-00CL

ONTARIO SUPERIOR COURT OF JUSTICE COMMERCIAL LIST

IN THE MATTER OF THE COMPANIES' CREDITORS ARRANGEMENT ACT, R.S.C. 1985, c. C-36, AS AMENDED

AND IN THE MATTER OF A PLAN OF COMPROMISE OR ARRANGEMENT OF JAGUAR MINING INC.

Applicant

NOTICE OF MOTION (Stay Extension) (Returnable: February 27, 2014)

JAGUAR MINING INC. (the "**Applicant**") will make a motion to a judge presiding over the Commercial List on Thursday, February 27, 2014, at 3:00 p.m., or as soon after that time as the motion can be heard, at 330 University Avenue, Toronto, Ontario.

PROPOSED METHOD OF HEARING: The motion is to be heard orally.

THE MOTION IS FOR:

- 1. An Order in the form of a draft order included in the Applicant's Motion Record that, among other things:
 - (a) abridges and validates the time for service of the Notice of Motion and Motion Record in respect of this motion and the Fourth Report of the Monitor, FTI Consulting Canada Inc. (the "Monitor"), to be served (the "Fourth Report"), and dispenses with further service thereof;
 - (b) extends the Stay Period, as such term is defined in the initial order granted by this Court on December 23, 2013 (as amended, the "**Initial Order**"), up to and including March 10, 2014;

- (c) approves certain amendments to the Applicant's Amended and Restated Plan of Compromise and Arrangement, dated February 5, 2014 (as may be further amended, the "Plan"); and
- (d) grants such other relief as counsel for the Applicant may request and this Court deems fit.

2. THE GROUNDS FOR THE MOTION ARE:

- (a) since the issuance of the Initial Order, the Applicant has continued to work in good faith and with due diligence to implement its Plan;
- (b) the Plan was approved by 100% of affected creditors voting on the Plan;
- (c) the Plan was approved by an order of this Court granted on February 6, 2014 (the "**Sanction Order**");
- (d) the Applicant cannot be certain that all conditions to implementation of the Plan will be satisfied by February 28, 2014, the date on which the Stay Period currently expires;
- (e) the Applicant has sufficient resources to meet its post-filing obligations during the extension of the Stay Period;
- (f) the Monitor and key stakeholders support the requested extension of the Stay Period;
- (g) the amendments to the Plan are purely administrative in nature and are required to better give effect to the implementation of the Plan and/or to cure any errors, omissions or ambiguities and are not materially adverse to the financial or economic interests of the affected unsecured creditors under the Plan; and
- (h) such further and other grounds as counsel for the Applicant may advise and this Court may permit.

THE FOLLOWING DOCUMENTARY EVIDENCE will be used at the hearing of the motion:

- (a) the Affidavit of T. Douglas Willock, sworn February 27, 2014, including the exhibits thereto;
- (b) the Fourth Report, to be filed; and
- (c) such further and other material as counsel for the Applicant may advise and this Court may permit.

February 27, 2014

NORTON ROSE FULBRIGHT CANADA LLP Royal Bank Plaza, South Tower, Suite 3800 200 Bay Street, P.O. Box 84 Toronto, Ontario M5J 2Z4 CANADA

Tony Reyes LSUC#: 28218V Tel: (416) 216-4825 Email: <u>Tony.Reyes@nortonrosefulbright.com</u>

Evan Cobb LSUC#: 55787N Tel: (416) 216-1929 Email: <u>Evan.Cobb@nortonrosefulbright.com</u>

Fax: (416) 216-3930

Lawyers for the Applicant, Jaguar Mining Inc.

TO THE ATTACHED SERVICE LIST

Court File No. CV-13-10383-00CL

ONTARIO SUPERIOR COURT OF JUSTICE COMMERCIAL LIST

IN THE MATTER OF THE COMPANIES' CREDITORS ARRANGEMENT ACT, R.S.C. 1985, c. C-36, AS AMENDED

AND IN THE MATTER OF A PLAN OF COMPROMISE OR ARRANGEMENT OF JAGUAR MINING INC.

Applicant

<u>Service List</u> (as at February 3, 2014)

Norton Rose Fulbright Canada LLP

Royal Bank Plaza, South Tower Suite 3800 200 Bay Street, P.O. Box 84 Toronto, Ontario M5J 2Z4

Tony Reyes

Tel: 416.216.4825 Email: tony.reyes@nortonrosefulbright.com

Evan Cobb

Tel: 416.216.1929 Email: evan.cobb@nortonrosefulbright.com

Counsel for the Applicant, Jaguar Mining Inc.

FTI Consulting Canada Inc.

79 Wellington Street West Suite 2010, P.O. Box 104 Toronto, Ontario M5K 1G8

Greg Watson

Tel: 416.649.8100 E-mail: greg.watson@fticonsulting.com

Jodi Porepa

Tel: 416.649.8070 E-mail: jodi.porepa@fticonsulting.com

Monitor

Osler Hoskin & Harcourt LLP

100 King Street West 1 First Canadian Place Suite 4600, P.O. Box 50 Toronto, Ontario M5X 1B8

Marc Wasserman

Tel: 416.862.4908 E-mail: <u>mwasserman@osler.com</u>

Michael De Lellis Tel: 416.862.5997 E-mail: mdelellis@osler.com

Counsel for FTI Consulting Canada Inc., as Monitor

Goodmans LLP

Bay Adelaide Centre Suite 3400 333 Bay Street Toronto, Ontario M5H 2S7

Robert Chadwick

Tel: 416.597.4285 Email: <u>rchadwick@goodmans.ca</u>

Melaney Wagner

Tel: 416.597.4258 Email: <u>mwagner@goodmans.ca</u>

Caroline Descours

Tel: 416.597.6275 Email: <u>cdescours@goodmans.ca</u>

Counsel for the Ad Hoc Committee of Holders of 5.5% Convertible Notes and 4.5% Convertible Notes

5

Davies Ward Phillips & Vineberg LLP

155 Wellington Street West Toronto, Ontario M5V 3J7

Robin Schwill

Tel: 416.863.5502 Email: <u>rschwill@dwpv.com</u>

Counsel to the Special Committee of the Board of Directors

Bennett Jones LLP 3400 One First Canadian Place P.O. Box 130 Toronto. Ontario M5X 1A4

Jeff Leon

Tel: 416.777.7472 Email: leonj@bennettjones.com

Sean Zweig

Tel: 416.777.6254 Email: <u>zweigs@bennettjones.com</u>

Counsel for The Bank of New York Mellon and BNY Trust Company of Canada, as co-trustees under the indentures governing the 4.5% Convertible Notes and the 5.5% Convertible Notes

6

Cassels Brock & Blackwell LLP Suite 2100, Scotia Plaza 40 King Street West Toronto, Ontario M5H 3C2

Joseph Bellissimo Tel: 416.860.6572 Email: jbellissimo@casselsbrock.com

Counsel for Global Resource Fund

Minden Gross LLP 145 King Street West, Suite 2200 Toronto, Ontario M5H 4G2

Raymond M. Slattery

Tel: 416.369.4149 Email: rslattery@mindengross.com

David T. Ullmann Tel: 416.369.4148 Email: dullmann@mindengross.com

Sepideh Nassabi Email: snassabi@mindengross.com

Fax: 416.864.9223

Lawyers for Daniel R. Titcomb, Robert J. Lloyd, James M. Roller, William E. Dow, Jeffrey Kirchhoff and Brazilian Resources, Inc.

Canada Revenue Agency

c/o Department of Justice Ontario Regional Office The Exchange Tower, Box 36 130 King Street West, Suite 3400 Toronto, Ontario M5X 1K6

Diane Winters

Tel: 416.973.3172 Fax: 416.973.0810 Email: diane.winters@justice.gc.ca

Blaney McMurtry LLP

2 Queen Street East, Suite 1500 Toronto, ON M5C 3G5

Lou Brzezinski

Tel: 416.593.2952 Fax: 416.594.5084 Email: lbrzezlnski@blaney.com

Counsel to Gilmore Clausen and Gary German

Supplemental Service List (as at February 3, 2014)

Sedgwick LLP

2 World Financial Center 225 Liberty Street 28th Floor New York, New York 10281-1008 USA

Laura K. Markovich

Tel: 212.422.0202 Fax: 212.422.0925 Email: laura.markovich@sedgwicklaw.com

Counsel to Catlin Underwriting Agencies Ltd., Liberty Mutual Insurance Europe Limited, Axis Specialty Europe Limited and Starr Underwriting Agents Limited

Allied World National Assurance Co. 225 Franklin Street

- 3 .- AA

8

Boston, MA 02110

2004 - M

and cont

David Colwick Email: David.Colwick@awac.com

Court File No. CV-13-10383-00CL

ONTARIO SUPERIOR COURT OF JUSTICE COMMERCIAL LIST

IN THE MATTER OF THE COMPANIES' CREDITORS ARRANGEMENT ACT, R.S.C. 1985, c. C-36, AS AMENDED

AND IN THE MATTER OF A PLAN OF COMPROMISE OR ARRANGEMENT OF JAGUAR MINING INC.

Applicant

Email Address List (as at February 3, 2014)

As required by Section 36 of the Commercial List E-Service Protocol, attached below is the Address List which may be copied and pasted into emails when serving Court Documents (as defined in the Protocol).

tony.reyes@nortonrosefulbright.com; evan.cobb@nortonrosefulbright.com; greg.watson@fticonsulting.com; jodi.porepa@fticonsulting.com; mwasserman@osler.com; MDeLellis@osler.com; rchadwick@goodmans.ca; mwagner@goodmans.ca; cdescours@goodmans.ca; rschwill@dwpv.com; leonj@bennettjones.com; ZweigS@bennettjones.com; JBellissimo@CasselsBrock.com; rslattery@mindengross.com; dullmann@mindengross.com; snassabi@mindengross.com; diane.winters@justice.gc.ca; lbrzezinski@blaney.com

Supplemental Email Address List (as at February 3, 2014)

laura.markovich@sedgwicklaw.com; David.Colwick@awac.com

DOCSTOR: 2906159\5

ACT,	
ARRANGEMENT A	
OF THE COMPANIES' CREDITORS	C-36, AS AMENDED
IN THE MATTER	R.S.C. 1985, c.

Court File No: CV-13-10383-00CL

and in the matter of a plan of compromise or arrangement of Jaguar mining inc.	
	ONTARIO SUPERIOR COURT OF JUSTICE (COMMERCIAL LIST) Proceeding commenced at Toronto
	NOTICE OF MOTION
	Norton Rose Fulbright Canada LLP Royal Bank Plaza, South Tower, Suite 3800 200 Bay Street, P.O. Box 84 Toronto, Ontario M5J 2Z4 CANADA
	Tony Reyes LSUC#: 28218V Tel: 416.216.4825 Email: tony.reyes@nortonrosefulbright.com
	Evan Cobb LSUC#: 55787N Tel: 416.216.1929 Email: evan.cobb@nortonrosefulbright.com
	Fax: 416.216.3930
	Lawyers for the Applicant, Jaguar Mining Inc.
DOCSTOR: 2944861\2	10

Tab 2

Takahan keung terdisi daraman dia dia kelalah di k

en en en el el deben de la complete en la constitue en el de la constitue en el de la constitue en el de la co

a shekara a s

Court File No. CV-13-10383-00CL

ONTARIO SUPERIOR COURT OF JUSTICE COMMERCIAL LIST

IN THE MATTER OF THE COMPANIES' CREDITORS ARRANGEMENT ACT, R.S.C. 1985, c. C-36, AS AMENDED

AND IN THE MATTER OF A PLAN OF COMPROMISE OR ARRANGEMENT OF JAGUAR MINING INC.

Applicant

AFFIDAVIT OF T. DOUGLAS WILLOCK (sworn February 27, 2014)

I, T. DOUGLAS WILLOCK, of the City of Toronto, MAKE OATH AND SAY:

1. I am the Chief Financial Officer of Jaguar Mining Inc. ("Jaguar" or the "Applicant"). I have held that position since January 10, 2013. As such, I have personal knowledge of the matters to which I hereinafter depose, except where otherwise stated. In preparing this affidavit I have also consulted, where necessary, with other members of Jaguar's management team or the management teams of its wholly-owned subsidiaries (together with Jaguar, the "Jaguar Group"). Where I have relied upon other sources of information, I have stated the source of that information and believe such information to be true.

2. I swear this affidavit in support of a motion by Jaguar for an Order pursuant to the *Companies' Creditors Arrangement Act*, R.S.C. 1985, c. C-36, as amended, (the "**CCAA**") to, among other things, extend the Stay Period (as such term is defined in the Initial Order of this Court dated December 23, 2013, as amended) to March 10, 2014.

3. All capitalized terms used herein and not otherwise defined have the meanings given to them in the Amended and Restated Plan of Compromise and Arrangement of the Applicant, dated January 31, 2014 (as further amended and restated on February 5, 2014, the "**Plan**").

Background

4. On December 23, 2013, the Applicant sought and obtained an Initial Order, a Claims Procedure Order and a Meeting Order in these proceedings. A copy of the Initial Order is attached hereto as Exhibit "A".

5. The principal objective of these proceedings is to effect a recapitalization and financing transaction (the "**Recapitalization**") on an expedited basis to provide a stronger financial foundation for the Jaguar Group going forward and additional liquidity to allow the Jaguar Group to continue to work towards its operational and financial goals.

6. The Recapitalization has been supported from the commencement of these proceedings, subject to certain conditions, by holders of 4.5% and/or 5.5% Senior Unsecured Convertible Notes issued by the Applicant (the "**Consenting Noteholders**"), including an ad hoc committee of holders of the Notes (the "**Ad Hoc Committee**"). Collectively, the Consenting Noteholders account for over 90% of the value of claims affected by the Recapitalization.

7. As described in my Affidavit sworn February 2, 2014, a copy of which is attached hereto (without exhibits) as Exhibit "B", the Plan was approved by 100% of the affected creditors voting on the Plan at the creditors' meeting held on January 31, 2014. The Plan would give effect to the Recapitalization.

8. The Plan was approved by an order of the court granted on February 6, 2014, a copy of which is attached hereto as Exhibit "C" (the "**Sanction Order**").

Progress Since Sanction Order

9. Since the issuance of the Sanction Order, the Jaguar Group has continued to operate in the ordinary course. The Applicant has operated within its cash flow forecast (as last filed with the Court on January 24, 2014) and has honoured its post-filing financial commitments as they have become due.

10. The Applicant has operated in compliance with the Initial Order, the Meeting Order, the Claims Procedure Order, and the Sanction Order, and has worked cooperatively with the Monitor throughout this period.

11. Steps taken to date towards the implementation of the Plan include:

- the Applicant has delivered notices to the Electing Eligible Investors pursuant to Section 4.1(d) of the Plan;
- (b) the Applicant has delivered notices to the Backstop Parties, pursuant to 4.1(f) of the Plan;
- (c) the Escrow Agent has accepted delivery from the Electing Eligible Investors and the Backstop Parties of funds in the full amount of the proceeds of the Share Offering under the Plan; and
- (d) the Applicant has worked with the Monitor, the Ad Hoc Committee, the Trustees,
 the Transfer Agent and DTC to prepare for the implementation of the Plan and
 the issuance of the New Common Shares.

12. The Applicant has made certain accommodations to Electing Eligible Investors who initiated wire transfers to the Escrow Agent prior to the Electing Eligible Investor Funding Deadline but whose Electing Eligible Investor Funding Amount was not received by the Escrow Agent prior to such funding deadline. Following consultation with the Monitor and counsel to the Ad Hoc Committee, the Applicant has directed the Escrow Agent to accept such wire transfers.

The only parties affected by such accommodation are the Backstop Parties, and the Applicant is unaware of any objection to this accommodation by the Backstop Parties.

13. The Applicant has continued to keep key stakeholders apprised of developments in these proceedings. Discussions between the Applicant and its senior secured lender, Global Resource Fund, regarding the Recapitalization have continued and the Applicant has continued to comply with its ongoing obligations under its senior secured credit facility with Global Resource Fund.

Plan Implementation

14. The Plan contemplates a number of steps involving a large number of parties.

15. Implementation of the Plan remains subject to the satisfaction or waiver of certain conditions precedent set out therein, including:

- (a) the Credit Agreement with Global Resource Fund (and the other Credit Documents) shall have been amended on or prior to three Business Days prior to the Implementation Date on terms acceptable to the Majority Backstop Parties;
- (b) composition of the board of directors of Jaguar effective as of the Implementation
 Date shall be satisfactory to the Majority Backstop Parties;
- (c) all senior officer and employee employment agreements shall have been modified to reflect the revised capital structure of the Applicant following implementation of the Plan, including, without limitation, to provide that the implementation of the Plan does not constitute a change of control under such agreements and no change of control payments shall be owing or payable to the Applicant's officers or employees in connection with the implementation of the Plan;
- (d) the terms of any Management Incentive Plan shall be acceptable to the Majority Backstop Parties; and

(e) the New Common Shares shall have been conditionally approved for listing on the TSX, the TSXV or such other Designated Offshore Securities Market acceptable to the Majority Consenting Noteholders without any vote or approval of the Existing Shareholders, subject only to receipt of customary final documentation.

16. The Applicant cannot be certain that all such conditions will be satisfied or waived by February 28, 2014, the date on which the Stay Period currently expires. As a result, the Applicant believes that it is prudent to seek a further brief extension of the Stay Period at this time.

17. The Support Agreement entered into with the Consenting Noteholders is subject to an outside date of February 28, 2014. The implementation of the Plan itself is also subject to an outside date of February 28, 2014, or such other date as the Applicant and the Majority Consenting Noteholders may agree. The Applicant and the Majority Consenting Noteholders may agree. The Applicant and the Majority Consenting Noteholders have agreed to extend the outside date under the Support Agreement and the Plan beyond February 28, 2014. The Applicant and the Majority Backstop Parties have similarly agreed to extend the purposes of the Backstop Agreement beyond February 28, 2014.

Request for a Stay Extension

18. The Applicant continues to believe that the Plan is the best available alternative in the circumstances following the consideration of available alternatives. Absent the Plan, liquidation would appear to be the likely remaining alternative for the Applicant, which would be detrimental to all stakeholders.

19. The Applicant requires more time to satisfy the conditions to implementation of the Plan sanctioned by the Court. The Applicant has no reasonable basis to conclude at this time that the conditions that remain outstanding will not be met.

20. The Applicant, with the assistance of the Monitor, has prepared an updated cash flow forecast (the "**Updated Cash Flow Statement**"), which covers the requested extension of the Stay Period. I understand that the Updated Cash Flow Statement will be attached as an appendix to the Monitor's Fourth Report to be served in connection with this motion.

21. The Updated Cash Flow Statement shows that the Applicant has sufficient resources to meet its post-filing obligations during the brief extension of the Stay Period.

22. I am advised that each of the Monitor and the Ad Hoc Committee support the requested extension of the Stay Period.

Amendments to Plan

23. The Plan has been amended to modify the definition of "SAR Plan" by deleting the existing definition and replacing it with the following:

"**SAR Plan**" means the Fourth Amended and Restated Share Appreciation Rights Plan of Jaguar, effective as of October 4, 2013

(the "SAR Amendment").

24. Prior to this amendment, the Plan referred to an out-of-date version of the Share Appreciation Rights Plan. The Applicant intended for this definition to refer to the current version of the Share Appreciation Rights Plan. This amendment to the Plan achieves that purpose.

25. The Plan has also been amended by deleting the existing Section 10.2(a) and replacing that section with the following:

Unsecured Creditor Common Shares in relation to a Disputed Distribution Claim of an Affected Unsecured Creditor will be, on or prior to the Implementation Date, either: (i) issued by the Applicant and held by the Applicant, in a segregated account; or

(ii) authorized by the Applicant's board of directors for issuance by the Applicant,

which in either case shall constitute the Disputed Distribution Claims Reserve for the benefit of the Affected Unsecured Creditors with Allowed Affected Unsecured Claims until the final determination of the Disputed Distribution Claim in accordance with the Claims Procedure Order and this Plan.

(together with the SAR Amendment, the "Amendments").

26. A copy of the Plan reflecting the Amendments is attached hereto as Exhibit "D".

27. The Amendments are purely administrative in nature and are required to better give effect to the implementation of the Plan and/or to cure any errors, omissions or ambiguities and are not materially adverse to the financial or economic interests of the Affected Unsecured Creditors under the Plan.

SWORN BEFORE ME at the City of Toronto, Province of Ontario, on February 27, 2014.

Commissioner for Taking Affidavits

T. DOUGLAS WILLOCK

Exhibit A

.

the arrest and

Concernent Marchanist, and the second s

Court File No. CV = 13 - 10383 - 00CL

8 |

ONTARIO SUPERIOR COURT OF JUSTICE COMMERCIAL LIST

This is Extribut 1 A ... referred to in the effective of T. DougerAS WILLOCK

THE HONOURABLE

MONDAY, THE 23RD

JUSTICE MORAWETZ

)) DAY OF DECEMBER, 2013^{Builden belore me, this...}

osy of...

IN THE MATTER OF THE COMPANIES' CREDITORS ARRANGEMENT ACT, R.S.C. 1985, c. C-36, AS AMENDED

AND IN THE MATTER OF A PLAN OF COMPROMISE OR ARRANGEMENT OF JAGUAR MINING INC.

Applicant

INITIAL ORDER

THIS APPLICATION, made by the Applicant, pursuant to the *Companies' Creditors Arrangement Act*, R.S.C. 1985, c. C-36, as amended (the "CCAA") was heard this day at 330 University Avenue, Toronto, Ontario.

ON READING the affidavit of David M. Petroff sworn December 23, 2013 and the Exhibits thereto (the "**Petroff Affidavit**"), the Pre-Filing Report of FTI Consulting Canada Inc. in its capacity as the Proposed Monitor (as defined in the Petroff Affidavit), dated December 21, 2013, and on being advised that the secured creditors who are likely to be affected by the charges created herein were given notice, and on hearing the submissions of counsel for the Applicant, FTI Consulting Canada Inc., as Proposed Monitor, the Ad Hoc Committee (as defined in the Petroff Affidavit), and Global Resource Fund, no one appearing for any other person although duly served as appears from the affidavit of service of Evan Cobb sworn December 23, 2013 and on reading the consent of FTI Consulting Canada Inc. to act as the Monitor (in such capacity, the "**Monitor**"),

DOCSTOR: 2885569\6

SERVICE

1. THIS COURT ORDERS that the time for service of the Notice of Application and the Application Record is hereby abridged and validated so that this Application is properly returnable today and hereby dispenses with further service thereof.

APPLICATION

2. THIS COURT ORDERS AND DECLARES that the Applicant is a company to which the CCAA applies.

PLAN OF ARRANGEMENT

3. THIS COURT ORDERS that the Applicant shall have the authority to file and may, subject to further order of this Court, file with this Court a plan of compromise or arrangement (hereinafter referred to as the "Plan").

POSSESSION OF PROPERTY AND OPERATIONS

4. THIS COURT ORDERS that the Applicant shall remain in possession and control of its current and future assets, undertakings and properties of every nature and kind whatsoever, and wherever situate including all proceeds thereof (the "Property"). Subject to further Order of this Court, the Applicant shall continue to carry on business in a manner consistent with the preservation of its business (the "Business") and Property. The Applicant shall be authorized and empowered to continue to retain and employ the employees, consultants, agents, experts, accountants, directors, counsel and such other persons, including counsel to the Special Committee (as defined in the Petroff Affidavit) (collectively "Assistants") currently retained or employed by it, with liberty to retain such further Assistants as it deems reasonably necessary or desirable in the ordinary course of business or for the carrying out of the terms of this Order.

5. THIS COURT ORDERS that the Applicant shall be entitled but not required to pay the following expenses whether incurred prior to or after this Order:

 (a) all outstanding and future wages, salaries, employee and pension benefits, vacation pay and expenses payable on or after the date of this Order, in each case incurred in the ordinary course of business and consistent with existing compensation policies

DOCSTOR. 2885569\6

and arrangements; and

(b) the fees and disbursements of any Assistants retained or employed by the Applicant in respect of these proceedings or in respect of the Applicant's public listingrequirements, at their standard rates and charges.

6. THIS COURT ORDERS that, except as otherwise provided to the contrary herein or in the Support Agreement, the Applicant shall be entitled but not required to pay all reasonable expenses incurred by the Applicant in carrying on the Business in the ordinary course after this Order, and in carrying out the provisions of this Order, which expenses shall include, without limitation:

- (a) all expenses and capital expenditures reasonably necessary for the preservation of the Property or the Business including, without limitation, payments on account of insurance (including directors and officers insurance), maintenance and security services; and
- (b) payment for goods or services actually supplied to the Applicant following the date of this Order.

7. THIS COURT ORDERS that the Applicant shall remit, in accordance with legal requirements, or pay:

- (a) any statutory deemed trust amounts in favour of the Crown in right of Canada or of any Province thereof or any other taxation authority which are required to be deducted from employees' wages, including, without limitation, amounts in respect of

 (i) employment insurance, (ii) Canada Pension Plan, (iii) Quebec Pension Plan, and
 (iv) income taxes;
- (b) all goods and services or other applicable sales taxes (collectively, "Sales Taxes") required to be remitted by the Applicant in connection with the sale of goods and services by the Applicant, but only where such Sales Taxes are accrued or collected after the date of this Order, or where such Sales Taxes were accrued or collected prior to the date of this Order but not required to be remitted until on or after the date of this Order, and

(c) any amount payable to the Grown in right of Canada or of any Province thereof or any political subdivision thereof or any other taxation authority in respect of municipal realty, municipal business or other taxes, assessments or levies of any nature or kind which are entitled at law to be paid in priority to claims of secured creditors and which are attributable to or in respect of the carrying on of the Business by the Applicant.

8. THIS COURT ORDERS that until a real property lease is disclaimed in accordance with the CCAA, the Applicant shall pay all amounts constituting rent or payable as rent under real property leases (including, for greater certainty, common area maintenance charges, utilities and realty taxes and any other amounts payable to the landlord under the lease) or as otherwise may be negotiated between the Applicant and the landlord from time to time ("Rent"), for the period commencing from and including the date of this Order, twice-monthly in equal payments on the first and fifteenth day of each month, in advance (but not in arrears). On the date of the first of such payments, any Rent relating to the period commencing from and including the date of this Order shall also be paid.

9. THIS COURT ORDERS that, except as specifically permitted herein, the Applicant is hereby directed, until further Order of this Court: (a) to make no payments of principal, interest thereon or otherwise on account of amounts owing by the Applicant to any of its creditors as of this date; (b) to grant no security interests, trust, liens, charges or encumbrances upon or in respect of any of its Property; and (c) to not grant credit or incur liabilities except in the ordinary course of the Business. Notwithstanding the foregoing, the Applicant is authorized and directed until further order of this Court to pay any monthly interest amounts that may become due and owing to Global Resource Fund under the Renvest Facility (as such term is defined in the Petroff Affidavit).

RESTRUCTURING

10. THIS COURT ORDERS that the Applicant shall, subject to such requirements as are imposed by the CCAA and the terms of the Support Agreement, have the right to:

 (a) permanently or temporarily cease, downsize or shut down any of its business or operations;

DOCSTOR: 2885569\6

- (b) terminate the employment of such of its employees as it deems appropriate;
- (c) retain a solicitation agent and an election agent (the "Solicitation/Election Agent") and permit it to obtain proxies and/or voting information and subscription election forms from registered and beneficial holders of the Notes (as defined in the Petroff Affidavit) in respect of the Plan and any amendments thereto; and
- (d) pursue all avenues of refinancing of its Business or Property, in whole or part, subject to prior approval of this Court being obtained before any material refinancing,

all of the foregoing to permit the Applicant to proceed with an orderly restructuring of the Business (the "Restructuring").

11. THIS COURT ORDERS that the Applicant shall provide each of the relevant landlords with notice of the Applicant's intention to remove any fixtures from any leased premises at least seven (7) days prior to the date of the intended removal. The relevant landlord shall be entitled to have a representative present in the leased premises to observe such removal and, if the landlord disputes the Applicant's entitlement to remove any such fixture under the provisions of the lease, such fixture shall remain on the premises and shall be dealt with as agreed between any applicable secured creditors, such landlord and the Applicant, or by further Order of this Court upon application by the Applicant on at least two (2) days notice to such landlord and any such secured creditors. If the Applicant disclaims the lease governing such leased premises in accordance with Section 32 of the CCAA, it shall not be required to pay Rent under such lease pending resolution of any such dispute (other than Rent payable for the notice period provided for in Section 32(5) of the CCAA), and the disclaimer of the lease shall be without prejudice to the Applicant's claim to the fixtures in dispute.

12. THIS COURT ORDERS that if a notice of disclaimer is delivered pursuant to Section 32 of the CCAA, then (a) during the notice period prior to the effective time of the disclaimer, the landlord may show the affected leased premises to prospective tenants during normal business hours, on giving the Applicant and the Monitor 24 hours' prior written notice, and (b) at the effective time of the disclaimer, the relevant landlord shall be entitled to take possession of any such leased premises without waiver of or prejudice to any claims or rights such landlord may have against the Applicant in respect of such lease or leased premises and such landlord shall be entitled to notify the Applicant of the basis on which it is taking possession and to gain

23

possession of and re-lease such leased premises to any third party or parties on such terms as such landlord considers advisable, provided that nothing herein shall relieve such landlord of its obligation to mitigate any damages claimed in connection therewith.

SUPPORT AGREEMENT AND BACKSTOP AGREEMENT

13. THIS COURT ORDERS that the Applicant is authorized and empowered to take all steps and actions in respect of, and to comply with all of its obligations pursuant to, the Support Agreement and the Backstop Agreement (each as defined in the Petroff Affidavit) and its various obligations thereunder, and that nothing in this Order shall be construed as waiving or modifying any of the rights, commitments or obligations of Jaguar, its Subsidiaries, the Consenting Noteholders (as defined in the Petroff Affidavit) and the Backstop Parties (äs defined in the Petroff Affidavit) under the Support Agreement and the Backstop Agreement, as applicable.

NO PROCEEDINGS AGAINST THE APPLICANT OR THE PROPERTY

14. THIS COURT ORDERS that until and including January 22, 2014, or such later date as this Court may order (the "Stay Period"), no proceeding or enforcement process in any court or tribunal (each, a "Proceeding") shall be commenced or continued against or in respect of the Applicant or the Monitor, or affecting the Business or the Property, except with the written consent of the Applicant and the Monitor, or with leave of this Court, and any and all Proceedings currently under way against or in respect of the Applicant or affecting the Business or the Property are hereby stayed and suspended pending further Order of this Court.

15. THIS COURT ORDERS that during the Stay Period, no Proceeding shall be commenced or continued: (i) against or in respect of any of the Applicant's direct or indirect subsidiaries (each a "Subsidiary" and, collectively, the "Subsidiarles") with respect to any guarantee, contribution or indemnity obligation, liability or claim in respect of, or that relates to, any agreement involving the Applicant, or the obligations, liabilities and claims of, against or affecting the Applicant or the Business (collectively, the "Applicant Related Liabilities"); (ii) against or in respect of any of a Subsidiary's current and future assets, undertakings and properties of every nature and kind whatsoever, and wherever situate including all proceeds thereof (the "Subsidiary Property") with respect to any Applicant Related Liabilities (the matters referred to in (i) and (ii) being, collectively, the "Applicant Related Proceedings Against

24

Subsidiaries"), except with the written consent of the Applicant and the Monitor, or with leave of this Court, and any and all Applicant Related Proceedings Against Subsidiaries currently under way by any Person are hereby stayed and suspended pending further order of this Court.

NO EXERCISE OF RIGHTS OR REMEDIES

16. THIS COURT ORDERS that during the Stay Period, all rights and remedies of any Individual, firm, corporation, governmental body or agency, or any other entities (all of the foregoing, collectively being "Persons" and each being a "Person") against or in respect of the Applicant or the Monitor, or affecting the Business or the Property, are hereby stayed and suspended except with the written consent of the Applicant and the Monitor, or leave of this Court, provided that nothing in this Order shall (i) empower the Applicant to carry on any business which the Applicant is not lawfully entitled to carry on, (ii) affect such investigations, actions, suits or proceedings by a regulatory body as are permitted by Section 11.1 of the CCAA, (iii) prevent the filling of any registration to preserve or perfect a security interest, or (iv) prevent the registration of a claim for lien.

17. THIS COURT ORDERS that during the Stay Period, all rights and remedies of any Person against or in respect of any Subsidiary or Subsidiary Property in respect of any Applicant Related Liabilities are hereby stayed and suspended except with the written consent of the Applicant and the Monitor, or leave of this Court, provided that nothing in this Order shall (i) empower the Subsidiary to carry on any business which the Subsidiary is not lawfully entitled to carry on, (ii) affect such investigations, actions, suits or proceedings by a regulatory body as are permitted by Section 11.1 of the CCAA, (iii) prevent the filing of any registration to preserve or perfect a security interest, or (iv) prevent the registration of a claim for lien.

NO INTERFERENCE WITH RIGHTS

18. THIS COURT ORDERS that during the Stay Period, no Person shall discontinue, fail to honour, alter, interfere with, repudiate, terminate or cease to perform any right, renewal right, contract, agreement, licence or permit in favour of or held by the Applicant, except with the written consent of the Applicant and the Monitor, or leave of this Court.

CONTINUATION OF SERVICES

19. THIS COURT ORDERS that during the Stay Period, all Persons having oral or written

agreements with the Applicant or statutory or regulatory mandates for the supply of goods and/or services, including without limitation all computer software, communication and other data services, centralized banking services, payroll services, insurance, transportation services, utility or other services to the Business or the Applicant, are hereby restrained until further Order of this Court from discontinuing, altering, interfering with or terminating the supply of such goods or services as may be required by the Applicant, and that the Applicant shall be entitled to the continued use of its current premises, telephone numbers, facsimile numbers, internet addresses and domain names, provided in each case that the normal prices or charges for all such goods or services received after the date of this Order are paid by the Applicant in accordance with normal payment practices of the Applicant or such other practices as may be agreed upon by the supplier or service provider and each of the Applicant and the Monitor, or as may be ordered by this Court.

NON-DEROGATION OF RIGHTS

20. THIS COURT ORDERS that, notwithstanding anything else in this Order, no Person shall be prohibited from requiring immediate payment for goods, services, use of lease or licensed property or other valuable consideration provided on or after the date of this Order, nor shall any Person be under any obligation on or after the date of this Order to advance or readvance any monies or otherwise extend any credit to the Applicant. Nothing in this Order shall derogate from the rights conferred and obligations imposed by the CCAA.

PROCEEDINGS AGAINST DIRECTORS AND OFFICERS

21. THIS COURT ORDERS that during the Stay Period, and except as permitted by subsection 11.03(2) of the CCAA, no Proceeding may be commenced or continued against any of the former, current or future directors or officers of the Applicant with respect to any claim against the directors or officers that arose before the date hereof and that relates to any obligations of the Applicant whereby the directors or officers are alleged under any law to be liable in their capacity as directors or officers for the payment or performance of such obligations, until a compromise or arrangement in respect of the Applicant, if one is filed, is sanctioned by this Court or is refused by the creditors of the Applicant or this Court.

25

DIRECTORS' AND OFFICERS' INDEMNIFICATION AND CHARGE

22. THIS COURT ORDERS that the Applicant shall indemnify its directors and officers against obligations and liabilities that they may incur as directors or officers of the Applicant after the commencement of the within proceedings, except to the extent that, with respect to any officer or director, the obligation or liability was incurred as a result of the director's or officer's gross negligence or wilful misconduct.

23. THIS COURT ORDERS that the directors and officers of the Applicant shall be entitled to the benefit of and are hereby granted a charge (the "Directors' Charge") on the Property, which charge shall not exceed an aggregate amount of \$150,000, as security for the indemnity provided in paragraph 22 of this Order. The Directors' Charge shall have the priority set out in paragraphs 37 and 40 herein.

24. THIS COURT ORDERS that, notwithstanding any language in any applicable insurance policy to the contrary, (a) no insurer shall be entitled to be subrogated to or claim the benefit of the Directors' Charge, and (b) the Applicant's directors and officers shall only be entitled to the benefit of the Directors' Charge to the extent that they do not have coverage under any directors' and officers' insurance policy, or to the extent that such coverage is insufficient to pay amounts indemnified in accordance with paragraph 22 of this Order.

APPOINTMENT OF MONITOR

25. THIS COURT ORDERS that FTI Consulting Canada Inc. is hereby appointed pursuant to the CCAA as the Monitor, an officer of this Court, to monitor the business and financial affairs of the Applicant with the powers and obligations set out in the CCAA or set forth herein and that the Applicant and its shareholders, officers, directors, and Assistants shall advise the Monitor of all material steps taken by the Applicant pursuant to this Order, and shall co-operate fully with the Monitor in the exercise of its powers and discharge of its obligations and provide the Monitor with the assistance that is necessary to enable the Monitor to adequately carry out the Monitor's functions.

26. THIS COURT ORDERS that the Monitor, in addition to its prescribed rights and obligations under the CCAA, is hereby directed and empowered to:

- (a) monitor the Applicant's receipts and disbursements;
- (b) report to this Court at such times and intervals as the Monitor may deem appropriate with respect to matters relating to the Property, the Business, and such other matters as may be relevant to the proceedings herein;
- (c) advise the Applicant in its preparation of the Applicant's cash flow statements;
- (d) advise the Applicant on any amendments to the Plan;
- (e) assist the Applicant, to the extent required by the Applicant, with the holding and administering of creditors' meetings for voting on the Plan;
- (f) have full and complete access to the Property, including the premises, books, records, data, including data in electronic form, and other financial documents of the Applicant, to the extent that is necessary to adequately assess the Applicant's business and financial affairs or to perform its duties arising under this Order;
- (g) assist the Solicitation/Election Agent to obtain proxies and/or voting information and subscription election forms from registered and beneficial holders of the Notes in respect of the Plan and any amendments thereto;
- (h) assist the Applicant, to the extent required by the Applicant, with its restructuring activities;
- (i) assist the Applicant, to the extent required by the Applicant, with any matters relating to any foreign proceedings commenced in relation to the Applicant, including retaining independent legal counsel, agents, experts, accountants or such other persons as the Monitor deems necessary or advisable respecting the exercise of this power;
- (j) engage in discussions with the Ad Hoc Committee and the Applicant's secured creditors, independent of the Applicant and, to the extent that any written reports with respect to these proceedings are delivered by the Monitor (or its advisors) to the Ad Hoc Committee (or its advisors), copies of those written reports shall be delivered by the Monitor (or its advisors) to Global Resource Fund (or its advisors) as soon as

27

reasonably practicable following delivery to the Ad Hoc Committee;

- (k) be at liberty to engage independent legal counsel or such other persons as the Monitor deems necessary or advisable respecting the exercise of its powers and performance of its obligations under this Order; and
- (I) perform such other duties as are required by this Order or by this Court from time to time.

27. THIS COURT ORDERS that the Monitor shall not take possession of the Property and shall take no part whatsoever in the management or supervision of the management of the Business and shall not, by fulfilling its obligations hereunder, be deemed to have taken or maintained possession or control of the Business or Property, or any part thereof.

28. THIS COURT ORDERS that nothing herein contained shall require the Monitor to occupy or to take control, care, charge, possession or management (separately and/or collectively, "Possession") of any of the Property (or any Subsidiary Property) that might be environmentally contaminated, might be a pollutant or a contaminant, or might cause or contribute to a spill, discharge, release or deposit of a substance contrary to any federal, provincial or other law respecting the protection, conservation, enhancement, remediation or rehabilitation of the environment or relating to the disposal of waste or other contamination including, without limitation, the Canadian Environmental Protection Act, the Ontario Environmental Protection Act, the Ontario Water Resources Act, or the Ontario Occupational Health and Safety Act and regulations thereunder (the "Environmental Legislation"), provided however that nothing herein shall exempt the Monitor from any duty to report or make disclosure imposed by applicable Environmental Legislation. The Monitor shall not, as a result of this Order or anything done in pursuance of the Monitor's duties and powers under this Order, be deemed to be in Possession of any of the Property (or any Subsidiary Property) within the meaning of any Environmental Legislation, unless it is actually in possession.

29. THIS COURT ORDERS that the Monitor shall provide any creditor of the Applicant with information provided by the Applicant in response to reasonable requests for information made in writing by such creditor addressed to the Monitor. The Monitor shall not have any responsibility or liability with respect to the information disseminated by it pursuant to this paragraph. In the case of information that the Monitor has been advised by the Applicant is

confidential, the Monitor shall not provide such information to creditors unless otherwise directed by this Court or on such terms as the Monitor and the Applicant may agree.

30. THIS COURT ORDERS that, in addition to the rights and protections afforded the Monitor under the CCAA or as an officer of this Court, the Monitor shall incur no liability or obligation as a result of its appointment or the carrying out of the provisions of this Order, save and except for any gross negligence or wilful misconduct on its part. Nothing in this Order shall derogate from the protections afforded the Monitor by the CCAA or any applicable legislation.

31. THIS COURT ORDERS that the Monitor, domestic and foreign counsel to the Monitor, domestic and foreign counsel to the Applicant, counsel to the Special Committee (as defined in the Petroff Affidavit) domestic and foreign counsel to the Ad Hoc Committee and counsel to Global Resource Fund shall be paid their reasonable fees and disbursements, in each case at their standard rates and charges, whether incurred prior to or after the date of this Order, by the Applicant as part of the costs of these proceedings; and (ii) the Financial Advisors (as defined in the Petroff Affidavit) shall be paid their reasonable fees and disbursements, in each case in accordance with the terms of the FA Engagement Letters (as defined in the Petroff Affidavit), whether incurred prior to or after the date of this Order. The Applicant is hereby authorized and directed to pay the accounts of the Monitor, domestic and foreign counsel for the Monitor, domestic and foreign counsel for the Applicant, domestic and foreign counsel for the Ad Hoc Committee and counsel to the Special Committee weekly, or on such basis as otherwise agreed by the Applicant and the applicable payee and, in addition, the Applicant is hereby authorized to pay to the Monitor and counsel for the Monitor retainers in the amounts of \$75,000 and \$40,000, respectively, to be held by them as security for payment of their respective fees and disbursements outstanding from time to time.

32. THIS COURT ORDERS that the Monitor and its legal counsel shall pass their accounts from time to time, and for this purpose the accounts of the Monitor and its legal counsel are hereby referred to a judge of the Commercial List of the Ontario Superior Court of Justice.

33. THIS COURT ORDERS that the Monitor, domestic and foreign counsel to the Monitor, the Applicant's domestic and foreign counsel, counsel to the Special Committee, domestic and foreign counsel to the Ad Hoc Committee and the Financial Advisors shall be entitled to the benefit of and are hereby granted a charge (the "Administration Charge") on the Property, which

charge shall not exceed an aggregate amount of \$5,000,000, as security for their professional fees and disbursements incurred at their standard rates and charges, and in the case of the Financial Advisors, professional fees and disbursements incurred pursuant to the terms of the FA Engagement Letters, both before and after the making of this Order in respect of these proceedings. The Administration Charge shall consist of two separate charges (the Primary Administration Charge and the Subordinate Administration Charge (each as defined below)) with the priorities set out in paragraphs 37 and 40 hereof.

APPROVAL OF FINANCIAL ADVISORS' ENGAGEMENT

34. THIS COURT ORDERS that the Applicant is authorized to continue the engagement of the Financial Advisors on the terms and conditions set out in the FA Engagement Letters.

35. THIS COURT ORDERS that the FA Engagement Letters be and are hereby ratified and confirmed and the Applicant is authorized to perform its obligations thereunder.

36. THIS COURT ORDERS that any claims of the Financial Advisors under the FA Engagement Letters shall be treated as unaffected in any Plan.

VALIDITY AND PRIORITY OF CHARGES CREATED BY THIS ORDER

37. THIS COURT ORDERS that the priorities of the Directors' Charge, the Primary Administration Charge, the Renvest Security (as defined below) and the Subordinated Administration Charge, as among them, shall be as follows:

First – Administration Charge (to the maximum amount of \$500,000) (the "Primary Administration Charge");

Second - Directors' Charge (to the maximum amount of \$150,000);

Third - Renvest Security; and

Fourth – the Administration Charge (to a maximum of \$4,500,000) (the "Subordinated Administration Charge").

38. THIS COURT ORDERS that notwithstanding anything to the contrary herein, each of the Financial Advisors shall only be entitled to the benefit of the Primary Administration Charge with

respect to their respective monthly work fees as set out in the terms and conditions of their respective FA Erigagement Letters.

31

39. THIS COURT ORDERS that the filing, registration or perfection of the Directors' Charge, or the Administration Charge (collectively, the "Charges") shall not be required, and that the Charges shall be valid and enforceable for all purposes, including as against any right, title or interest filed, registered, recorded or perfected subsequent to the Charges coming into existence, notwithstanding any such failure to file, register, record or perfect.

40. THIS COURT ORDERS that each of the Directors' Charge and the Administration Charge (all as constituted and defined herein) shall constitute a charge on the Property and, except as provided in Paragraph 37, with respect to the Subordinated Administration Charge, such Charges shall rank in priority to all other security interests, trusts, liens, charges and encumbrances, claims of secured creditors, statutory or otherwise (collectively, "Encumbrances") in favour of any Person, with the exception of any Encumbrances ranking in priority to the security granted by the Applicant to secure the obligations under the Renvest Facility prior to the date hereof (the "Renvest Security").

41. THIS COURT ORDERS that except as otherwise expressly provided for herein, or as may be approved by this Court, the Applicant shall not grant any Encumbrances over any Property that rank in priority to, or *pari passu* with, any of the Directors' Charge, the Administration Charge, or the Renvest Security unless the Applicant also obtains the prior written consent of the Monitor, and the beneficiaries of the Directors' Charge and the Administration Charge, and (if such Encumbrances rank in priority to, or *pari passu* with, the Renvest Security) Global Resource Fund, or further Order of this Court.

42. THIS COURT ORDERS that the Directors' Charge and the Administration Charge shall not be rendered invalid or unenforceable and the rights and remedies of the chargees entitled to the benefit of the Charges (collectively, the "Chargees") thereunder shall not otherwise be limited or impaired in any way by (a) the pendency of these proceedings and the declarations of insolvency made herein; (b) any application(s) for bankruptcy order(s) issued pursuant to BIA, or any bankruptcy order made pursuant to such applications; (c) the filing of any assignments for the general benefit of creditors made pursuant to the BIA; (d) the provisions of any federal or provincial statutes; or (e) any negative covenants, prohibitions or other similar provisions with
respect to borrowings, incurring debt or the creation of Encumbrances, contained in any existing loan documents, lease, sublease, offer to lease or other agreement (collectively, an "Agreement") which binds the Applicant, and notwithstanding any provision to the contrary in any Agreement;

- the creation of the Charges shall not create or be deemed to constitute a breach by the Applicant of any Agreement to which it is a party;
- (b) none of the Chargees shall have any liability to any Person whatsoever as a result of any breach of any Agreement caused by or resulting from the creation of the Charges; and
- (c) the payments made by the Applicant pursuant to this Order and the granting of the Charges, do not and will not constitute preferences, fraudulent conveyances, transfers at undervalue, oppressive conduct, or other challengeable or voidable transactions under any applicable law.

43. THIS COURT ORDERS that any Charge created by this Order over leases of real property in Canada shall only be a Charge in the Applicant's interest in such real property leases.

SERVICE AND NOTICE

44. THIS COURT ORDERS that the Monitor shall (i) as soon as practicable after the granting of this Order, publish in the Globe and Mail (National Edition) and the Wall Street Journal a notice containing the information prescribed under the CCAA, (ii) within five days after the date of this Order, (A) make this Order publicly available in the manner prescribed under the CCAA, (B) send, in the prescribed manner, a notice to every known creditor who has a claim against the Applicant of more than \$1000, and (C) prepare a list showing the names and addresses of those creditors and the estimated amounts of those claims, and make it publicly available in the prescribed manner, all in accordance with Section 23(1)(a) of the CCAA and the regulations made thereunder. The list included in subparagraph (C) above shall not include the names, addresses or estimated amounts of those creditors who are individuals or any personal information in respect of an individual.

45. THIS COURT ORDERS that the Applicant and the Monitor be at liberty to serve this

Order, any other materials and orders in these proceedings, any notices or other correspondence, by forwarding true copies thereof by prepaid ordinary mail, courier, personal delivery or electronic transmission to the Applicant's creditors or other interested parties at their respective addresses as last shown on the records of the Applicant and that any such service or notice by courier, personal delivery or electronic transmission shall be deemed to be received on the next business day following the date of forwarding thereof, or if sent by ordinary mail, on the third business day after mailing.

46. THIS COURT ORDERS that the Applicant, the Monitor, the Ad Hoc Committee, Global Resource Fund and any party who has filed a Notice of Appearance may serve any court materials in these proceedings by e-mailing a PDF or other electronic copy of such materials to counsels' email addresses as recorded on the Service List from time to time, and the Monitor may post a copy of any or all such materials on its website at:

http://cfcanada.fticonsulting.com/jaguar.

47. THIS COURT ORDERS that all written reports delivered by the Applicant (or its advisors) to the Ad Hoc Committee (or its advisors) with respect to these proceedings shall also be delivered by the Applicant (or its advisors) to Global Resource Fund (or its advisors) as soon as reasonably practicable following delivery to the Ad Hoc Committee.

SEALING OF CONFIDENTIAL EXHIBITS

48. THIS COURT ORDERS that Confidential Exhibits "A" and "B" be and are hereby sealed pending further Order of the Court and shall not form part of the public record.

GENERAL

49. THIS COURT ORDERS that the Applicant or the Monitor may from time to time apply to this Court for advice and directions in the discharge of its powers and duties hereunder.

50. THIS COURT ORDERS that nothing in this Order shall prevent the Monitor from acting as an interim receiver, a receiver, a receiver and manager, or a trustee in bankruptcy of the Applicant, the Business or the Property.

51. THIS COURT HEREBY REQUESTS the aid and recognition of any court, tribunal,

regulatory or administrative body having jurisdiction in Canada, the United States, Brazil or elsewhere to give effect to this Order and to assist the Applicant, the Monitor and their respective agents in carrying out the terms of this Order. All courts, tribunals, regulatory and administrative bodies are hereby respectfully requested to make such orders and to provide such assistance to the Applicant and to the Monitor, as an officer of this Court, as may be necessary or desirable to give effect to this Order, to grant representative status to the Monitor in any foreign proceeding, or to assist the Applicant and the Monitor and their respective agents in carrying out the terms of this Order.

52. THIS COURT ORDERS that each of the Applicant and the Monitor be at liberty and is hereby authorized and empowered to apply to any court, tribunal, regulatory or administrative body, wherever located, for the recognition of this Order and for assistance in carrying out the terms of this Order, and that the Monitor is authorized and empowered to act as a representative in respect of the within proceedings for the purpose of having these proceedings recognized in a jurisdiction outside Canada.

53. THIS COURT ORDERS that any interested party (including the Applicant and the Monitor) may apply to this Court to vary or amend this Order on not less than seven (7) days notice to the Applicant, the Monitor, Global Resource Fund, the Ad Hoc Committee and any other party or parties likely to be affected by the order sought or upon such other notice, if any, as this Court may order.

54. THIS COURT ORDERS that this Order and all of its provisions are effective as of 12:01 a.m. Eastern Standard/Daylight Time on the date of this Order.

GEC 2 3 XBB

DOCSTOR: 2885569\6

ONTARIO SUPERIOR COURT OF JUSTICE COMMERCIAL LIST Proceeding commenced at Teronto
INITIAL ORDER
Norton Rose Fulbright Canada LLP Royal Bank Plaza, South Tower, Suite 3800 200 Bay Street, P.O. Box 84 Toronto, Ontario M5J 2Z4 CANADA
Tony Reyes LSUC#: 28218V Tel: 416/216.4825 Email: tony.reyes@nortonrosefulbright.com
Evan Cobb LSUC#: 55787N Tel: 416.216.1929 Ernail: evan.cobb@nortonrosefulbright.com
Fax: 416.216.3930
Lawyers for the Applicant, Jaguar Mining Inc.

DOCSTOR: 2885569/6

Exhibit B

1.00383.85494.1

14.474

10 Court File No. CV-13-10383-00CL This is Family ONTARIO ---FSf0tford to in the SUPERIOR COURT OF JUSTICE COMMERCIAL LIST strato before marithis FEBRUARY, 201 IN THE MATTER OF THE COMPANIES' CREDITORS ARRANGEMENT ACT, R.S.C. 1985, c. C-36, AS AMENDED AND IN THE MATTER OF A PLAN OF COMPROMISE OR ARRANGEMENT OF JAGUAR MINING INC. Applicant

(1944至1958年)。

13109553

AFFIDAVIT OF T. DOUGLAS WILLOCK (sworn February 2, 2014)

I, T. DOUGLAS WILLOCK, of the City of Toronto, MAKE OATH AND SAY:

1. I am the Chief Financial Officer of Jaguar Mining Inc. ("Jaguar" or the "Applicant"). I have held that position since January 10, 2013. As such, I have personal knowledge of the matters to which I hereinafter depose, except where otherwise stated. In preparing this affidavit I have also consulted, where necessary, with other members of Jaguar's management team or the management teams of its wholly-owned subsidiaries (together with Jaguar, the "Jaguar Group"). Where I have relied upon other sources of information, I have stated the source of that information and believe such information to be true.

2. I swear this affidavit in support of a motion by Jaguar for an Order (the ***Sanction Order**") sanctioning Jaguar's plan of compromise and arrangement dated

December 23, 2013 (as amended and restated on January 31, 2014, and as may be further amended, restated, modified or supplemented in accordance with its terms (the "**Plan**")) pursuant to the *Companies' Creditors Arrangement Act*, R.S.C. 1985, c. C-36, as amended (the "**CCAA**"). This affidavit supplements and updates information contained in: (i) the Affidavit of David M. Petroff, sworn December 23, 2013, filed in support of Jaguar's initial application in these proceedings (the "**Initial Petroff Affidavit**", a copy of which, without exhibits, is attached as Exhibit "A" hereto), and the reports filed by FTI Consulting Canada Inc. in its capacity as the monitor appointed in respect of these proceedings (the "**Monitor**").

0.0600.000-244628

11

enserver and the second of the

3. Capitalized terms used but not otherwise defined herein have the meanings given to them in the Plan.

4. Jaguar believes that the Plan is the best available alternative in the circumstances and that its implementation on an expedited basis is vital to Jaguar and its stakeholders. Unless the Plan is implemented on an expedited timeline, liquidation would appear to be the likely remaining alternative, which would be detrimental to all stakeholders.

5. The Plan and the relief sought under the Sanction Order are supported by the Ad Hoc Committee of Noteholders, representing over 80% of the Affected Creditor Class, and the Monitor. The Plan received the support of 100% of the Affected Unsecured Creditors voting at the Meeting.

6. References to "\$" or "dollars" herein are to U.S. dollars, for ease of reference. The revenues of the Jaguar Group are in U.S. dollars and Brazilian reais. The

expenditures of the Jaguar Group are in U.S. dollars, Brazillan reais, and Canadian dollars.

I. BACKGROUND

7. As stated in the Initial Petroff Affidavit, the principal objective of these proceedings is to effect a recapitalization and financing transaction (the "**Recapitalization**") on an expedited basis to provide a stronger financial foundation for the Jaguar Group going forward and additional liquidity to allow the Jaguar Group to continue to work towards its operational and financial goals. In particular, the Recapitalization would result in a reduction of over \$268 million of debt and new liquidity upon exit of approximately \$50 million.

8. Jaguar must continue moving forward with the Recapitalization as efficiently and expeditiously as possible due to an imminent liquidity crisis.

II. JAGUAR'S CASH POSITION

9. As noted in the Initial Petroff Affidavit, the Jaguar Group has immediate needs for additional funds. The cash available to the Jaguar Group is not sufficient to cover its costs for any material length of time beyond February 2014 at current gold prices. This is consistent with cash flow forecasts filed at the time of Jaguar's initial application on December 23, 2013, and actual expenditures of Jaguar have not differed significantly from those cash flow forecasts, except with respect to certain timing issues relating to receipt of professional accounts.

10. In its Second Report to the Court dated January 24, 2014 (the "Monitor's Second Report"), the Monitor commented on Jaguar's actual receipts and

disbursements as well as its prior cash flow forecasts. In this report, the Monitor noted that certain variances existed when the actual net cash flow for the period from January 4, 2014 to January 17, 2014 was compared against the underlying cash flow forecast, but that such variances were primarily attributed to certain timing issues related to the receipt of professional accounts. Assuming the Plan is implemented on the timeline proposed, Jaguar's liquidity requirements through to the proposed implementation of the Plan are expected to be met by existing cash available to Jaguar.

13

11. Jaguar is the public parent corporation of other corporations in the Jaguar Group that carry on active gold mining and exploration in Brazil. I am in regular contact with the senior management of those corporations, and can confirm that no significant additional cash resources are available to provide further funding from the subsidiary level at current gold prices.

III. CLAIMS PROCEDURE

12. I am advised by the Monitor that, aside from the claims in respect of the Notes, the Monitor received 14 proofs of claim at or before 5:00 p.m. on January 22, 2014, being the claims bar date established by the Claims Procedure Order.

13. I am advised by the Monitor that, in aggregate, those claims were less than \$1,000,000.

14. I am further advised by the Monitor that the Monitor has responded to claimants who submitted proofs of claim in accordance with the Claims Procedure Order.

V. CREDITORS' MEETING AND SANCTION HEARING

15. To the best of my knowledge, Jaguar and the Monitor have complied with all of the notification and service requirements contained in the Initial Order, the Meeting Order and the Claims Procedure Order. In the Monitor's Second Report, the Monitor has also noted (at paragraph 19) that nothing has come to its attention that would suggest that Jaguar has not been in compliance with those Orders and/or the CCAA generally.

ANTERNE MARKAGE SALESALESE DER STER SERVER AND AND ANTER SERVER AND ANTER SERVER AND ANTER SERVER SERVER AND AN

16. The meeting of Affected Unsecured Creditors, initially scheduled for January 28, 2014, was postponed to January 31, 2014. In accordance with paragraph 46 of the Meeting Order dated December 23, 2013 (the "Meeting Order"), the Monitor provided notice to the Applicant that the Monitor, in consultation with the Applicant and the Majority Consenting Noteholders, decided to postpone the Meeting to Friday, January 31, 2014 at 10:00 a.m. Jaguar issued a press release on January 24, 2014 announcing the January 31, 2014 date for the Meeting and the corresponding extended voting and election deadlines. In accordance with the Meeting Order, the Monitor also provided written notice of the postponement of the Meeting and extension of the corresponding voting and election deadlines on January 25, 2014 by way of email to the service list, and the Monitor updated the Monitor's website and posted a copy of the press release issued by Jaguar in this regard.

17. The Monitor adjourned the Meeting on Friday January 31, 2014 at 10:00 a.m., initially to 2:00 p.m., and then subsequently to 4:00 p.m. to allow discussions regarding the Plan to continue. The Monitor provided notice of these adjournments by email to the Service List.

14

- (ABBERGERE), LORDORA, L. SANGER

The Meeting Order contemplated (in paragraph 54) that the motion seeking the · 18. Sanction Order would be brought on January 30, 2014, or as soon thereafter as the matter can be heard. The hearing date was set for February 3, 2014 rather than the original date anticipated (January 30, 2014) due to the lack of available court time on January 30, 2014, or on January 31, 2014. Notice that the hearing date for the Sanction Order would be February 3, 2014 was included in the Monitor's Second Report and also given by the Monitor by way of e-mail to the service list on January 27, 2014. I am advised by Evan Cobb of Norton Rose Fulbright Canada LLP, that the motion seeking the Sanction Order is now expected to be heard on February 6, 2014.

Jaguar instructed its counsel, Norton Rose Fulbright Canada LLP, to ensure that 19. a representative was present at the places and times initially proposed for the Meeting and the hearing of the motion seeking the Sanction Order so that such representative could personally meet anyone that might possibly attend at those times, and advise them of the revised dates and times. I am advised by Evan Cobb of Norton Rose Fulbright Canada LLP that no one attended at the location of the Meeting or the hearing for the Sanction Order at the initially proposed times.

VI. AMENDMENTS TO THE PLAN

Jaguar has prepared a supplement (the "Plan Supplement") to the Plan of 20, Compromise and Arrangement dated December 23, 2013 (the "Plan as Originally Filed") that was included in the Initial Petroff Affidavit, filed with this Court, and provided to Affected Unsecured Creditors in accordance with the Meeting Order. A copy of the Plan Supplement is attached hereto as Exhibit "B".

and the second secon

The Plan Supplement includes an amended and restated version of the Plan, 21. which replaces the Plan as Originally Filed, and supplements the Plan and the Information Package with, among other things: (A) a draft copy of the Articles of Reorganization, (B) the form of Monitor's certificate to be delivered pursuant to Section 12.6 of the Plan, (C) a summary of the escrow arrangements established to administer the Disputed Distribution Claims Reserve; (D) a summary of the escrow arrangements established with the Escrow Agent; and (E) a draft of the proposed Sanction Order.

Amendments to the Plan include the addition to the definition of "Excluded 22. Claims" under the Plan of (a) claims against the Applicant or its Directors or Officers as may be agreed upon by the Applicant and the Majority Consenting Noteholders (the "Agreed Excluded Litigation Claims"); and (b) a claim that Canada Revenue Agency filed under the Claims Procedure Order relating to certain limited GST/HST amounts that Canada Revenue Agency currently asserts are unpaid and held in trust by Jaguar pending payment. Under the amendments to the Plan, recovery in respect of the Agreed Excluded Litigation Claims (if any such claims are identified and agreed to between Jaguar and the Majority Consenting Noteholders) are limited solely to proceeds from applicable insurance policies and the holders of such Agreed Excluded Litigation Claims shall have no right to seek or obtain recovery on such claims from any other Persons, including the Applicant, the Subsidiaries, or any Directors or Officers.

Further, under the Plan as Originally Filed, recovery in respect of Section 5.1(2) 23, Directors/Officer Claims against the Named Directors and Officers were directed to proceeds from applicable insurance policies. The amendments to the Plan direct recovery in respect of Section 5.1(2) Director/Officer Claims against any Director or Officer to proceeds from applicable insurance policies.

17

The amendments to the Plan provide that nothing in the Plan will prejudice, 24. compromise, release or otherwise affect any right or defence of any insured, or of any insurer under any insurance policy, in respect of an Agreed Excluded Litigation Claim or a Section 5.1(2) Director/Officer Claim.

Other changes to the Plan are of an administrative nature required to better give 25. effect to the implementation of the Plan and/or to cure any errors, omissions or ambiguities and are not materially adverse to the financial or economic interests of the Affected Unsecured Creditors under the Plan.

The amended and restated version of the Plan was provided to the Monitor for: 26. (i) distribution in accordance with paragraph 9 of the Meeting Order at the Meeting; (ii) communication to Affected Unsecured Creditors at the Meeting on January 31, 2014; and (iii) posting on the Monitor's website. I am advised by Norton Rose Fulbright Canada LLP that the Plan Supplement will also be sent to the service list and filed with the Court.

VII. 2012 Litigation

As described in the Initial Petroff Affidavit, litigation was commenced on March 27. 27, 2012 by Daniel Titcomb, the former chief executive officer of Jaguar, and certain other associated parties (the "2012 Litigation Plaintiffs"), which lawsuit is currently proceeding in the United States Federal Court. This lawsuit primarily alleges claims in respect of the equity interests of the 2012 Litigation Plaintiffs in Jaguar and also alleges certain claims related to the previous employment of Mr. Titcomb at Jaguar.

The 2012 Litigation Plaintiffs did not file a proof of claim prior to the claims bar 28. date established by the Claims Procedure Order.

VIII BERGENBERGER

As previously indicated in the Initial Petroff Affidavit, Jaguar and its Board of 29, Directors believe this lawsuit to be without merit. However, the lawsuit has been a continuing distraction for Jaguar and its Board of Directors, including those current directors who are defendants in this lawsuit despite having had no involvement with the board or management of Jaguar during the time period to which the lawsuit relates. Jaguar believes that it is important to resolve the distraction created by these legacy issues as part of this CCAA process.

At the comeback hearing in these proceedings on January 14, 2014, the 2012 30. Litigation Plaintiffs reserved their rights to argue that the claims of the 2012 Litigation Plaintiffs cannot be compromised by the Plan. Jaguar disagrees with any argument that the claims of the 2012 Litigation Plaintiffs (which claims are primarily in respect of equity interests) cannot be compromised by the Plan and believes that such claims are not valid or proper claims against Jaguar. The matter has not been resolved at this time.

29. The amended and restated Plan includes a framework that could facilitate the Implementation of a potential resolution of this matter under the Plan, if the Applicant and the Majority Consenting Noteholders agree to any such resolution. Otherwise, the Applicant would take the position that the claims of the 2012 Litigation Plaintiffs are Equity Claims, irrevocably compromised, released and barred under the Plan, or otherwise not valid or proper claims against Jaguar.

VIII. **RESULTS OF THE MEETING**

The vote on the amended and restated Plan, as presented at the Meeting, was 31. approved by 100% of the Affected Unsecured Creditors that voted, in person or by

19

proxy. The aggregate dollar value of claims voted at the meeting was in excess of \$225 million, representing over 80% of the Affected Creditor Class.

REQUEST FOR COURT APPROVAL OF THE PLAN IX.

Jaguar cannot sustain and service its current high levels of debt and continues 32. to face an imminent liquidity crisis. Absent the approval of a transaction such as the Plan and expedited implementation, Jaguar will not have sufficient liquidity to continue operations and liquidation would appear to be the likely remaining alternative, with holders of Notes experiencing a significant shortfall in the amounts owed to them. A liquidation would be detrimental to all stakeholders with an economic interest in Jaguar or the Jaguar Group as a whole, including the hundreds of employees of the Jaguar Group.

The Plan is described in detail in the Initial Petroff Affidavit and the Monitor's 33. Second Report. In general terms, the Plan provides for:

- the conversion of approximately \$268.5 million of Notes, as well as (a) certain other proven unsecured claims, into equity;
- a reduction of pro-forma funded debt from \$323 million as at September (b) 30, 2013 to \$54 million upon the implementation of the Plan;
- a reduction of projected annual cash interest payments by \$13.1 million; (C)
- an investment of approximately \$50 million in new equity raised by way (d) of a backstopped share offering, the proceeds of which will be available for use in Jaguar's operations;
- the retention of Existing Shares by Existing Shareholders, which will (e) account for 0.9% of the common shares of Jaguar upon completion of the Share Offering; and

34. Jaguar continues to believe that the Plan is the best way to reduce Jaguar's debt levels, increase liquidity for the Jaguar Group's operations and allow the Jaguar Group to make certain necessary capital investments and accelerate operational improvements.

35. The Plan also provides for certain releases and injunctions in favour of the Released Parties (which include the Applicant, the Subsidiaries, the Monitor and their advisors and certain associated parties), the Named Directors and Officers (which include the current directors and officers of Jaguar and such other directors and officers as agreed to by the Majority Consenting Noteholders prior to the Meeting); and the Noteholder Released Parties (which include the Noteholders, the Trustees and their advisors and certain associated parties), subject to carve outs for, among other things, fraud, wilful misconduct, and claims that cannot be compromised under Section 5.1(2) of the CCAA,

36. I am advised by Evan Cobb of Norton Rose Fulbright Canada LLP that on January 31, 2014, prior to the Meeting, an email was received from counsel to the Majority Consenting Noteholders confirming agreement that the Named Directors and Officers are: the current and former directors and officers of Jaguar except any current or former directors or officers of Jaguar who are plaintiffs in the United States District Court for the District of New Hampshire Civil Action No.: 1:13-cv-00428-JL or who have commenced any proceedings against Jaguar or any of its affiliates.

37. The releases and injunctions in the Plan were negotiated as part of the overall framework of compromises in the Plan and Jaguar believes the releases and injunctions

-21

in favour of the released parties are necessary and facilitate the successful completion of the Plan and the Recapitalization.

Jaguar believes that the releases contained in the Plan are all rationally 38. connected to the Plan, given, among other things, the Subsidiaries are the operating entities of the Jaguar Group and are the entities in which most of the Jaguar Group's value resides; directors and officers have overseen the Applicant's strategic review process and provided guidance and stability throughout the restructuring process; the advisors have been retained in a role specifically to assist with the development and implementation of the Plan, which could not be successfully achieved without these parties; and the Noteholders have agreed to compromise their claims and contributed to the development and supported the Recapitalization and the Plan.

As discussed in the Initial Petroff Affidavit and the Monitor's Second Report, 39, Jaguar has, with the assistance of its advisors, considered numerous strategic alternatives. The efforts of Jaguar and its advisors have shown that a comprehensive restructuring plan involving a debt for equity exchange and an investment of new money is the best available alternative.

40. Implementation of the Plan will allow Jaguar to address its current financial issues and to work towards its operational and financial goals for the benefit of its stakeholders.

Jaguar believes the Plan is fair and reasonable to affected parties and that the 41. approval of the Plan by the Court is justified and appropriate.

(4) 这时候这些影响。

CONCLUSION Х.

Throughout the course of the CCAA proceeding, Jaguar has acted in good faith 42. and with due diligence. The Company has complied with the requirements of the CCAA Jaguar has achieved strong support for the and the Orders of this Court. Recapitalization and Plan.

Jaguar and its board of directors believe that the Plan is the best available 43. alternative in the circumstances following the consideration of available alternatives. Absent the Plan, liquidation would appear to be the likely remaining alternative, which would be detrimental to all stakeholders. The implementation of the Plan on an expedited basis is vital to Jaguar and its stakeholders.

I understand that the Plan and the relief sought under the Sanction Order are 44. supported by the Ad Hoc Committee of Noteholders, representing over 80% of the Affected Creditor Class, and the Monitor. The Plan was approved at the Meeting by all Affected Unsecured Creditors who voted.

Accordingly, Jaguar respectfully requests that this Court approve the Plan and 45. grant the Sanction Order.

SWORN BEFORE ME at the City of Toronto, on February 2, 2014.

Commissioner for Taking Affidavits

T, Douglas Willock

Exhibit C

1.82

12612-620

469000 biteberaris si

	· .
Court File No.	CV-13-10383-00CL
	· · · · · ·



ONTARIO SUPERIOR COURT OF JUSTICE COMMERCIAL LIST

efficient of The Dawgers WILLS

THE HONOURABLE REGIONAL SENIOR

JUSTICE MORAWETZ

IN THE MATTER OF THE COMPANIES' CREDITORS ARRANGEMENT ACT, R.S.C. 1985, c. C-36, AS AMENDED

THURSDAY, THE 6TH

DAY OF FEBRUARY, 2014.

AND IN THE MATTER OF A PLAN OF COMPROMISE OR ARRANGEMENT OF JAGUAR MINING INC.

Applicant

ORDER (Plan Sanction)

THIS MOTION made by Jaguar Mining Inc. (the "Applicant") for an Order (the "Sanction Order") approving and sanctioning the amended and restated plan of compromise and arrangement dated January 31, 2014 (and as It may be further amended, restated, modified or supplemented from time to time in accordance with its terms) (the "Plan"), as approved by the Affected Unsecured Creditors of the Applicant on January 31, 2014, and which Plan is attached as Schedule "A" to this Sanction Order, was heard this day at 330 University Avenue, Toronto, Ontario.

ON READING the within Notice of Motion, the Affidavit of T. Douglas Willock sworn February 2, 2014 and the Affidavit of T. Douglas Willock sworn February 5, 2014, including the Exhibits thereto, the Third Report of FTI Consulting Canada Inc., in its capacity as Monitor (in such capacity, the "Monitor"), dated February 3, 2014, (the "Third Report"), the Second

DOCSTOR; 2902739\4C

eI:

Report of the Monitor dated January 24, 2014 (the **"Second Report"**) and upon hearing the submissions of counsel for the Applicant, the Monitor, the Ad Hoc Committee (as defined in the Plan), Global Resource Fund, Daniel Titcomb et al. and such other interested parties as were present, no one else appearing although duly served as appears from the affidavit of service of Evan Cobb sworn Februrary 4, 2014 and February 5, 2014, and upon being advised by counsel to the Applicant prior to this motion that the Sanction Order will be relied upon by the Applicant as an approval of the Plan for the purpose of relying on the exemption from the registration requirements of the *United States Securities Act of 1933*, as amended, pursuant to section 3(a)(10) thereof for the issuance of the Unsecured Creditor Common Shares and the Early Consent Shares to the extent they may be deemed to be securities,

DEFINITIONS

1. **THIS COURT ORDERS** that all capitalized terms not otherwise defined herein shall be as defined in the Plan and in the Meeting Order granted in this proceeding on December 23, 2013 (the "**Meeting Order**"), as applicable.

SERVICE

2. **THIS COURT ORDERS** that the time for service of the Notice of Motion and the Motion Record in support of this Motion and the Third Report be and is hereby abridged and validated, such that this Motion is properly returnable today and that any further service of the Notice of Motion, the Motion Record or the Third Report is hereby dispensed with.

3. **THIS COURT ORDERS AND DECLARES** that there has been good and sufficient service, delivery and notice of the Plan, the Meeting Order and the Information Package to all Persons upon which notice, service and delivery were required, and that the Meeting was duly conducted in conformity with the *Companies' Creditors Arrangement Act* (the "**CCAA**") and all

other Orders of this Court in this proceeding (the "CCAA Proceeding").

SANCTION OF THE PLAN

4. THIS COURT ORDERS AND DECLARES that:

- a) the relevant class of creditors of the Applicant for the purposes of voting to approve the Plan is the Affected Creditor Class;
- b) the Plan has been approved by the Required Majority of Affected Unsecured Creditors,
 all in conformity with the CCAA and the terms of the Meeting Order;
- c) the Court is satisfied that the Applicant has acted, and is acting, in good faith and with due diligence, and has complied with the provisions of the CCAA and the Orders of this Court made in the CCAA Proceedings in all respects;
- d) the Court is satisfied that the Applicant has not done nor has it purported to do anything that is not authorized by the CCAA; and
- e) the Plan, all terms and conditions thereof, and the matters and the transactions contemplated thereby, are fair and reasonable to the parties affected.

5. THIS COURT ORDERS AND DECLARES that the Plan (including, without limitation, the transactions, arrangements, reorganizations, assignments, cancellations, compromises, settlements, extinguishments, discharges, injunctions and releases set out therein) is hereby sanctioned and approved pursuant to Section 6 of the CCAA.

PLAN IMPLEMENTATION

6. THIS COURT ORDERS that on the Implementation Date, the Plan (including, without limitation, the transactions, arrangements, reorganizations, assignments, cancellations, compromises, settlements, extinguishments, discharges, injunctions and releases set out therein) shall be, and shall be deemed to be, implemented, binding and effective in accordance with the provisions of the Plan as of the Implementation Date at the Implementation Time, or at such other time or times and in the manner set forth in the Plan, and shall enure to the benefit of and shall be binding on the Applicant, the Affected Unsecured Creditors, all Existing Equity Holders, all holders of Equity Claims, the Released Parties, the Noteholder Released Parties, the Director/Officer Claims, all holders of Director/Officer Indemnity Claims, all holders of Director/Officer Claims, all holders of Released Claims and all holders of Noteholder Released Claims and all other Persons named or referred to in, affected by, or subject to the Plan, including, without limitation, their respective heirs, administrators, executors, legal personal representatives, successors and assigns, as provided for in the Plan and this Sanction Order.

7. **THIS COURT ORDERS** that the Applicant and the Monitor, as the case may be, are hereby authorized and directed to take all steps and actions necessary or appropriate to implement the Plan in accordance with and subject to its terms and conditions, and enter into, execute, deliver, complete, implement and consummate all of the steps, transactions, distributions, deliveries, allocations, instruments and agreements contemplated by, and subject to the terms and conditions of, the Plan, and all such steps and actions are hereby approved. Neither the Applicant nor the Monitor shall incur any liability as a result of acting in accordance with the terms of the Plan and this Sanction Order.

DOCSTOR; 2902739\4C

8. THIS COURT ORDERS that upon the satisfaction or waiver, as applicable, of the conditions set out in Section 12.3 of the Plan in accordance with the terms of the Plan, as confirmed by the Applicant (or counsel on its behalf) and Goodmans LLP on behalf of the Majority Consenting Noteholders and the Majority Backstop Parties, to the Monitor in writing, the Monitor is authorized and directed to deliver to the Applicant (or counsel on its behalf) and Goodmans LLP a certificate, substantially in the form attached as Schedule "B" hereto (the "Monitor's Certificate"), signed by the Monitor, certifying that all conditions precedent set out in Section 12.3 have been satisfied or waived and that the Implementation Date has occurred. The Monitor shall file the Monitor's Certificate with this Court as soon as reasonably practicable.

9. **THIS COURT ORDERS** that the steps to be taken and the transactions, arrangements, reorganizations, assignments, cancellations, compromises, settlements, extinguishments, discharges, injunctions and releases to be effected on the Implementation Date are and shall be deemed to occur and be effected in the sequential order and at the times contemplated by Section 7.4 of the Plan, without any further act or formality, on the Implementation Date beginning at the Implementation Time.

10. THIS COURT ORDERS AND DECLARES that the Applicant, the Monitor, the Majority Consenting Noteholders and the Majority Backstop Parties are hereby authorized and empowered to exercise all consent and approval rights provided for in the Plan in the manner set forth in the Plan, whether prior to or after the Implementation Date.

11. THIS COURT ORDERS that the Applicant, the Monitor, the Trustees, DTC, the Transfer Agent, the Escrow Agent and any other Person required to make any distributions, deliveries or allocations or take any steps or actions related thereto pursuant to the Plan are hereby authorized and directed to complete such distributions, deliveries or allocations and to

DOCSTOR: 2902739\4C

take any such related steps or actions, as the case may be, in accordance with the terms of the Plan, and such distributions, deliveries and allocations, and steps and actions related thereto, are hereby approved.

12. **THIS COURT ORDERS** that, subject to the payment of any amounts secured by the Charges (as such term is defined in the Initial Order) that remain owing on the Implementation Date, if any, each of the Charges shall be terminated, discharged and released on the Implementation Date.

EFFECT OF PLAN IMPLEMENTATION

13. THIS COURT ORDERS that subject to the performance by the Applicant of its obligations under the Plan, and except to the extent expressiv contemplated by the Plan or this Sanction Order, all obligations or agreements to which the Applicant is party immediately prior to the Implementation Date will be and remain in full force and effect as at the Implementation Date, unamended except as they may have been amended by agreement of the parties subsequent to the Filing Date in accordance with the Plan, and no Person who is a party to any such obligations or agreements shall, following the Implementation Date, accelerate, terminate, rescind, refuse to renew, refuse to perform or otherwise disclaim or repudiate its obligations thereunder, or enforce or exercise (or purport to enforce or exercise) any right or remedy (including any right of set-off, option, dilution or other remedy) or make any demand under or in respect of any such obligation or agreement, by reason of:

- (a) any defaults or events of default arising as a result of the financial condition or insolvency of the Applicant on or prior to the Implementation Date;
- (b) the fact that the Applicant has sought or obtained relief under the CCAA or that the Plan has been implemented by the Applicant;

- (c) any changes in share ownership of the Applicant arising from implementation of the Plan (except in respect of existing, written senior officer and employee employment agreements of Persons who remain senior officers and employees of Jaguar as of the Implementation Date and any payments due under such agreements, which may only be waived by the senior officers and employees who are parties to such agreements);
- (d) the effect on the Applicant of the completion of any of the transactions contemplated by the Plan;
- (e) any compromises, settlements, restructurings, recapitalizations, reorganizations or arrangements effected pursuant to the Plan; or
- (f) any other event(s) which occurred on or prior to the Implementation Date which would have entitled any Person to enforce rights and remedies, subject to any express provisions to the contrary in any agreements entered into with the Applicant after the Filing Date.

For greater certainty, nothing in this paragraph shall waive, compromise or discharge any obligations of the Applicant in respect of any Excluded Claim.

14. THIS COURT ORDERS that from and after the Implementation Date, all Persons shall be deemed to have waived any and all defaults of the Applicant then existing or previously committed by the Applicant or caused by the Applicant or any of the provisions of the Plan or this Sanction Order or non-compliance with any covenant, warranty, representation, term, provision, condition or obligation, express or implied, in any contract, agreement, mortgage, security agreement, indenture, trust Indenture, loan agreement, commitment letter, agreement for sale, real property lease, personal property lease or other agreement, written or oral, and any amendments or supplements thereto, existing between such Person and the Applicant. Any and all notices of default, acceleration of payments and demands for payments under any instrument, or other notices, including without limitation, any notices of intention to proceed to

enforce security, arising from any of such aforesaid defaults shall be deemed to have been rescinded and withdrawn.

15. THIS COURT ORDERS that, as of the Implementation Date, each Affected Unsecured Creditor, each holder of a Director/Officer Indemnity Claim, each holder of a Director/Officer Claim, each holder of an Equity Claim and any person having any other Released Claim shall be deemed to have consented and agreed to all of the provisions of the Plan in their entirety and, in particular, each Affected Unsecured Creditor, each holder of a Director/Officer Indemnity Claim, each holder of a Director/Officer Claim, each holder of an Equity Claim and any person having any other Released Claim shall be deemed:

- (a) to have granted, executed and delivered to the Monitor and the Applicant all consents, releases, assignments, waivers or agreements, statutory or otherwise, required to implement and carry out the Plan in its entirety; and
- (b) to have agreed that if there is any conflict between the provisions, express or implied, of any agreement or other arrangement, written or oral, existing between such Affected Unsecured Creditor, holder of a Director/Officer Indemnity Claim, holder of a Director/Officer Claim, holder of an Equity Claim and any person having any other Released Claim and the Applicant as of the Implementation Date and the provisions of the Plan, the provisions of the Plan take precedence and priority, and the provisions of such agreement or other arrangements shall be deemed to be amended accordingly.

16. **THIS COURT ORDERS** that pursuant to Section 6(2) of the CCAA, the articles of the Applicant shall be amended on the Implementation Date in accordance with the Articles of Reorganization.

DOCSTOR: 2902739\4C

17. **THIS COURT ORDERS** that (i) in accordance with the Articles of Reorganization, substantially in the form of Schedule "C" hereto, any fractional Common Shares immediately following the consolidation of Common Shares pursuant to section Section 7.4(a) of the Plan shall be cancelled without any liability, payment or other compensation in respect thereof; and (ii) the Rights, Shareholder Rights Plan, Existing Share Options, Stock Option Plan (and for greater certainty, not including any Existing Common Shares that remain issued and outstanding immediately following the cancelled without any liability, payment or other compensation in respect to Section 7.4(a) of the Plan) shall be cancelled without any liability, payment or other compensation in respect thereof.

18. **THIS COURT ORDERS** that the New Common Shares shall be deemed to be issued and outstanding as fully-paid and non-assessable shares in the capital of the Applicant, on the Implementation Date and at the time specified in Section 7.4 of the Plan.

19. **THIS COURT ORDERS** that, on the Implementation Date, following completion of the steps in the sequence set forth in Section 7.4 of the Plan, all debentures, notes, certificates, agreements, invoices and other Instruments evidencing Affected Unsecured Claims (including for greater certainty the Notes and the Indentures) shall not entitle the holder thereof to any compensation or participation and shall be and are hereby deemed to be cancelled and shall be and are hereby deemed to be null and void, and the obligations of the Applicant thereunder or in any way related thereto shall be satisfied and discharged except to the extent expressly set forth in section 6.07 of the Indentures with respect to the Trustees' claims, which section 0.07 of the Indentures shall remain in effect until two months following the Implementation Date or such later date agreed to by the Applicant, the Monitor, the Trustees and the Majority Consenting Noteholders.

DOCSTOR: 2902739\4C

RELEASES AND INJUNCTIONS

20. **THIS COURT ORDERS** that, subject to paragraph 21 of this Sanction Order, on the Implementation Date, in accordance with section 11.1 of the Plan and the sequence set forth in section 7.4 of the Plan, the Released Parties, the Named Directors and Officers and the Noteholder Released Parties shall be released and discharged from any and all Released Claims and any and all Noteholder Released Claims, as applicable, and any and all Released Claims and Noteholder Released Claims shall be fully, finally and irrevocably waived, discharged, released, cancelled and barred as against the Released Parties, the Named Directors and Officers and the Noteholder Released Laims and Officers and the Noteholder Released Claims shall be fully, finally and irrevocably waived, discharged, released, cancelled and barred as against the Released Parties, the Named Directors and Officers and the Noteholder Released Parties, as applicable, all to the fullest extent permitted by Applicable Law.

21. **THIS COURT ORDERS** that, notwithstanding any other provision of this Sanction Order, Continuing Other Director/Officer Claims and Non-Released Director/Officer Claims and, for greater certainty, Section 5.1(2) Director/Officer Claims, Agreed Excluded Director/Officer Litigation Claims and Agreed Excluded Jaguar Litigation Claims shall not be compromised, released, discharged, cancelled or barred by this Sanction Order or the Plan, provided that from and after the Implementation Date: (i) any Person having, or claiming any entitlement or compensation relating to, a Section 5.1(2) Director/Officer Claim or an Agreed Excluded Director/Officer Litigation Claim will be irrevocably limited to recovery in respect of such Section 5.1(2) Director/Officer Claim or Agreed Excluded Director/Officer Litigation Claims solely from the proceeds of applicable Director/Officer Insurance Policies and Persons with Section 5.1(2) Director/Officer Claims and Agreed Excluded Director/Officer Litigation Claims will have no right to, and shall not, directly or indirectly, make any claims or seek any recoveries from the Applicant, any of the Subsidiaries, any of the Directors or Officers, or any other Released Party or Noteholder Released Party, other than enforcing such Person's rights

59

to be paid by the applicable insurer(s) the proceeds of the applicable Director/Officer Insurance Policies; and (ii) any Person having, or claiming any entitlement or compensation relating to, an Agreed Excluded Jaguar Litigation Claim will be irrevocably limited to recovery in respect of such Agreed Excluded Jaguar Litigation Claim solely from the proceeds of applicable Jaguar Insurance Policies and Persons with Agreed Excluded Jaguar Litigation Claims will have no right to, and shall not, directly or indirectly, make any claims or seek any recoveries from the Applicant, any of the Subsidiaries, any of the Directors or Officers, or any other Released Party or Noteholder Released Party, other than enforcing such Person's rights to be paid by the applicable insurer(s) the proceeds of the applicable Jaguar Insurance Policies.

THIS COURT ORDERS that all Persons shall be permanently and forever barred, 22. estopped, stayed and enjoined, from and after the Implementation Time, with respect to any and all Released Claims and Noteholder Released Claims, from: (i) commencing, conducting or continuing in any manner, directly or indirectly, any action, suits, demands or other proceedings of any nature or kind whatsoever (including, without limitation, any proceeding in a judicial, arbitral, administrative or other forum) against the Released Parties, the Named Directors and Officers and the Noteholder Released Partles; (ii) enforcing, levying, attaching, collecting or otherwise recovering or enforcing by any manner or means, directly or indirectly, any judgment, award, decree or order against the Released Parties, the Named Directors and Officers and the Noteholder Released Parties or their respective property; (iii) commencing, conducting or continuing in any manner, directly or indirectly, any action, suits or demands, including without limitation by way of contribution or indemnity or other relief, in common law, or in equity, or for breach of trust or breach of fiduciary duty or under the provisions of any statute or regulation, or other proceedings of any nature or kind whatsoever (including, without limitation, any proceeding in a judicial, arbitral, administrative or other forum) against any

ĥА

Person who makes such a claim or might reasonably be expected to make such a claim, in any manner or forum, against one or more of the Released Parties, the Named Directors and Officers and Noteholder Released Parties; (iv) creating, perfecting, asserting or otherwise enforcing, directly or indirectly, any lien or encumbrance of any kind against the Released Parties, the Named Directors and Officers and the Noteholder Released Parties or their respective property; or (iv) taking any actions to interfere with the implementation or consummation of the Plan; provided, however, that the foregoing shall not apply to the enforcement of any obligations under the Plan.

23. THIS COURT ORDERS that paragraph 22 shall apply to Section 5.1(2) Director/Officer Claims, Agreed Excluded Director/Officer Litigation Claims and Agreed Excluded Jaguar Litigation Claims in the same manner as Released Claims, except to the extent that the rights of a holder of such Section 5.1(2) Director/Officer Claims, Agreed Excluded Director/Officer Litigation Claims and/or Agreed Excluded Jaguar Litigation Claims to enforce such rights against an insurer in respect of a Directors/Officer Insurance Policy and/or a Jaguar Insurance Policy, as applicable; are expressly preserved pursuant Section 11.1(a)(iii) and/or Section 11.1(b)(i) of the Plan.

24. **THIS COURT ORDERS** that nothing in this Sanction Order prejudices, compromises, releases or otherwise affects any right or defence of (i) any insurer in respect of a Director/Officer Insurance Policy or a Jaguar Insurance Policy, or (ii) any insured in respect of a Section 5.1(2) Director/Officer Claim, an Agreed Excluded Director/Officer Litigation Claim or an Agreed Excluded Jaguar Litigation Claim.

INITIAL CCAA ORDER AND OTHER ORDERS

25. THIS COURT ORDERS that:

- (a) other than as expressly set out herein, the provisions of the Initial Order shall terminate, including the Stay Period (as defined in the Initial Order), on the Implementation Date except to the extent of the protections granted therein in favour of the Monitor; and
- (b) all other Orders made in the CCAA Proceeding shall continue in full force and effect in accordance with their respective terms, except to the extent that such Orders are varied by or are inconsistent with this Sanction Order or any further Order of this Court.

THE MONITOR

26. **THIS COURT ORDERS** that the activities and conduct of the Monitor in relation to the Applicant, the CCAA Proceedings, and in conducting and administering the Meeting on January 31, 2014 (as more particularly described in the Third Report) be and are hereby ratified and approved.

27. **THIS COURT ORDERS** that the Pre-Filing Report of the Monitor dated December 23, 2013 (the **"Pre-Filing Report"**), the First Report of the Monitor dated January 13, 2014 (the **"First Report"**), and the Second Report and the conduct and activities of the Monitor as described therein are hereby approved.

28. **THIS COURT ORDERS** that the fees and disbursements of the Monitor and Osler, Hoskin & Harcourt LLP, as counsel to the Monitor, as described in the Third Report be and are hereby approved.

29. THIS COURT ORDERS AND DECLARES that the Monitor has satisfied all of its obligations up to and including the date of this Sanction Order, and that: (i) in carrying out the terms of this Sanction Order and the Plan and in performing its duties as Monitor in the CCAA Proceedings, the Monitor shall have all the protections given to it by the CCAA, the Initial Order, the Meeting Order and the Claims Procedure Order, and as an officer of the Court, including the stay of proceedings in its favour; (ii) the Monitor shall incur no liability or obligation for any act or omission as a result of carrying out the provisions of this Sanction Order and the Plan and in performing its duties as Monitor in the CCAA Proceedings, save and except for any gross negligence or wilful misconduct on its part; (iii) the Monitor shall be entitled to rely on the books and records of the Applicant and any information provided by the Applicant without independent investigation; and (iv) the Monitor shall not be liable for any claims or damages resulting from any errors or omissions in such books, records or information, or with respect to any such information disclosed to or provided by the Monitor, including with respect to reliance thereon by any Person.

30. **THIS COURT ORDERS** that any claims against the Monitor in connection with the performance of its duties as Monitor are hereby released, stayed, extinguished and forever barred and the Monitor shall have no liability in respect thereof.

31. **THIS COURT ORDERS** that no action or other proceeding shall be commenced against the Monitor in any way arising from or related to its capacity or conduct as Monitor except with prior leave pursuant to an order of this Court made on prior written notice to the Monitor and provided any such order granting leave includes a term granting the Monitor security for its costs and the costs of its coursel in connection with any proposed action or proceeding, such security to be on terms this Court deems just and appropriate.

32. THIS COURT ORDERS that as of the Implementation Time, the Monitor shall be

DOCSTOR: 2902739\4C

discharged and released from its duties other than those obligations, duties and responsibilities (i) necessary or required to give effect to the terms of the Plan and this Sanction Order, (ii) in relation to the claims procedure and all matters relating thereto as set out in the Claims Procedure Order, and (iii) in connection with the completion by the Monitor of all other matters for which it is responsible in connection with the Plan or pursuant to the Orders of this Court made in the CCAA Proceeding.

GENERAL PROVISIONS

33. **THIS COURT ORDERS** that the Applicant, the Monitor and any other Interested parties are hereby granted leave to apply to this Court for such further advice, directions or assistance as may be necessary to give effect to the terms of the Plan.

EFFECT, RECOGNITION AND ASSISTANCE

34. **THIS COURT ORDERS** that this Sanction Order shall have full force and effect in all provinces and territories in Canada and abroad and as against all Persons against whom it may apply.

35. THIS COURT HEREBY REQUESTS the aid and recognition of any court, tribunal, regulatory or administrative body having jurisdiction in Canada, the United States, Brazil or elsewhere to give effect to this Sanction Order and to assist the Applicant, the Monitor and their respective agents in carrying out the terms of this Sanction Order. All courts, tribunals, regulatory and administrative bodies are hereby respectfully requested to make such orders and to provide such assistance to the Applicant and to the Monitor, as an officer of this Court, as may be necessary or desirable to give effect to this Sanction Order, to grant representative status to the Monitor in any foreign proceeding, or to assist the Applicant and the Monitor and their respective agents in carrying out the terms of this Sanction Order.

DOCSTOR: 2902739\4C

36. **THIS COURT ORDERS** that each of the Applicant and the Monitor be at liberty and is hereby authorized and empowered to apply to any court, tribunal, regulatory or administrative body, wherever located, for the recognition of this Sanction Order and for assistance in carrying out the terms of this Sanction Order, and that the Monitor is authorized and empowered to act as a representative in respect of the CCAA Proceedings for the purpose of

having the CCAA Proceedings recognized in a jurisdiction outside Canada.

37. THE LOWER ORDERS BELIEVE A to the Affectivit of T. Doyles 50 Willock sworn Februs 5, 2010 he reduce peolis for the adder. N Hoan Rit.

ENTERED AT / INSCRIT À TORONTO ON / BOOK NO: LE / DANS LE REGISTRE NO.:

FEB 6 - 2014

. . .

Schedule "A"
Court File No. CV-13-10383-00CL

ONTARIO SUPERIOR COURT OF JUSTICE COMMERCIAL LIST

IN THE MATTER OF THE COMPANIES' CREDITORS ARRANGEMENT ACT, R.S.C. 1985, c. C-36, AS AMENDED

AND IN THE MATTER OF A PLAN OF COMPROMISE AND ARRANGEMENT OF JAGUAR MINING INC.

AMENDED AND RESTATED PLAN OF COMPROMISE AND ARRANGEMENT

PURSUANT TO THE COMPANIES' CREDITORS ARRANGEMENT ACT

OF JAGUAR MINING INC.

FEBRUARY 5, 2014

RECITALS

- (A) Jaguar Mining Inc. (the "Applicant" or "Jaguar") is a debtor company (as such term is defined in the *Companies' Creditors Arrangement Act*, R.S.C. 1985, c. C-36, as amended (the "CCAA").
- (B) On December 23, 2013, the Honourable Justice Morawetz of the Ontario Superior Court of Justice (Commercial List) (the "Court") granted the following Orders pursuant to the CCAA:
 - (i) an Initial Order in respect of the Applicant (as such Order may be amended, restated or varied from time to time, the "Initial Order");
 - (ii) a Plan Filing and Meeting Order (as such Order may be amended, restated or varied from time to time, the "Meeting Order") pursuant to which, among other things, the Applicant was authorized to file a plan of compromise and arrangement and to convene a meeting of affected creditors to consider and vote on the plan of compromise and arrangement, as may be amended, restated, modified or supplemented from time to time; and

66

DOCSTOR: 2926319\2A

- (iii) a Claims Procedure Order (as such Order may be amended, restated or varied from time to time, the "Claims Procedure Order"), which, among other things, established the procedures by which claims of affected creditors shall be filed in these proceedings.
- (C) This Amended and Restated Plan of Compromise and Arrangement will be filed on February 6, 2014 with the consent of the Majority Consenting Noteholders (as hereinafter defined).
- (D) Mineração Serras Do Oeste Ltda. ("MSOL"), Mineração Turmalina Ltda. ("MTL"), and MCT Mineração Ltda. ("MCT"), each incorporated under the laws of Brazil, are wholly-owned subsidiaries of Jaguar and are not applicants in the CCAA Proceedings.
- (E) The purpose of this Plan is to facilitate the continuation of the business of the Jaguar Group (as hereinafter defined) as a going concern, address certain liabilities of the Applicant, and effect a recapitalization and financing transaction on an expedited basis to provide a stronger financial foundation for the Jaguar Group going forward and additional liquidity to allow the Jaguar Group to continue to work towards its operational and financial goals from and after the Implementation Date in the expectation that all Persons (as hereinafter defined) with an economic interest in the Jaguar Group will derive a greater benefit from the implementation of this Plan than would otherwise result.

NOW THEREFORE the Applicant hereby proposes and presents this Plan under the CCAA.

ARTICLE 1 - INTERPRETATION

1.1 Definitions

In this Plan and the Recitals, unless otherwise stated or unless the subject matter or context otherwise requires:

"4.5% Convertible Note Indenture" means the Indenture dated as of September 15, 2009 among Jaguar, as issuer, The Bank of New York Mellon as trustee and BNY Trust Company of Canada as co-trustee pursuant to which Jaguar issued the 4.5% convertible notes;

"5.5% Convertible Note Indenture" means the Indenture dated as of February 9, 2011 among Jaguar as issuer, The Bank of New York Mellon as trustee and BNY Trust Company of Canada as co-trustee pursuant to which Jaguar issued the 5.5% convertible notes;

"Accrued Interest Claim" means, with respect to a particular Participating Eligible Investor or Funding Backstop Party, all unpaid interest accrued under the Notes at the applicable rate under the Indentures owing as at the Record Date to such Participating Eligible Investor or Funding Backstop Party;

"Accrued Interest Claims" means the aggregate of all unpaid interest accrued under the Notes at the applicable rate under the Indentures owing as at the Record Date to the Participating Eligible Investors and Funding Backstop Parties;

"Accrued Interest Offering Shares" means 9,044,203 New Common Shares;

"Ad Hoc Committee" means the ad hoc committee of Noteholders represented by the Advisors;

"Administration Charge" has the meaning given to that term in the Initial Order;

"Advisors" means Goodmans LLP, Houilhan Lokey Capital, Inc., Dias Carneiro Advogados, Behre

67

DOCSTOR: 2926319\2A DOCSTOR: 2926319\2A

Dolbear & Company (USA), Inc. and Stroock & Stroock & Lavan LLP;

"Affected Creditor Class" has the meaning given to that term in Section 3.1;

"Affected Unsecured Claims" means all Claims against the Applicant that are not Equity Claims;

"Affected Unsecured Creditor" means the holder of an Affected Unsecured Claim in respect of and to the extent of such Affected Unsecured Claim;

"Agreed Excluded Director/Officer Litigation Claims" means any claims against a Director and/or Officer that the Majority Consenting Noteholders and the Applicant have agreed, prior to the Implementation Date, and as set out on Schedule "A" hereto, will constitute Excluded Claims for the purposes of this Plan;

"Agreed Excluded Jaguar Litigation Claims" means any claims against Jaguar that the Majority Consenting Noteholders and the Applicant have agreed, prior to the Implementation Date, and as set out on Schedule "B" hereto, will constitute Excluded Claims for the purposes of this Plan;

*Agreed Excluded Litigation" means any proceeding commenced by any Agreed Excluded Litigation Claimant in respect of any Agreed Excluded Litigation Claims, subject to the terms of this Plan;

*Agreed Excluded Litigation Claimants" means any Persons and, if applicable, each of their respective parents, subsidiaries, associated, affiliated and related companies, corporations and Persons, and each of their directors, officers, employees, agents, affiliates, and trustees, that have asserted an Agreed Excluded Director/Officer Litigation Claim and/or an Agreed Excluded Jaguar Litigation Claim, as agreed to by the Majority Consenting Noteholders and the Applicant prior to the Implementation Date and as set out on Schedule "C" hereto;

"Agreed Excluded Litigation Claims" means, collectively, the Agreed Excluded Jaguar Litigation Claims and the Agreed Excluded Director/Officer Litigation Claims;

"Allowed" means, with respect to a Claim, any Claim or any portion thereof that has been finally allowed as a Distribution Claim (as defined in the Claims Procedure Order) for purposes of receiving distributions under this Plan in accordance with the Claims Procedure Order and the CCAA;

"Applicable Law" means, with respect to any Person, property, transaction, event or other matter, any Law relating or applicable to such Person, property, transaction, event or other matter, including, where appropriate, any interpretation of the Law (or any part) by any Person, court or tribunal having jurisdiction over it, or charged with its administration or interpretation;

"Applicant" has the meaning given to that term in Recital A;

"Articles of Reorganization" means the Articles of Reorganization of Jaguar to be filed pursuant to Section 186 of the OBCA and in accordance with Section 7.4(a) hereof, in form and substance satisfactory to Jaguar and the Majority Consenting Noteholders;

"Assumed Backstop Commitment" means, in the event of a Backstop Default/Termination, If any, a Backstop Commitment, or a portion thereof, assumed by an Assuming Backstop Party from a Defaulting Backstop Party, Objecting Backstop Party, Breaching Backstop Party or Non-Delivering Backstop Party, as applicable, in accordance with the terms and conditions of this Plan and the Backstop Agreement:

"Assuming Backstop Party" means, in the event of a Backstop Default/Termination, if any, a Non-Defaulting Backstop Party, Non-Objecting Backstop Party, Non-Breaching/Non-Delivering Backstop Party, or such other party acceptable to the Backstop Parties and Jaguar in each case in accordance with the Backstop Agreement, that executes a Backstop Consent Agreement and that has assumed the

DOCSTOR: 2926319\2A DOCSTOR: 2926319\2A

obligations (and rights), or a portion thereof, of a Defaulting Backstop Party, Objecting Backstop Party, Breaching Backstop Party or Non-Delivering Backstop Party, as applicable, under the Backstop Agreement, in accordance with the terms and conditions of this Plan and the Backstop Agreement. For greater certainty, any Assuming Backstop Party that has complied with its obligations under this Plan and the Backstop Agreement shall constitute and be treated as a Funding Backstop Party for purposes of this Plan;

"Backstop Agreement" means the backstop agreement dated November 13, 2013 (as amended from time to time) between certain Noteholders, Jaguar, MCT, MSOL and MTL, together with any Backstop Consent Agreements executed by other parties from time to time;

"Backstop Commitment" means, in respect of each Backstop Party, the commitment set forth on such Backstop Party's signature page to the Backstop Agreement or a Backstop Consent Agreement, as applicable, which commitment may be reduced in accordance with and subject to the terms and conditions of the Backstop Agreement and this Plan;

"Backstop Commitment Reduction Election" has the meaning given to such term in Section 4.1(c);

"Backstop Commitment Shares" means 11,111,111 New Common Shares;

"Backstop Consent Agreement" means an agreement substantially in the form of Schedule B to the Backstop Agreement;

"Backstop Consideration Commitment" means, in respect of each Backstop Party, the commitment set forth on such Backstop Party's signature page to the Backstop Agreement or a Backstop Consent Agreement, as applicable, which commitment, for greater certainty, shall not be reduced as a result of a Backstop Commitment Reduction Election;

"Backstop Default/Termination" means any of the following: (a) a breach by a Breaching Backstop Party under section 10(b)(i) or (ii) of the Backstop Agreement in respect of which the Backstop Agreement has been terminated with respect to such Breaching Backstop Party in accordance with its terms; (b) a failure by a Defaulting Backstop Party to meet its obligations in respect of its Backstop Commitment on or before the Backstop Funding Deadline; (c) a failure by a Non-Delivering Backstop Party to deliver an executed Rep Letter to Jaguar by the Election Deadline or if a representation or warranty made in such Rep Letter becomes untrue; and (d) the termination by an Objecting Backstop Party of its obligations under the Backstop Agreement in accordance with section 8(c) thereof;

"Backstop Funding Deadline" has the meaning given to such term in Section 4.1(g);

***Backstop** Parties" means those Noteholders that have entered into the Backstop Agreement (including a Backstop Consent Agreement), and a "**Backstop Party**" means any one of the Backstop Parties, and their permitted assignees;

"Backstop Payment Amount" has the meaning given to such term in Section 4.1(f);

"Backstop Purchase Obligation" means the obligation of a Backstop Party to purchase Backstopped Shares in accordance with the terms and conditions of the Backstop Agreement and this Plan;

"Backstopped Shares" has the meaning given to such term in Section 4.1(f);

"Beneficial Noteholder" means a beneficial or entitlement holder of Notes holding such Notes in a securities account with a depository, a depository participant or other securities intermediary including, for greater certainty, such depository participant or other securities intermediary only if and to the extent such depository participant or other securities intermediary only if and to the extent such depository participant or other securities as a principal for its own account;

DOCSTOR: 2926319\2A DOCSTOR: 2926319\2A

· ·

"Bradesco" means Banco Bradesco S.A.;

"Breaching Backstop Party" means a Backstop Party that has breached the Backstop Agreement under section 10(b)(i) or (ii) thereof and in respect of whom the Backstop Agreement has been terminated in accordance with its terms;

"Business Day" means any day, other than a Saturday, or a Sunday or a statutory or civic holiday, on which banks are generally open for business in Toronto, Ontario and New York, New York;

"CCAA Proceedings" means the proceedings commenced by the Applicant under the CCAA as contemplated by the initial Order;

"CRA Claim" means the claim as described in the proof of claim, dated January 21, 2014, filed by Canada Revenue Agency in the CCAA Proceedings in the amount of \$5,969.13;

"Charges" has the meaning ascribed thereto in the Initial Order;

"Claim" means:

- any right or claim, including any Tax Claim, of any Person that may be asserted or made í. in whole or in part against the Applicant, in any capacity, whether or not asserted or made, In connection with any indebtedness, liability or obligation of any kind whatsoever of the Applicant, and any interest accrued thereon or costs payable in respect thereof, in existence on the Filing Date, or which is based on an event, fact, act or omission which occurred in whole or in part prior to the Filing Date, whether at law or in equity, Including by reason of the commission of a tort (intentional or unintentional), by reason of any breach of contract or other agreement (oral or written), by reason of any breach of duty (including, any legal, statutory, equitable or fiduciary duty) or by reason of any equity interest, right of ownership of or title to property or assets or right to a trust or deemed trust (statutory, express, implied, resulting, constructive or otherwise), and together with any security enforcement costs or legal costs associated with any such claim, and whether or not any indebtedness, liability or obligation is reduced to judgment, liquidated, unliquidated, fixed, contingent, matured, unmatured, disputed, undisputed, legal. equitable, secured, unsecured, perfected, unperfected, present or future, known or unknown, by guarantee, warranty, surety or otherwise, and whether or not any right or claim is executory or anticipatory in nature, including any claim arising from or caused by the termination, disclaimer, resiliation, assignment or repudiation by the Applicant of any contract, lease or other agreement, whether written or oral, any claim made or asserted against the Applicant through any affiliate, subsidiary, associated or related person, or any right or ability of any Person to advance a claim for an accounting, reconciliation, contribution, indemnity, restitution or otherwise with respect to any matter, grievance, action (including any class action or proceeding before an administrative tribunal), cause or chose in action, whether existing at present or commenced in the future, and including any other claims that would have been claims provable in bankruptcy had the Applicant become bankrupt on the Filing Date, including for greater certainty any Equity Claim and any claim against the Applicant for indemnification by Director or Officer in respect of a Director/Officer Claim but excluding any such indemnification claims covered by the Directors' Charge (each, a "Pre-filing Claim", and collectively, the "Pre-filing Claims");
- ii. any right or claim of any Person against the Applicant in connection with any indebtedness, liability or obligation of any kind whatsoever owed by the Applicant to such Person arising out of the restructuring, disclaimer, resiliation, termination or breach by the Applicant on or after the Filing Date of any contract, lease or other agreement whether written or oral (each, a "Restructuring Period Claim", and collectively, the

"Restructuring Period Claims"); and

iii. any right or claim of any Person against one or more of the Directors or Officers howsoever arising, whether or not such right or claim is reduced to judgment, liquidated, unliquidated, fixed, contingent, matured, unmatured, disputed, undisputed, legal, equitable, secured, unsecured, perfected, unperfected, present, future, known, or unknown, by guarantee, surety or otherwise, and whether or not such right is executory or anticipatory in nature, including the right or ability of any Person to advance a claim for contribution or Indemnity or otherwise with respect to any matter, action, cause or chose in action, whether existing at present or commenced in the future, including any right of contribution or indemnity, for which any Director or Officer is alleged to be by statute or otherwise by law liable to pay in his or her capacity as a Director or Officer (each a "Director/Officer Claim", and collectively, the "Director/Officer Claims"),

in each case other than any Excluded Claim;

"Commitment Reduction Electing Backstopper" has the meaning given to such term in Section 4.1(c);

"Common Share Consolidation" has the meaning given to such term in Section 7.4(a);

"Common Shares" means the common shares in the capital of Jaguar that are duly issued and outstanding at any time;

"Consenting Noteholder" means any Noteholder that has executed the Support Agreement (including a consent agreement substantially in the form of Schedule C thereto), in respect of whom the Support Agreement has not been terminated;

"Consolidation Number" means the quotient (to five decimal places) determined by dividing the number of Existing Shares by 1,000,000, which as of the date of this Plan is 86,39636.

"Continuing Other Director/Officer Claims" means Director/Officer Claims against the Other Directors and/or Officers;

"Court" has the meaning given to that term in Recital B;

"Credit Agreement" means the credit agreement made as of December 17, 2012 between Jaguar, as borrower, the Subsidiaries, as guarantors, and Global Resource Fund, as lender.

"Creditor" means any Person having a Claim and includes without limitation the transferee or assignee of a Claim transferred and recognized as a Creditor in accordance with the Claims Procedure Order or a trustee, executor, liquidator, receiver, receiver and manager, or other Person acting on behalf of or through such Person;

"Crown" means Her Majesty in right of Canada or a province of Canada;

"Crown Claim" means any Claim of the Crown, for all amounts that were outstanding at the Filing Date and are of a kind that could be subject to a demand under:

- subsection 224(1.2) of the ITA;
- ii. any provision of the Canada Pension Plan or of the Employment Insurance Act that refers to subsection 224(1.2) of the ITA and provides for the collection of a contribution, as defined in the Canada Pension Plan, or an employee's premium, or employer's premium, as defined in the Employment Insurance Act, and of any related interest, penalties or

other amounts;

- iii. any provision of provincial legislation that has a similar purpose to subsection 224(1.2) of the ITA, or that refers to that subsection, to the extent that it provides for the collection of a sum, and of any related interest, penalties or other amounts, where the sum:
 - a. has been withheld or deducted by a person from a payment to another person and is in respect of a tax similar in nature to the income tax imposed on individuals under the ITA; or
 - b. is of the same nature as a contribution under the Canada Pension Plan if the province is a "province providing a comprehensive pension plan" as defined in subsection 3(1) of the Canada Pension Plan and the provincial legislation establishes a "provincial pension plan" as defined in that subsection.

"Defaulting Backstop Party" means a Backstop Party that has failed to meet its obligations in respect of its Backstop Commitment on or before the Backstop Funding Deadline;

"Designated Offshore Securities Market" has the meaning given to that term in Rule 902 of Regulation S.

"Direct Registration System Advice" means, if applicable, a statement delivered by the Transfer Agent or any such Person's agent to any Person entitled to receive New Common Shares pursuant to the Plan indicating the number of New Common Shares registered in the name of or as directed by the applicable Person in a direct registration account administered by the Transfer Agent in which those Persons entitled to receive New Common Shares pursuant to the Plan will hold such New Common Shares in registered form and including, if applicable, a securities law legend;

"Director" means anyone who is or was, or may be deemed to be or have been, whether by statute, operation of law or otherwise, a director or *de facto* director of Jaguar;

"Director Defence Costs Indemnity Claim" means any existing or future right of any current director (as at the date of this Plan) of Jaguar who is a defendant to any Agreed Excluded Director/Officer Litigation Claims against Jaguar for indemnification of reasonable defence costs incurred by such current director of Jaguar (whether or not a director of Jaguar at the time such claim for indemnification is made) in connection with defending against such Agreed Excluded Director/Officer Litigation Claims solely to the extent that such defence costs are not covered by insurance and for which such Director or Officer of Jaguar;

"Director/Officer Claim" has the meaning given to that term in the definition of Claim;

"Director/Officer Indemnity Claim" means any existing or future right of any Director or Officer of Jaguar against Jaguar that arose or arises as a result of (i) any Person filing a Proof of Claim (as defined in the Claims Procedure Order) in respect of a Director/Officer Claim in respect of such Director or Officer of Jaguar or (ii) any Agreed Excluded Litigation Claims and/or any Agreed Excluded Litigation, in each case for which such Director or Officer of Jaguar is entitled to be indemnified by Jaguar, other than a Director Defence Costs Indemnity Claim;

"Director/Officer Insurance Policy" means any insurance policy pursuant to which any Director or Officer is insured, in his or her capacity as a Director or Officer;

"Directors' Charge" has the meaning given to that term in the initial Order;

"Disputed Distribution Claim" means an Affected Unsecured Claim (including a contingent Affected Unsecured Claim which may crystallize upon the occurrence of an event or events occurring after the

DOCSTOR: 2926319\2A DOCSTOR: 2926319\2A

Filing Date) or such portion thereof which has not been allowed as a Distribution Claim (as defined in the Claims Procedure Order), which is validly disputed for distribution purposes in accordance with the Claims Procedure Order and which remains subject to adjudication for distribution purposes in accordance with the Claims Procedure Order;

"Disputed Distribution Claims Reserve" means the reserve, if any, to be established by the Applicant on the Implementation Date, which shall be comprised of the Unsecured Creditor Common Shares that would have been delivered in respect of Disputed Distribution Claims if such Disputed Distribution Claims had been Allowed Claims as of such date;

"Disputed Voting Claim" means an Affected Unsecured Claim (including a contingent Affected Unsecured Claim which may crystallize upon the occurrence of an event or events occurring after the Filing Date) or such portion thereof which has not been allowed as a Voting Claim, which is validly disputed for voting purposes in accordance with the Claims Procedure Order and which remains subject to adjudication for voting purposes in accordance with the Claims Procedure Order;

"Distribution Claim" means the amount of the Affected Unsecured Claim of an Affected Unsecured Creditor against the Applicant as finally accepted and determined for distribution purposes in accordance with this Claims Procedure Order and the CCAA;

"Distribution Record Date" means the Business Day immediately before the implementation Date;

"DSU Plan" means the Deferred Share Unit Plan for non-executive directors adopted in November of 2008 by Jaguar, as amended from time to time;

***DSU/RSU/SAR Notice**" means a notice delivered by Goodmans to Jaguar prior to the date scheduled for the hearing of the motion for the Sanction Order, If, in satisfaction of Section 12.3(g) hereof, Jaguar and the Majority Consenting Noteholders have agreed to terminate the DSU Plan, the RSU Plan, and/or the SAR Plan:

"DTC" means The Depository Trust Company, or any successor thereof;

"Early Consent Deadline" means November 26, 2013 (or such other date as the Applicant, the Monitor and the Majority Consenting Noteholders may agree);

"Early Consent Shares" means 5,000,000 New Common Shares;

"Early Consenting Noteholder" means any Noteholder that has executed the Support Agreement (including a consent agreement substantially in the form of Schedule C thereto) on or before the Early Consent Deadline and in respect of whom the Support Agreement has not been terminated;

"Election Deadline" means 5:00 p.m. on the second Business Day before the Meeting (or such other time or date as the Applicant and the Majority Consenting Noteholders may agree);

"Election Form" has the meaning given to that term in Section 4.1(b);

"Electing Eligible Investor" means an Eligible Investor who has completed and submitted an Election Form on or prior to the Election Deadline to participate in the Share Offering in accordance with the Meeting Order, provided that an Electing Eligible Investor that irrevocably elects under Section 4.1(b) to participate in the Share Offering and subscribes for such number of Offering Shares that is less than such Eligible Investor's Pro Rata Share of all Offering Shares offered pursuant to the Share Offering shall be deemed to be an Electing Eligible Investor only in respect of such lesser amount, and shall not be treated as an Electing Eligible Investor in respect of the balance;

"Electing Eligible Investor Funding Amount" has the meaning given to that term in Section 4.1(d);

"Electing Eligible Investor Funding Deadline" has the meaning given to that term in Section 4.1(e);

"Eligible Investor" means a person that: (i) is a Noteholder as at the Subscription Record Date; and (ii) has delivered an executed Rep Letter to Jaguar on or before the Election Deadline and the information set forth in such Rep Letter is true and correct as of the Implementation Date, and such person's permitted assignees;

"Eligible Voting Creditors" means Affected Unsecured Creditors holding Voting Claims or Disputed Voting Claims;

"Employee Priority Claims" means the following claims of Jaguar's employees and former employees:

- claims equal to the amounts that such employees and former employees would have been qualified to receive under paragraph 136(I)(d) of the Bankruptcy and Insolvency Act (Canada) If Jaguar had become bankrupt on the Filing Date; and
- ii. claims for wages, salaries, commissions or compensation for services rendered by them after the Filing Date and on or before the date of the Sanction Order, together with, in the case of travelling salespersons, disbursements properly incurred by them in and about Jaguar's business during the same period.

"Equity Claim" has the meaning set forth in section 2(1) of the CCAA;

"Escrow Agent" means an independent third party escrow agent agreed to by Jaguar and the Majority Backstop Parties, in each case acting reasonably;

"Escrow Agreement" means the escrow agreement entered into by the Escrow Agent, Jaguar and the applicable Participating Eligible Investors and Funding Backstop Parties in connection with the Share Offering;

"Excluded Claim" means

- i. any claims secured by any of the Charges;
- ii. any Section 5.1(2) Director/Officer Claims;
- III, any claims that cannot be compromised pursuant to subsection 19(2) of the CCAA, provided that no claims that have been or may be asserted by any Agreed Excluded Litigation Claimant shall constitute claims that cannot be compromised pursuant to subsection 19(2) of the CCAA for purposes of this Plan;
- iv. any claims of the Subsidiaries against the Applicant;
- v. any Secured Claims;
- vi. any Employee Priority Claims against the Applicant;
- vii. any Crown Claims against the Applicant;
- viii. the Trustees' claims under Section 6.07 of the 4.5% Convertible Note Indenture and the 5.5% Convertible Note Indenture, if any;

- ix. any Post-Filing Claims;
- any claims of Persons who, at the Filing Date, are senior officers or employees of the Applicant, in respect of their employment arrangements or any termination of such arrangements;
- xi. the Renvest Claim;
- xii, the Agreed Excluded Director/Officer Litigation Claims;
- xiji. the Agreed Excluded Jaguar Litigation Claims; and
- xiv, the CRA Claim.

*Excluded Creditor" means a Person who has an Excluded Claim, but only in respect of and to the extent of such Excluded Claim;

"Existing Equity Holders" means, collectively, the Existing Shareholders and, as context requires, the Registered Holders or beneficial holders of Existing Share Options and the Registered Holders or beneficial holders of Rights, in their capacities as such;

"Existing Shareholders" means, as context requires, Registered Holders or beneficial holders of the Existing Shares, in their capacities as such;

"Existing Share Options" means all rights, options, warrants and other securities (other than the Notes) convertible or exchangeable into equity securities of Jaguar;

"Existing Shares" means all common shares of Jaguar that are issued and outstanding at the applicable time prior to the Implementation Time;

"Filing Date" means December 23, 2013;

"Funding Backstop Party" means a Backstop Party (I) in respect of whom the Backstop Agreement has not been terminated and (ii) unless such Backstop Party's Backstop Commitment has been reduced to zero in accordance with the Backstop Agreement and this Plan, who has deposited in escrow with the Escrow Agent either (a) its Backstop Payment Amount in full in cash; or (b) a qualified letter of credit in the full amount of its Backstop Payment Amount, in each case by the Backstop Funding Deadline and in accordance with the Backstop Agreement and Section 4.1(g) of this Plan;

"Governmental Entity" means any government, regulatory authority, governmental department, agency, commission, bureau, official, minister, Crown corporation, court, board, tribunal or dispute settlement panel or other law, rule or regulation-making organization or entity: (a) having or purporting to have jurisdiction on behalf of any nation, province, territory or state or any other geographic or political subdivision of any of them; or (b) exercising, or entitled or purporting to exercise any administrative, executive, judicial, legislative, policy, regulatory or taxing authority or power;

"Implementation Date" means the Business Day on which this Plan becomes effective, which shall be the Business Day on which the Monitor has filed with the Court the certificate contemplated in Section 12.6 hereof, or such other date as the Applicant, the Monitor and the Majority Consenting Noteholders may agree;

"Implementation Time" means 12:01 a.m. on the Implementation Date (or such other time as the Applicant, the Monitor and the Majority Consenting Noteholders may agree);

"Indentures" means the 4.5% Convertible Note Indenture and the 5.5% Convertible Note Indenture;

"Initial Order" has the meaning given to that term in Recital B;

"ITA" means the Income Tax Act, R.S.C. 1985, c.1 (5th Supp.);

"Itaú BBA" means Banco Itaú BBA S.A.;

"Jaguar Group" means, collectively, Jaguar, MSOL, MCT, MTL.;

"Jaguar Insurance Policy" means any insurance policy pursuant to which Jaguar is insured and any Director or Officer is insured, in his or her capacity as a Director or Officer;

***Law**" means any law, statute, order, decree, consent decree, judgment, rule regulation, ordinance or other pronouncement having the effect of law whether in Canada, the United States, Brazil or any other country, or any domestic or foreign state, county, province, city or other political subdivision or of any Governmental Entity;

"Letter of Transmittal" means a letter of transmittal to be used by Registered Holders of Existing Shares to obtain replacement share certificates reflecting the Common Share Consolidation;

"Majority Backstop Parties" means the Backstop Parties (other than Defaulting Backstop Parties) having at least 66^{-2/3} % of the aggregate Backstop Commitment of the Backstop Parties (other than Defaulting Backstop Parties) at the time that a consent, approval, waiver or agreement is sought pursuant to the terms of this Plan;

"Majority Consenting Noteholders" means Consenting Noteholders holding at least a majority of the aggregate principal amount of all Notes held by all Consenting Noteholders at the time that a consent, approval, waiver or agreement is sought pursuant to the terms of this Plan;

"MCT" has the meaning given to that term in Recital C;

"MSOL" has the meaning given to that term in Recital C;

"MTL" has the meaning given to that term in Recital C;

"Meeting" means a meeting of the Affected Unsecured Creditors called for the purpose of considering and voting in respect of this Plan;

"Monitor" means FTI Consulting Canada Inc., in its capacity as Court-appointed Monitor of Jaguar in the CCAA Proceedings;

"Named Directors and Officers" means the current directors and officers of Jaguar and such other directors and officers as agreed to by the Majority Consenting Noteholders prior to the Meeting;

"New Board" means the board of directors in place from and after the Implementation Date, the composition and size of which shall be satisfactory to the Majority Backstop Parties, subject to applicable Law;

"New Common Shares" means the 110,111,111 Common Shares to be issued by Jaguar on the Implementation Date in accordance with the steps set out in Section 7.4;

"Non-Breaching/Non-Delivering Backstop Parties" means those Backstop Parties that are neither

Breaching Backstop Parties nor Non-Delivering Backstop Parties;

· .

"Non-Defaulting Backstop Parties" means those Backstop Parties that are not Defaulting Backstop Parties:

"Non-Delivering Backstop Party" means a Backstop Party (who is not otherwise an Objecting Backstop Party) that has not delivered an executed Rep Letter to Jaguar by the Election Deadline or for whom a representation or warranty made in such Rep Letter becomes untrue;

"Non-Objecting Backstop Parties" means those Backstop Parties that are not Objecting Backstop Parties;

"Non-Released Director/Officer Claims" means Director/Officer Claims against the Directors and Officers of Jaguar in respect of which such Director or Officer has been adjudged by the express terms of a judgment rendered on a final determination on the merits to have committed fraud or wilful misconduct, but excluding any claims that have been or may be asserted by any Agreed Excluded Litigation Claimants;

"Noteholder Released Claim" means the matters that are subject to release and discharge pursuant to Section 11.1(c);

"Noteholder Released Party" has the meaning given to that term in Section 11.1(c);

"Noteholder Voting Record Date" means December 19, 2013;

"Noteholders" means, as the context requires, the Registered Holders or beneficial holders of the Notes, in their capacities as such;

"Noteholders Allowed Claim" means all principal amounts outstanding and all accrued interest under the Notes as at the applicable record date under this Plan as determined in accordance with the Claims Procedure Order for purposes of voting on, and receiving distributions under, this Plan;

"Noteholder's Allowed Claim" means, in respect of a particular Noteholder, all principal amounts outstanding and accrued interest under the Notes owing to such Noteholder as at the applicable record date under this Plan as determined in accordance with the Claims Procedure Order for purposes of voting on, and receiving distributions under, this Plan;

"Notes" means, collectively, the notes issued by Jaguar under and pursuant to the Indentures;

"Objecting Backstop Party" means a Backstop Party that has terminated its obligations under the Backstop Agreement in accordance with section 8(c) thereof;

"Offering Shares" means the 70,955,797 New Common Shares to be issued by Jaguar pursuant to the Share Offering;

"Offered Shares" means, collectively, the Offering Shares (including the Backstopped Shares), the Accrued Interest Offering Shares, and the Backstop Commitment Shares;

"Officer" means anyone who is or was, or may be deemed to be or have been, whether by statute, operation of law or otherwise, an officer or *de facto* officer of Jaguar;

"Order" means any order of the Court in the CCAA Proceedings;

"Other Directors and/or Officers" means any Directors and/or Officers other than the Named Directors

DOCSTOR: 2926319\2A DOCSTOR: 2926319\2A

and Officers;

"Outside Date" means February 28, 2014 (or such other date as the Applicant and the Majority Consenting Noteholders may agree);

"Participant Holder" has the meaning ascribed thereto in the Meeting Order;

"Participating Eligible Investor" has the meaning given to that term in Section 4.1(h);

"Participating Eligible Investor Shares" has the meaning given to that term in Section 4.1(h);

"Party" means a party to the Support Agreement and/or to the Backstop Agreement, and any reference to a Party includes its successors and permitted assigns; and "Parties" means every Party;

"**Person**" is to be broadly interpreted and includes any individual, firm, corporation, limited or unlimited liability company, general or limited partnership, association, trust, unincorporated organization, joint venture, Government Entity or any agency, officer or instrumentality thereof or any other entity, wherever situate or domiciled, and whether or not having legal status;

"Plan" means this Amended and Restated Plan of Compromise and Arrangement and any amendments, modifications or supplements hereto made in accordance with the terms hereof or made at the direction of the Court in the Sanction Order or otherwise with the consent of Jaguar and the Majority Consenting Noteholders, each acting reasonably;

"Plan Resolution" means the resolution of the Affected Unsecured Creditors relating to this Plan considered at the Meeting;

"Post-Filing Claim" means any claims against the Applicant that arose from the provision of authorized goods and services provided or otherwise incurred on or after the Filing Date in the ordinary course of business, but specifically excluding any Restructuring Period Claim;

"Pre-filing Claim" has the meaning given to that term in the definition of Claim;

"Pro Rata Share" means.

- (a) in respect of Unsecured Creditor Common Shares, the percentage that an Affected Unsecured Creditor's Allowed Affected Unsecured Claim calculated as at the Record Date bears to the aggregate of all Allowed Affected Unsecured Claims calculated as at the Record Date and all Disputed Distribution Claims calculated as at the Record Date;
- (b) in respect of the Early Consent Shares, the percentage that an Early Consenting Noteholder's Noteholder's Allowed Claim calculated as at the Record Date bears to the aggregate of all Early Consenting Noteholders' Noteholder's Allowed Claims calculated as at the Record Date;
- (c) in respect of the Subscription Privilege, the percentage that an Eligible Investor's Noteholder's Allowed Claim calculated as at the Record Date bears to the Noteholders Allowed Claim calculated as at the Record Date, subject to adjustment pursuant to Section 5.2(c) hereof;
- (d) in respect of the Accrued Interest Offering Shares, the percentage that a Participating Eligible Investor's Accrued Interest Claim or a Funding Backstop Party's Accrued Interest Claim (without duplication), as applicable, bears to the aggregate of all Accrued Interest

Claims;

- (e) in respect of the Backstop Commitment Shares, the percentage that a Funding Backstop Party's Backstop Consideration Commitment bears to the aggregate of all Funding Backstop Parties' Backstop Consideration Commitments; and
- (i) in respect of the Backstopped Shares, the percentage that a Backstop Party's Backstop
 Commitment bears to the aggregate of all Backstop Commitments.

"Record Date" means December 31, 2013;

"Registered Holder" means (i) in respect of the Notes, the holder of such Notes as recorded on the books and records of the Trustees, (ii) in respect of the Existing Shares, the holder of such Existing Shares as recorded on the share register maintained by the Transfer Agent, and (iii) in respect of the Existing Share Options, the holder of such Existing Share Options as recorded on the books and records of Jaguar;

"Regulation S" means Regulation S as promulgated by the US Securities Commission under the US Securities Act;

"Released Claims" means the matters that are subject to release and discharge pursuant to Section 11.1(a) and (b) hereof;

"Released Party" has the meaning given to that term in Section 11.1(b);

"Renvest Claim" means any claim for amounts owing by the Applicant to Global Resource Fund, pursuant to the Credit Agreement or pursuant to any Credit Document (as such term is defined in the Credit Agreement).

"Rep Letter" means a letter from a Noteholder, or an Assuming Backstop Party who is not a Noteholder, or an Affected Unsecured Creditor with an Allowed Affected Unsecured Claim who is not a Noteholder, if applicable in accordance with Section 5.2(c) hereof, to Jaguar containing representations and warrantles relating to such Person's eligibility to acquire the Offering Shares (including the Backstopped Shares), Accrued Interest Offering Shares, or Backstop Commitment Shares under US Securities Laws, in a form acceptable to such Person and Jaguar, each acting reasonably;

"Required Majority" means a majority in number of Affected Unsecured Creditors representing at least two thirds in value of the Voting Claims of Affected Unsecured Creditors who are entitled to vote at the Meeting in accordance with the Meeting Order and who are present and voting in person or by proxy on the Plan Resolution at the Meeting;

"Restructuring Period Claim" has the meaning given to that term in the definition of Claim;

"Rights" means the rights issued pursuant to the Shareholder Rights Plan;

***RSU Plan**" means the restricted share unit plan for senior officers, employees and consultants adopted in November of 2008 by Jaguar, as amended from time to time;

"SAR Plan" means the Third Amended and Restated Share Appreciation Rights Plan of Jaguar, effective as of December 8, 2010;

"Sanction Order" means the Order of the Court sanctioning and approving this Plan pursuant to section 6(1) of the CCAA, which shall include such terms as may be necessary or appropriate to (i) give effect to this Plan, in form and substance satisfactory to the Applicant and the Majority Consenting Noteholders,

each acting reasonably, and (ii) allow Jaguar to rely on the exemption from registration set forth in section 3(a)(10) of the US Securities Act;

"Section 5.1(2) Director/Officer Claim" means any claim against any Director and/or Officer that is not permitted to be compromised pursuant to section 5.1(2) of the CCAA, but only to the extent not so permitted, provided that any Director/Officer Claim that qualifies as a Non-Released Director/Officer Claim shall not constitute a Section 5.1(2) Director/Officer Claim for the purposes of Section 11.1(a) hereof; and provided further that no claims that have been or may be asserted by any Agreed Excluded Litigation Claimant shall constitute Section 5.1(2) Director/Officer Claims for the purposes of this Plan;

"Secured Claims" means that portion of a Claim that Is (I) secured by security validly charging or encumbering property or assets of the Applicant (including statutory and possessory liens that create security interests) but only up to the value of such collateral, and (ii) duly and properly perfected in accordance with the relevant legislation in the appropriate jurisdiction as of the Filing Date;

"Share Offering" means the offering by Jaguar of Offering Shares at the Subscription Price in accordance with this Plan;

"Shareholder Rights Plan" means the Shareholder Rights Plan Agreement dated May 2, 2013 between Jaguar Mining Inc. and Computershare Investor Services Inc. as Rights Agent;

"Solicitation/Election Agent" means Globic Advisors Inc., or any successor solicitation or election agent;

"Stock Option Plan" means the stock option plan of Jaguar in effect as of the Filing Date;

"Subscription Price" means \$0,7047 per Offering Share;

"Subscription Privilege" means the right of an Eligible Investor to participate in the Share Offering by electing, in accordance with the provisions of this Plan, to subscribe for and purchase from Jaguar up to its Pro Rata Share of Offering Shares under the Share Offering;

"Subscription Record Date" means December 19, 2013;

"Subsidiaries" means, collectively, MTL, MSOL and MCT, and "Subsidiary" means any one of the Subsidiaries;

"Support Agreement" means the Support Agreement made November 13, 2013 (as amended from time to time) between Jaguar, the Subsidiaries and the Noteholders party thereto, together with any consent agreements executed by other Noteholders from time to time, substantially in the form of Schedule C thereto;

"Tax" or "Taxes" means any and all taxes, duties, fees, premiums, assessments, imposts, levies and other charges of any kind whatsoever, including all Interest, penalties, fines, additions to tax or other additional amounts in respect thereof, and including those levied on, or measured by, or referred to as, income, gross receipts, profits, capital, transfer, land transfer, sales, goods and services, harmonized sales, use, value-added, excise, stamp, withholding, business, franchising, property, development, occupancy, employer health, payroll, employment, health, social services, education and social security taxes, all surfaxes, all customs duties and import and export taxes, countervall and anti-dumping, all licence, franchise and registration fees and all employment insurance, health insurance and Canada, Quebec and other government pension plan premiums or contributions;

"Tax Claim" means any Claim against the Applicant for any Taxes in respect of any taxation year or period;

*Transfer Agent" means Computershare Investor Services Inc.;

"Trustees" means The Bank of New York Mellon, as trustee, and BNY Trust Company of Canada, as cotrustee, under each of the Indentures; 81

"TSX" means Toronto Stock Exchange;

"TSXV" means TSX Venture Exchange;

"Undeliverable Distribution" has the meaning given to that term in Section 8.3;

"Unsecured Creditor Common Shares" means 14,000,000 New Common Shares;

"US Dollars" or "US\$" means the lawful currency of the United States of America;

"US Securities Act" means the United States Securities Act of 1933, as amended from time to time, and the rules and regulations promulgated thereunder, or any successor statute;

"US Securities Commission" means the United States Securities and Exchange Commission;

"US Securities Laws" means, collectively, the Sarbanes-Oxley Act of 2002 ("Sarbanes-Oxley"), the US Securities Act, as amended, the United States Securities Exchange Act of 1934, as amended, the rules and regulations of the US Securities Commission, the auditing principles, rules, standards and practices applicable to auditors of "issuers" (as defined in Sarbanes-Oxley) promulgated or approved by the Public Company Accounting Oversight Board and, as applicable, the rules of the New York Stock Exchange;

"Voting Claim" means the amount of the Affected Unsecured Claim of an Affected Unsecured Creditor against the Applicant as finally accepted and determined for purposes of voting at the Meeting, in accordance with the provisions of the Claims Procedure Order and the CCAA; and

"Voting Deadline" means 10 a.m. on the Business Day prior to the Meeting.

1.2 Certain Rules of Interpretation

For the purposes of this Plan:

- (a) Unless otherwise expressly provided herein, any reference in this Plan to an instrument, agreement or an Order or an existing document or exhibit filed or to be filed means such instrument, agreement, Order, document or exhibit as it may have been or may be amended, modified, or supplemented in accordance with its terms;
- (b) The division of this Plan into articles and sections are for convenience of reference only and do not affect the construction or interpretation of this Plan, nor are the descriptive headings of articles and sections intended as complete or accurate descriptions of the content thereof;
- (c) The use of words in the singular or plural, or with a particular gender, including a definition, shall not limit the scope or exclude the application of any provision of this Plan to such Person (or Persons) or circumstances as the context otherwise permits;
- (d) The words "includes" and "including" and similar terms of inclusion shall not, unless expressly modified by the words "only" or "solely", be construed as terms of limitation, but rather shall mean "includes but is not limited to" and "including but not limited to", so that references to included matters shall be regarded as illustrative without being either

characterizing or exhaustive;

- (e) Unless otherwise specified, all references to time herein and in any document issued pursuant hereto mean local time in Toronto, Ontario and any reference to an event occurring on a Business Day shall mean prior to 5:00 p.m. on such Business Day;
- (f) Unless otherwise specified, time periods within or following which any payment is to be made or act is to be done shall be calculated by excluding the day on which the period commences and including the day on which the period ends;
- (g) Unless otherwise provided, any reference to a statute or other enactment of parliament, a legislature or other Governmental Entity includes all regulations made thereunder, all amendments to or re-enactments of such statute or regulations in force from time to time, and, if applicable, any statute or regulation that supplements or supersedes such statute or regulation;
- (h) References to a specific Recital, Article or Section shall, unless something in the subject matter or context is inconsistent therewith, be construed as references to that specific Recital, Article or Section of this Plan, whereas the terms "this Plan", "hereof", "herein", "hereto", "hereunder" and similar expressions shall be deemed to refer generally to this Plan and not to any particular Recital, Article, Section or other portion of this Plan and include any documents supplemental hereto; and
- (i) The word "or" is not exclusive.

1.3 Governing Law

This Plan shall be governed by and construed in accordance with the laws of Ontario and the federal laws of Canada applicable therein. All questions as to the interpretation or application of this Plan and all proceedings taken in connection with this Plan and its provisions shall be subject to the jurisdiction of the Court.

1.4 Currency

Unless otherwise stated, all references in this Plan to sums of money are expressed in, and all payments provided for herein shall be made in, US Dollars.

1.5 Date for Any Action

If the date on which any action is required to be taken hereunder by a Person is not a Business Day, such action shall be required to be taken on the next succeeding day which is a Business Day.

1.6 Time

Time shall be of the essence in this Plan.

ARTICLE 2- PURPOSE AND EFFECT OF THIS PLAN

2.1 Purpose

The purpose of this Plan is to facilitate the continuation of the business of the Jaguar Group as a going concern, address certain liabilities of the Applicant, and effect a recapitalization and financing transaction on an expedited basis to provide a stronger financial foundation for the Jaguar Group going forward and additional liquidity to allow the Jaguar Group to continue to work towards its operational and financial

goals from and after the implementation Date in the expectation that all Persons with an economic interest in the Jaguar Group will derive a greater benefit from the implementation of this Plan than would otherwise result.

<u>83</u>

2.2 Effectiveness

Subject to the satisfaction, completion or walver (to the extent permitted pursuant to Section 12.4) of the conditions precedent set out herein, this Plan will become effective in the sequence described in Section 7.4 from and after the Implementation Time and shall be binding on and enure to the benefit of the Jaguar Group, the Affected Unsecured Creditors, all Existing Equity Holders, all holders of Equity Claims, the Released Parties, the Noteholder Released Parties and all other Persons as provided for herein, or subject to, this Plan and their respective successors and assigns and their respective heirs, executors, administrators and other legal representatives, successors and assigns.

2.3 Persons Not Affected

For greater certainty, except as provided in Sections 11.1(a)(iii), 11.1(b)(l), 11.2, 12.2(c) and 13.1, this Pian does not affect the holders of Excluded Claims to the extent of those Excluded Claims. Nothing in this Plan shall affect the Jaguar Group's rights and defences, both legal and equitable, with respect to any Excluded Claims, including, but not limited to, all rights with respect to legal and equitable defences or entitlements to set-offs or recoupments against such Excluded Claims. Nothing herein shall constitute a waiver of any right of either the Monitor or the Applicant to dispute the quantum of an Excluded Claim.

ARTICLE 3- CLASSIFICATION, VOTING CLAIMS AND RELATED MATTERS

3.1 Classes

For the purposes of considering and voting on the Plan Resolution, there shall be one class of stakeholders, consisting of Affected Unsecured Creditors (the "Affected Creditor Class").

3.2 Meeting

- (a) The Meeting shall be held in accordance with this Plan, the Meeting Order and any further Order in the CCAA Proceedings. Subject to the terms of any further Order in the CCAA Proceedings, the only Persons entitled to notice of, to attend or to speak at the Meeting are the Eligible Voting Creditors (or their respective duly appointed proxyholders), representatives of the Monitor, the Applicant, the Consenting Noteholders, all such parties' financial and legal advisors, the Chair (as defined in the Meeting Order), the Secretary (as defined in the Meeting Order) and the Scrutineers (as defined in the Meeting Order). Any other person may be admitted to the Meeting only by invitation of the Applicant or the Chair.
- (b) For the purposes of voting at the Meeting, each Affected Unsecured Creditor (including a Beneficial Noteholder with respect to its Noteholder's Allowed Claim) shall be entitled to one vote as a member of the Affected Creditor Class.
- (c) For the purposes of voting at the Meeting, the Voting Claim of any Beneficial Noteholder shall be deemed to be equal to its Noteholder's Allowed Claim as at the Noteholder Voting Record Date. Registered Holders of Notes, in their capacities as such, will not be entitled to vote at the Meeting.

3.3 Required Majority

In order to be approved, this Plan must receive the affirmative vote of the Required Majority of the

Affected Creditor Class.

3.4 Excluded Claims

Excluded Creditors shall not be entitled to vote or (except as otherwise expressly stated in the Meeting Order) attend in respect of their Excluded Claims at any meeting to consider and approve this Plan.

84

3.5 Existing Equity Holders and Holders of Equity Claims

Existing Equity Holders and holders of Equity Claims shall not be entitled to attend or vote in respect of their Equity Claims at any meeting to consider and approve this Plan.

3.6 Crown Claims

All Crown Claims in respect of all amounts that were outstanding at the Filing Date shall be paid in full to the Crown within six months of the Sanction Order, as required by subsection 6(3) of the CCAA.

3.7 Payments to Employees

Immediately after the date of the Sanction Order, the Applicant will pay in full all Employee Priority Claims, if any, to its employees and former employees.

ARTICLE 4 - ELECTIONS AND SHARE OFFERING

4.1 Participation In Share Offering

- (a) Each Noteholder that is an Eligible Investor shall be entitled to participate in the Share Offering.
- (b) Pursuant to and in accordance with the Meeting Order, there shall be delivered an election form (an "Election Form") to each Participant Holder of the Notes, as of the Subscription Record Date, together with instructions to deliver such Election Form (or copies thereof) to the applicable Beneficial Noteholders to the extent such Participant Holder is not also the Beneficial Noteholder of such Notes. Each Eligible Investor shall have the right, but not the obligation, to irrevocably elect to exercise its Subscription Privilege, with such subscription to be conditioned upon the implementation of this Plan and effective on the implementation Date in accordance with Section 7.4. In order to exercise its Subscription Privilege, such Eligible Investor shall return, or cause to be returned, the duly executed Election Form (including a Rep Letter) in accordance with the Meeting Order, so that it is received by the Solicitation/Election Agent on or before the Election Deadline.
- (c) An Electing Eligible Investor that is also a Backstop Party may elect, in accordance with the Election Form, to have its Backstop Commitment reduced by the total funds that such Electing Eligible Investor deposits into escrow on or before the Electing Eligible Investor Funding Deadline in respect of Offering Shares that such Electing Eligible Investor subscribes for pursuant to the exercise of all or part of its Subscription Privilege, provided that such Backstop Commitment shall not be reduced below zero (the "Backstop Commitment Reduction Election", with a Backstop Party so electing being a "Commitment Reduction Electing Backstopper").
- (d) Following the issuance of the Sanction Order, but in any event by 5:00 p.m. on the tenth Business Day prior to the expected Implementation Date, Jaguar shall Inform each Electing Eligible Investor of (i) the expected Implementation Date, (ii) the number of

Offering Shares that, subject to compliance with the procedures described in this Plan, will be acquired by such Electing Eligible Investor on the Implementation Date pursuant to the Subscription Privilege; and (iii) the amount of funds (in cash) required to be deposited in escrow with the Escrow Agent by such Electing Eligible Investor to purchase such Offering Shares pursuant to the Share Offering (the "Electing Eligible Investor Funding Amount") by the Electing Eligible Investor Funding Deadline.

(e) Each Electing Eligible Investor must deposit its Electing Eligible Investor Funding Amount in escrow with the Escrow Agent so that it is received by the Escrow Agent by no later than 11:00 a.m. on the seventh Business Day prior to the expected Implementation Date (the "Electing Eligible Investor Funding Deadline"). If an Electing Eligible Investor deposits less than the full amount of its Electing Eligible Investor Funding Amount by the Electing Eligible Investor Funding Deadline, then (i) the funds so deposited by such Electing Eligible Investor shall be returned to such Electing Eligible Investor within five Business Days following the Electing Eligible Investor Funding Deadline; and (ii) such Eligible Investor shall be deemed to have ceased, as of the Electing Eligible Investor Funding Deadline, to be an Electing Eligible Investor and Its subscription for Offering Shares pursuant to the Subscription Privilege and right to receive Accrued Interest Offering Shares shall be null and void.

- (f) As soon as practicable but in any event no later than 11:00 a.m. one Business Day after the Electing Eligible Investor Funding Deadline, Jaguar shall inform each Backstop Party (other than a Backstop Party in respect of whom the Backstop Agreement has been terminated) of (i) the total number of Offering Shares not validly subscribed for pursuant to the Subscription Privilege (the "Backstopped Shares"), (ii) the number of Backstopped Shares to be acquired by such Backstop Party pursuant to its Backstop Commitment, based upon its Pro Rata Share of the Backstopped Shares, and (iii) the amount of funds (by way of cash or a letter of credit) required to be deposited in escrow with the Escrow Agent by such party to purchase such Backstopped Shares (the "Backstop Payment Amount") by the Backstop Funding Deadline.
- (g) Each Backstop Party (other than a Backstop Party in respect of whom the Backstop Agreement has been terminated) shall deliver to the Escrow Agent and the Escrow Agent shall have received, not later than 2:00 p.m. (Toronto time) on the day that is five Business Days prior to the expected implementation Date (the "Backstop Funding Deadline"), either:
 - cash in an amount equal to the full amount of such Backstop Party's Backstop Payment Amount; or
 - (ii) a letter of credit, in form and substance reasonably satisfactory to Jaguar, having a face amount equal to such Backstop Party's Backstop Payment Amount, and Issued by a financial institution having an equity market capitalization of at least \$10,000,000,000 and a credit rating of at least A+ from Standard & Poor's or A1 from Moody's,

in each case: (1) to be held in escrow in accordance with the Escrow Agreement until all conditions to the Share Offering have been satisfied or waived in accordance with the Backstop Agreement and with irrevocable instructions to use such cash or letter of credit, as applicable, to the extent required to enable such Backstop Party to comply with its Backstop Purchase Obligation; and (2) provided for greater certainty that, if a Backstop Party (A) has exercised all or part of its Subscription Privilege and has paid its Electing Eligible Investor Funding Amount on or before the Electing Eligible Investor Funding Deadline, and (B) is a Commitment Reduction Electing Backstop Party shall not be required to zero, such Backstop Party shall not be required to

DOCSTOR: 2926319\2A DOCSTOR: 2926319\2A

20

deliver cash or a letter of credit to the Escrow Agent.

- (h) An Electing Eligible Investor who complies with Section 4.1(e) (the "Participating Eligible Investor") shall participate in the Share Offering and shall be deemed to have subscribed for Offering Shares in an amount equal to the Electing Eligible Investor
 Funding Amount deposited in escrow with the Escrow Agent by that Participating Eligible Investor in accordance with Section 4.1(e) divided by the Subscription Price (the "Participating Eligible Investor Shares").
- (i) Each Funding Backstop Party shall be deemed to have subscribed for its Pro Rata Share of the Backstopped Shares.
- (j) On or prior to the implementation Date, Jaguar shall inform: (i) each Participating Eligible Investor of the number of Accrued interest Offering Shares to be allocated to such Participating Eligible Investor in accordance with section 5.1(b); and (ii) each Funding Backstop Party of the number of Accrued Interest Offering Shares and the number of Backstop Commitment Shares to be allocated to such Funding Backstop Party in accordance with section 5.1(b).
- In the event of a Backstop Default/Termination, provided that the Backstop Agreement (k) remains in full force and effect with respect to other Backstop Parties thereafter, Jaguar shall, in accordance with the Backstop Agreement, provide the applicable Backstop Parties, or such other parties acceptable to the Backstop Parties and Jaguar in accordance with the Backstop Agreement that will execute a Backstop Consent Agreement, with an opportunity to assume the obligations (and rights) of a Defaulting Backstop Party, Objecting Backstop Party, Breaching Backstop Party or Non-Delivering Backstop Party, as applicable, in each case in accordance with and subject to the terms and conditions of this Plan and the Backstop Agreement. Any Assuming Backstop Party shall comply with its obligations in connection with its Assumed Backstop Commitment and shall be entitled to receive the applicable Offered Shares under this Plan in connection with such Assumed Backstop Commitment, subject to such Assuming Backstop Party having complied with its obligations under this Plan and the Backstop Agreement and such other terms and conditions under this Plan and the Backstop Agreement. For greater certainty, any Assuming Backstop Party that has complied with its obligations under this Plan and the Backstop Agreement shall constitute and be treated as a Funding Backstop Party for purposes of this Plan.

ARTICLE 5 - TREATMENT OF CLAIMS

5.1 Treatment of Noteholders

- (a) For the purposes of distributions under this Plan, the Distribution Claim of any Beneficial Noteholder shall be deemed to be equal to its Noteholder's Allowed Claim.
- (b) On the Implementation Date and in accordance with the steps and sequence as set forth in this Plan, each Noteholder shall and shall be deemed to irrevocably and finally exchange its Notes for the following consideration which shall and shall be deemed to be received in full and final settlement of its Notes and its Noteholder's Allowed Claim:
 - (i) its Pro Rata Share of the Unsecured Creditor Common Shares;
 - (ii) its Pro Rata Share of the Early Consent Shares, if such Noteholder is an Early Consenting Noteholder;
 - (iii) its Pro Rata Share of Accrued Interest Offering Shares if such Noteholder is a

Participating Eligible Investor and/or a Funding Backstop Party, provided that in no event shall a Participating Eligible Investor or a Funding Backstop Party receive a greater number of Accrued Interest Offering Shares than Offering Shares (including Backstopped Shares, as applicable) received by such person. Any Accrued Interest Offering Shares remaining after the allocation of the Accrued Interest Offering Shares to Participating Eligible Investors and Funding Backstop Parties pursuant to the immediately preceding sentence shall be reallocated among those Participating Eligible Investors and/or Funding Backstop Parties who have received less Accrued Interest Offering Shares than Offering Shares (including Backstopped Shares, as applicable) on a *pro rata* basis based on Accrued Interest Claims of such Participating Eligible Investors and/or Funding Backstop Parties (calculated as at the Record Date); and 87

- (iv) its Pro Rata Share of the Backstop Commitment Shares, if such Noteholder is a Funding Backstop Party.
- (c) On the implementation Date and in accordance with the steps and sequence as set forth in this Plan, each Participating Eligible Investor shall receive its Participating Eligible Investor Shares and each Funding Backstop Party shall receive its Pro Rata Share of the Backstopped Shares.
- (d) After giving effect to the terms of this Section 5.1, the obligations of Jaguar with respect to the Notes of each Noteholder shall, and shall be deemed to, have been irrevocably and finally extinguished and each Noteholder shall have no further right, title or interest in or to the Notes or its Noteholder's Allowed Claim.

5.2 Treatment of Affected Unsecured Creditors Other Than Noteholders

- (a) On the Implementation Date and in accordance with the steps and sequence as set forth in this Plan, each Affected Unsecured Creditor (except for a Noteholder in respect of its Noteholder's Allowed Claim, which shall be dealt with in accordance with Section 5.1) shall receive its Pro Rata Share of the Unsecured Creditor Common Shares and shall be deemed to irrevocably and finally exchange its Affected Unsecured Claim for its Pro Rata Share of the Unsecured Creditor Common Shares, which shall and shall be deemed to be received in full and final settlement of its Affected Unsecured Claim.
- (b) After giving effect to the terms of this Section 5.2, the obligations of Jaguar with respect to such Affected Unsecured Creditor's Affected Unsecured Claim shall, and shall be deemed to, have been irrevocably and finally extinguished and such Affected Unsecured Creditor shall have no further right, title or interest in or to the Affected Unsecured Claim.
- (c) With the consent of the Monitor and the Majority Backstop Parties, an Affected Unsecured Creditor with an Allowed Affected Unsecured Claim who is not a Noteholder may be entitled to participate in the Share Offering for Its Pro Rata Share of the Offering Shares (calculated as if the Affected Unsecured Creditor's Allowed Affected Unsecured Claim was a Noteholder's Allowed Claim); provided that any such Affected Unsecured Creditor completes and submits an Election Form and Rep Letter on or prior to the Election Deadline and complies with all of the obligations of a Participating Eligible Investor in accordance with the terms and conditions of the Plan, including without limitation Section 4.1(e) hereof, in which case, such Affected Unsecured Creditor shall be treated as an Eligible Investor for the purpose of the Offering Shares and each Eligible Investor's Subscription Privilege will be adjusted accordingly.

5.3 Treatment of Existing Equity Holders

- (a) Each Existing Shareholder shall retain its Existing Shares subject to the Common Share Consolidation pursuant to Section 7.4(a) and in accordance with the steps and sequences set forth herein.
- (b) Pursuant to this Plan and in accordance with the steps and sequences set forth herein, all Existing Share Options, Rights and the Shareholder Rights Plan shall be cancelled and shall be deemed to be cancelled without the need for any repayment of capital thereof or any other liability, payment or compensation therefor and for greater certainty, no holders of Existing Share Options or Rights shall be entitled to receive any interest, dividends, premium or other payment in connection therewith.

5.4 Equity Claims

All Equity Claims shall be fully, finally and irrevocably and forever compromised, released, discharged, cancelled and barred on the Implementation Date. Holders of Equity Claims shall not receive any consideration or distributions under this Plan and shall not be entitled to vote on this Plan at the Meeting. Notwithstanding the foregoing, Existing Shareholders shall be entitled to continue to hold their Existing Shares in accordance with the terms of this Plan, subject to the Common Share Consolidation.

5.5 Claims of the Trustees

The Trustees' claims under Section 6.07 of the 4.5% Convertible Note Indenture and the 5.5% Convertible Note Indenture shall be unaffected by this Plan.

5.6 Application of Plan Distributions

- (a) All amounts paid or payable hereunder on account of the Noteholders Allowed Claim (including, for greater certainty, any securities received hereunder) shall be applied as follows: (i) first, in respect of the principal amount of the obligations to which such Noteholders Allowed Claim relate, and (ii) second, if such principal amounts have been fully repaid, in respect of any accrued but unpaid interest on such obligations.
- (b) In the event that a Funding Backstop Party is not a Noteholder, such Funding Backstop Party shall receive its Backstop Commitment Shares as a fee.

ARTICLE 6 - MEETING

6.1 Meeting

The Meeting to consider and vote on this Plan shall be conducted in accordance with the terms of the Claims Procedure Order and the Meeting Order.

6.2 Acceptance of Plan

If this Plan is approved by the Required Majority entitled to vote at the Meeting, then this Plan shall be deemed to have been agreed to, accepted and approved by the Affected Unsecured Creditors and shall be binding upon all Affected Unsecured Creditors, if the Sanction Order is granted and the conditions described in Section 12.3 hereof have been satisfied or waived, as applicable.

DOCSTOR: 2926319\2A DOCSTOR: 2926319\2A

ARTICLE 7 - IMPLEMENTATION

7.1 Administration Charge

On the Implementation Date, all outstanding, invoiced obligations, liabilities, fees and disbursements secured by the Administration Charge shall be fully paid by the Applicant. Upon receipt by the Monitor of confirmation from each of the beneficiarles of the Administration Charge that payments of the amounts secured by the Administration Charge have been made, the Monitor shall file a certificate with the Court confirming same and thereafter, the Administration Charge shall be and be deemed to be discharged from the assets of the Applicant, without the need for any other formality.

7.2 Corporate Authorizations

The adoption, execution, delivery, implementation and consummation of all matters contemplated under this Plan involving corporate action of any members of the Jaguar Group will occur and be effective as of the Implementation Date (or such other date as Jaguar and the Majority Consenting Noteholders may agree), and will be authorized and approved under this Plan and by the Court, where appropriate, as part of the Sanction Order, in all respects and for all purposes without any requirement of further action by shareholders, directors or officers of the Jaguar Group. All necessary approvals to take actions shall be deemed to have been obtained from the directors or the shareholders of the Jaguar Group, as applicable, including the deemed passing by any class of shareholders of any resolution or special resolution and no shareholders' agreement or agreement between a shareholder and another Person limiting in any way the right to vote shares held by such shareholder or shareholders with respect to any of the steps contemplated by this Plan shall be deemed to be effective and no such agreement shall have any force or effect.

7.3 Fractional Interests

No certificates representing fractional Common Shares shall be allocated under this Plan, and fractional share interests shall not entitle the owner thereof to vote or to any rights of a shareholder of Jaguar. Any legal, equitable, contractual and any other rights or claims (whether actual or contingent, and whether or not previously asserted) of any Person with respect to fractional Common Shares pursuant to this Plan shall be rounded down to the nearest whole number without compensation therefor.

7.4 Implementation Date Transactions

Commencing at the Implementation Time, the following events or transactions will occur, or be deemed to have occurred and be taken and effected, in the following order in five minute increments and at the times set out in this Section 7.4 (or in such other manner or order or at such other time or times as Jaguar and the Majority Consenting Noteholders may agree, acting reasonably), without any further act or formality required on the part of any Person, except as may be expressly provided herein:

- (a) Articles of Reorganization shall be filed under the OBCA to amend the articles of Jaguar to effect a consolidation (the "Common Share Consolidation") of the issued and outstanding Common Shares on the basis of one post-consolidation Common Share for each Consolidation Number of Common Shares outstanding immediately prior to the Common Share Consolidation. Any fractional Interests in the consolidated Common Shares will, without any further act or formality, be cancelled without payment of any consideration therefor. Following the completion of such consolidation, the stated capital of the Common Shares shall be equal to the stated capital of the Common Shares immediately prior to consolidation.
- (b) The following shall occur concurrently:
 - (i) the Rights and the Shareholder Rights Plan shall be cancelled and shall be

DOCSTOR: 2926319\2A DOCSTOR: 2926319\2A

deemed to be cancelled without the need for any repayment of capital thereof or any other compensation therefor and shall cease to be of any further force or effect; 90

- (ii) any and all Existing Share Options and the Stock Option Plan shall be cancelled and shall be deemed to be cancelled without the need for any repayment of capital thereof or any other compensation therefor and shall cease to be of any further force or effect;
- (iii) if the DSU/RSU/SAR Notice is delivered, the DSU Plan, the RSU Plan and/or the SAR Plan, as set out in the DSU/RSU/SAR Notice shall be cancelled and shall be deemed to be cancelled without the need for any repayment of capital thereof or any other compensation therefor and shall cease to be of any further force or effect; and
- (iv) all Equity Claims shall be fully, finally, irrevocably and forever compromised, released, discharged, cancelled and barred without any consideration or distributions therefor.
- (c) In exchange for, and in full and final settlement of, the Noteholders Allowed Claim as at the Implementation Date, Jaguar shall issue:
 - (i) to each Noteholder its Pro Rata Share of Unsecured Creditor Common Shares;
 - (ii) to each Early Consenting Noteholder its Pro Rata Share of the Early Consent Shares;
 - (iii) to each Participating Eligible Investor and Funding Backstop Party the number of Accrued Interest Offering Shares such Participating Eligible Investor or Funding Backstop Party is entitled to receive in accordance with Section 5.1(b); and
 - (iv) to each Funding Backstop Party, its Pro Rata Share of the Backstop Commitment Shares,

which New Common Shares shall be distributed in the manner described in Section 8.2 hereof. Upon issuance of these New Common Shares, the Noteholders Allowed Claim shall and shall be deemed to be irrevocably and finally extinguished and such Noteholder shall have no further right, title or interest in and to the Notes or its Noteholder's Allowed Claim.

- (d) The Notes and the Indentures will not entitle any Noteholder to any compensation or participation other than as expressly provided for in this Plan and shall be cancelled and will thereupon be null and void, and the obligations of the Applicant thereunder or in any way related thereto shall be satisfied and discharged, except to the extent expressly set forth in section 6.07 of the Indentures, which section shall remain in effect until two months following the Implementation Date or such later date agreed to by the Applicant, the Monitor, the Trustees and the Majority Consenting Noteholders.
- (e) In exchange for, and in full and final settlement of, its Affected Unsecured Claim, Jaguar shall issue to each Affected Unsecured Creditor, other than the Noteholders, its Pro Rata Share of the Unsecured Creditor Common Shares;
- (f) The following shall occur concurrently:
 - (i) Jaguar shall issue to each Participating Eligible Investor its Participating Eligible

DOCSTOR: 2926319\2A DOCSTOR: 2926319\2A

Investor Shares in accordance with Section 5.1(c) hereof in consideration for its Electing Eligible Investor Funding Amount, which Participating Eligible Investor Shares shall be distributed in the manner described in Section 8.2 hereof; and

- (ii) Jaguar shall issue to each Funding Backstop Party the number of Backstopped Shares such Funding Backstop Party is entitled to receive in accordance with Section 5.1(c) hereof in consideration for such Funding Backstop Party's Backstop Payment Amount, which Backstopped Shares shall be distributed in the manner described in Section 8.2 hereof.
- (g) The releases and injunctions referred to in Section 11 shall become effective.

LINE IN TRANSMITTER PROVINCES OF A CONTRACT AND A

- (h) The directors of Jaguar immediately prior to the Implementation Time shall be deemed to have resigned and the New Board shall be deemed to have been appointed.
- (i) The Escrow Agent shall be deemed to be holding the Electing Eligible Investor Funding Amounts and the Backstop Payment Amounts for Jaguar and shall release from escrow such amounts to Jaguar in accordance with the Escrow Agreement.
- (j) Jaguar shall pay: (i) all of the reasonable fees and expenses of the Advisors for services rendered to the Ad Hoc Committee up to and including the implementation Date, (ii) the reasonable accrued and unpaid third party expenses of any of the Consenting Noteholders up to an amount agreed to by the Majority Backstop Parties; (iii) the fees and expenses of Jaguar's financial advisors in connection with the transactions contemplated under this Plan pursuant to their engagement letter, as amended, with Jaguar, subject to a maximum amount agreed to by the Majority Backstop Parties, (iv) the reasonable fees and expenses of Jaguar's Canadian and U.S. legal advisors and legal advisor to the special committee of the board of directors of Jaguar, and (v) amounts owing to the Trustees under Section 6.07 of the 4.5% Convertible Note Indenture

ARTICLE 8 - ISSUANCE AND DISTRIBUTION OF NEW COMMON SHARES

8.1 Issuance of New Common Shares

الاعترمين الدروانين وتتترك وتشتخ

All New Common Shares issued and outstanding as part of the implementation of this Plan shall be deemed to be issued and outstanding as fully-paid and non-assessable. The amount added to the stated capital of the Common Shares as a result of the issuance of New Common Shares in accordance with this Plan shall be equal to the fair market value of the consideration received by Jaguar for the issuance of such New Common Shares.

8.2 Delivery of New Common Shares

- (a) Jaguar shall use its commercially reasonable best efforts to cause the delivery of the New Common Shares to be distributed under this Plan no later than the second Business Day following the Implementation Date (or such other date as Jaguar and the Majority Consenting Noteholders may agree).
- (b) The Notes are held by DTC (as sole Registered Holder) through its nominee company CEDE & Co. DTC will surrender, or will cause the surrender of, the certificates, if any, representing the Notes to the Trustees in exchange for New Common Shares as contemplated in this Plan.
- (c) The delivery of Unsecured Creditor Common Shares to Noteholders in exchange for the Notes will be made through the facilities of DTC to Participant Holders who, in turn will

make delivery of the Unsecured Creditor Common Shares to the Beneficial Noteholders pursuant to standing instructions and customary practices of DTC. If for any reason the New Common Shares are not DTC eligible, then the delivery of the Unsecured Creditor Common Shares shall be made (at the election of Jaguar with the consent of the Monitor and the Majority Consenting Noteholders) either (i) by delivery of a Direct Registration System Advice to each Noteholder or (ii) by delivery of a share certificate to each Noteholder, in either case based on registration instructions received by, or on behalf of, the Monitor from Participant Holders in such manner as the Monitor determines reasonable in the circumstances.

يداد الاحتجاز فالمتعاد المارية المراجع المراجع المراجع

- (d) The delivery of Early Consent Shares to Early Consenting Noteholders will be made (at the election of Jaguar with the consent of the Monitor and the Majority Consenting Noteholders) either; (i) by delivery of a Direct Registration System Advice to each Early Consenting Noteholder; or (ii) by delivery of a share certificate to each Early Consenting Noteholder, in any case based on registration and delivery instructions contained in the Rep Letter.
- (e) The delivery of Offering Shares, Backstopped Shares, Backstop Commitment Shares and Accrued Interest Offering Shares to the Participating Eligible Investors and the Funding Backstop Parties will be made (at the election of Jaguar with the consent of the Monitor and the Majority Consenting Noteholders) either (I) by delivery of a Direct Registration System Advice to each Participating Eligible Investor and Funding Backstop Party or (ii) by delivery of a share certificate to each Participating Eligible Investor and Funding Backstop Party, in either case based on registration and delivery instructions contained in the Election Forms in the case of Participating Eligible Investors and in the Rep Letter In the case of Funding Backstop Parties.
- (f) The delivery of New Common Shares to Affected Unsecured Creditors (other than Noteholders) in consideration for their Affected Unsecured Claims will be made (at the election of Jaguar with the consent of the Monitor and the Majority Consenting Noteholders) either (i) by delivery of a Direct Registration System Advice to each of the Affected Unsecured Creditors (other than Noteholders) or (ii) by delivery of a share certificate to each of the Affected Unsecured Creditors (other than Noteholders), in either case based on registration and delivery instructions received by the Monitor pursuant to the Claims Procedure Order and the Meeting Order.
- (g) Jaguar, the Monitor and the Trustees will have no liability or obligation in respect of all deliveries from DTC, or its nominee, to Participant Holders or from Participant Holders to Beneficial Noteholders.
- (h) Upon receipt of and in accordance with written instructions from the Monitor, the Trustees shall instruct DTC to, and DTC shall: (i) establish an escrow position representing the respective positions of the Noteholders as of the Implementation Date for the purpose of making distributions to the Noteholders on and after the Implementation Date; and (ii) block any further trading in the Notes, effective as of the close of business on the Distribution Record Date, all in accordance with the customary practices and procedures of DTC.
- (i) Unless a securities law legend is not required by US Securities Laws, the Direct Registration System Advices and share certificates delivered pursuant to this Section 8.2 shall have legends affixed thereon in substantially the form provided for in the Rep Letter.

8.3 Undeliverable Distributions

If any distribution of New Common Shares is undeliverable (that is for greater certainty that cannot be

properly registered or delivered to the intended recipient because of inadequate or incorrect registration or delivery information or otherwise) (an **"Undeliverable Distribution**") it shall be delivered to the Escrow Agent, which shall hold such Undeliverable Distribution in escrow, and administered in accordance with this Section 8.3. No further distributions in respect of an Undeliverable Distribution shall be made unless and until the Escrow Agent is notified by the applicable Person of its current address and/or registration information, as applicable, at which time the Escrow Agent shall make such distributions to such Person. All claims for Undeliverable Distributions must be made on or before the date that is the 365th day following the Implementation Date, after which the right to receive distributions under this Plan In respect of such an Undeliverable Distribution shall be fully, finally, irrevocably and forever compromised, released, discharged, cancelled and barred without any compensation therefor, notwithstanding any federal, provincial, or state laws to the contrary, and any New Common Shares that are the subject of such Undeliverable Distribution shall be cancelled.

ARTICLE 9 - RELEASE OF FUNDS FROM ESCROW

9.1 Release of Funds from Escrow

The Escrow Agent shall release any Electing Eligible Investor Funding Amounts and Backstop Payment Amounts, or portions thereof, as follows and in accordance with the terms of the Escrow Agreement:

- (a) If an Electing Eligible Investor deposits less than the full amount of its Electing Eligible Investor Funding Amount by the Electing Eligible Investor Funding Deadline, such party shall cease to be an Electing Eligible Investor and the Escrow Agent shall return such funds so deposited by such Electing Eligible Investor to such Electing Eligible Investor in accordance with Section 4.1(e) hereof.
- (b) On the Implementation Date, the Escrow Agent shall release from escrow to Jaguar, at the applicable time, the applicable Electing Eligible Investor Funding Amounts and Backstop Payment Amounts pursuant to and in accordance with Section 7.4 hereof.
- (c) If this Plan is terminated for any reason or not implemented in accordance with the terms hereof by the Outside Date, the Escrow Agent shall as soon as practicable return all Electing Eligible Investor Funding Amounts and Backstop Payment Amounts to the applicable Participating Eligible Investors and Funding Backstop Parties.
- (d) If any Electing Eligible Investor or Funding Backstop Party provides to the Escrow Agent more than its applicable Electing Eligible Investor Funding Amount or Backstop Payment Amount under this Plan, the Escrow Agent shall as soon as practicable return any excess funds to such Electing Eligible Investor or Funding Backstop Party.

ARTICLE 10 – PROCEDURE FOR DISTRIBUTIONS REGARDING DISPUTED DISTRIBUTION CLAIMS

10.1 No Distribution Pending Allowance

DOCSTOR: 2926319\2A DOCSTOR: 2926319\2A

An Affected Unsecured Creditor holding a Disputed Distribution Claim will not be entitled to receive a distribution under the Plan in respect of such Disputed Distribution Claim or any portion thereof unless and until, and then only to the extent that, such Disputed Distribution Claim becomes an Allowed Affected Unsecured Claim.

10.2 Distributions After Disputed Distribution Claims Resolved

(a) Distributions of Unsecured Creditor Common Shares in relation to a Disputed Distribution Claim of an Affected Unsecured Creditor will be held by the Applicant, in a segregated account constituting the Disputed Distribution Claims Reserve, for the benefit of the Affected Unsecured Creditors with Allowed Affected Unsecured Claims until the final determination of the Disputed Distribution Claim in accordance with the Claims Procedure Order and this Plan.

94

- (b) To the extent that any Disputed Distribution Claim becomes an Allowed Affected Unsecured Claim in accordance with this Plan, the Applicant shall distribute to the holder of such Allowed Affected Unsecured Claim, that number of Unsecured Creditor Common Shares from the Disputed Distribution Claims Reserve equal to such Affected Unsecured Creditor's Pro Rata Share of Unsecured Creditor Common Shares.
- (c) On the date that all Disputed Distribution Claims have been finally resolved in accordance with the Claims Procedure Order and any required distributions contemplated in section (b) have been made, if (i) the aggregate number of Unsecured Creditor Common Shares remaining in the Disputed Distribution Claims Reserve is less than 14,000, the Applicant shall cancel those Unsecured Creditor Common Shares; or (ii) the aggregate number of Unsecured Creditor Common Shares; or (iii) the aggregate number of Unsecured Creditor Common Shares; or (ii) the aggregate number of Unsecured Creditor Common Shares remaining in the Disputed Distribution Claims Reserve is equal to or greater than 14,000, the Applicant shall distribute such Unsecured Creditor Common Shares to the Affected Unsecured Creditors with Allowed Affected Unsecured Claims such that after giving effect to such distributions each such Affected Unsecured Creditor has received its applicable Pro Rata Share of such Unsecured Creditor Common Shares.

ARTICLE 11- RELEASES

11.1 Release

- (a) On the implementation Date, the following shall be fully, finally, irrevocably and forever compromised, released, discharged, cancelled and barred:
 - (i) all Affected Unsecured Claims;
 - (ii) all Equity Claims;
 - (iii) all Director/Officer Claims other than Continuing Other Director/Officer Claims and Non-Released Director/Officer Claims and also (for greater certainty) excluding Section 5.1(2) Director/Officer Claims and any Agreed Excluded Director/Officer Litigation Claims; provided that any Section 5.1(2) Director/Officer Claims and any Agreed Excluded Director/Officer Litigation Claims shall be limited to recovery from any insurance proceeds payable in respect of such Section 5.1(2) Director/Officer Claims or Agreed Excluded Director/Officer Litigation Claims, as applicable, pursuant to the Director/Officer Insurance Policies, and any Persons with any such Section 5.1(2) Director/Officer Claims or Agreed Excluded Director/Officer Litigation Claims shall have no right to, and shall not, make any claim or seek any recoveries from any Person (including Jaguar, any of Its Subsidiaries or any Director or Officer), other than enforcing such Person's rights to be paid from the proceeds of a Director/Officer Insurance Policy by the applicable Insurer(s); provided that nothing in this Plan shall prejudice, compromise, release or otherwise affect any right or defence of any insurer in respect of a Director/Officer Insurance Policy or any insured in respect of a Section 5.1(2) Director/Officer Claim or Agreed Excluded Director/Officer Litigation Claim; and
 - (Iv) all Director/Officer Indemnity Claims.

DOCSTOR: 2926319\2A DOCSTOR: 2926319\2A

On the Implementation Date, the Applicant, the Subsidiaries, and each of their respective financial advisors, legal counsel and agents, the Monitor, legal counsel to the Monitor, and legal counsel to the special committee of the board of directors of Jaguar (collectively, the "Released Parties") shall be released and discharged from any and all demands, claims, actions, causes of action, counterclaims, suits, debts, sums of money, accounts, covenants, damages, judgments, expenses, executions, liens and other recoveries on account of any indebtedness, liability, obligation, demand or cause of action of whatever nature that any Person (including any Person who may claim contribution or indemnification against or from them) may be entitled to assert, whether known or unknown, matured or unmatured, direct, indirect or derivative, foreseen or unforeseen, existing or hereafter arising, based in whole or in part on any act or omission, transaction, dealing or other occurrence existing or taking place on or prior to the Implementation Date that are in any way relating to, arising out of or in connection with (i) the restructuring, disclaimer, resiliation, breach or termination of any contract, lease, agreement or other arrangement, whether written or oral; (ii) the business and affairs of the Applicant or the Subsidiarles; (iii) the Notes; (iv) the Indentures; (v) the Existing Shares; (vi) the Existing Share Options; (vii) the Shareholder Rights Plan; (viii) Equity Claims; (ix) the Support Agreement; (x) the Backstop Agreement; (xi) this Plan; or (xii) the CCAA Proceedings; provided, however, that nothing in this Section 11.1 will release or discharge:

95

- the Applicant or any of the Subsidiaries from or in respect of (x) any Excluded (i) Claim. (v) its obligation to Affected Unsecured Creditors under this Plan or under any Order, or (z) its obligations under the Backstop Agreement or the Support Agreement; provided that any Agreed Excluded Jaguar Litigation Claims shall be limited to recovery from any insurance proceeds payable in respect of such Agreed Excluded Jaguar Litigation Claims pursuant to the Jaguar Insurance Policies, and any Persons with any such Agreed Excluded Jaguar Litigation Claims against the Applicant shall have no right to, and shall not, make any claim or seek any recoveries from any Person (including Jaguar, any of its Subsidiaries or any Director or Officer), other than enforcing such Person's rights to be paid from the proceeds of a Jaguar Insurance Policy by the applicable Insurer(s); provided further that nothing in this Plan shall prejudice, compromise, release or otherwise affect any right or defence of any insurer in respect of a Jaguar Insurance Policy or any insured in respect of an Agreed Excluded Jaguar Litigation Claim; or
- a Released Party if the Released Party is adjudged by the express terms of a judgment rendered on a final determination on the merits to have committed fraud or wilful misconduct.
- (c) At the Implementation Time, each of the Noteholders, the Ad Hoc Committee, the Trustees, and each of their respective present and former shareholders, officers, directors, and the Advisors and the Trustees' counsel (collectively, the "Noteholder Released Parties") will be released and discharged from any and all demands, claims, actions, causes of action, counterclaims, suits, debts, sums of money, accounts, covenants, damages, judgments, expenses, executions, liens and other recoveries on account of any indebtedness, liability, obligation, demand or cause of action of whatever nature that any Person (including any Person who may claim contribution or Indemnification against or from them) may be entitled to assert whether known or unknown, matured or unmatured, direct, indirect or derivative, foreseen or unforeseen, existing or hereafter arising, based in whole or in part on any act or omission, transaction, dealing or other occurrence existing or taking place on or prior to the Implementation Date that are In any way relating to, arising out of or In connection with: (I) the Notes; (II)

DOCSTOR; 2926319\2A DOCSTOR; 2926319\2A

(b)

the Indentures; (iii) the Existing Shares; (iv) the Existing Share Options; (v) the Shareholder Rights Plan; (vi) Equity Claims; (vii) the Support Agreement; (viii) the Backstop Agreement; (ix) this Plan; or (x) the CCAA Proceedings, and any other matters or actions related directly or indirectly to the foregoing; provided that nothing in this Section 11.1(c) will release or discharge a Noteholder Released Party in respect of their obligations under this Plan, the Backstop Agreement, the Support Agreement, any Election Form and provided further that nothing in this Section 11.1(c) will release or discharge a Noteholder Released Party is adjudged by the express terms of a judgment rendered on a final determination on the merits to have committed fraud or wilful misconduct.

11.2 Injunctions

All Persons (regardless of whether or not such Persons are Affected Unsecured Creditors) are permanently and forever barred, estopped, stayed and enjoined, on and after the Implementation Time, with respect to any and all Released Claims or Noteholder Released Claims, from (i) commencing, conducting or continuing in any manner, directly or indirectly, any action, suits, demands or other proceedings of any nature or kind whatsoever (including, without limitation, any proceeding in a judicial, arbitral, administrative or other forum) against the Released Parties, the Named Directors and Officers and the Noteholder Released Parties; (ii) enforcing, levying, attaching, collecting or otherwise recovering or enforcing by any manner or means, directly or indirectly, any judgment, award, decree or order against the Released Parties, the Named Directors and Officers and Noteholder Released Parties or their property; (iii) commencing, conducting or continuing in any manner, directly or indirectly, any action, suits or demands, including without limitation, by way of contribution or indemnity or other relief, in common law, or In equity, or for breach of trust or breach of fiduciary duty or under the provisions of any statute or regulation, or other proceedings of any nature or kind whatsoever (including, without limitation, any proceeding in a judicial, arbitral, administrative or other forum) against any Person who makes such a claim or might reasonably be expected to make such a claim, in any manner or forum, against one or more of the Released Parties, the Named Directors and Officers and Noteholder Released Parties; (iv) creating, perfecting, asserting or otherwise enforcing, directly or indirectly, any lien or encumbrance of any kind against the Released Parties, the Named Directors and Officers and Noteholder Released Parties or their property; or (v) taking any actions to interfere with the implementation or consummation of this Plan; provided, however, that the foregoing shall not apply to the enforcement of any obligations under this Plan. For greater certainty, the provisions of this Section 11.2 shall apply to Section 5.1(2) Director/Officer Claims, Agreed Excluded Director/Officer Litigation Claims and Agreed Excluded Jaguar Litigation Claims in the same manner as Released Claims, except to the extent that the rights of a holder of such Section 5.1(2) Director/Officer Claims, Agreed Excluded Director/Officer Litigation Claims and/or Agreed Excluded Jaguar Litigation Claims to enforce such claims against an insurer in respect of a Directors/Officer Insurance Policy and/or a Jaguar Insurance Policy, as applicable, are expressly preserved pursuant to Section 11.1(a)(iii) and/or Section 11.1(b)(i) hereof.

11.3 Timing of Releases and Injunctions

All releases and injunctions set forth in this Article 11 shall become effective on the implementation Date at the time or times and in the manner set forth in Section 7.4 hereof.

11.4 Knowledge of Claims

Each Person to which Section 11.1 hereof applies shall be deemed to have granted the releases set forth in Section 11.1 notwithstanding that it may hereafter discover facts in addition to, or different from, those which it now knows or believes to be true, and without regard to the subsequent discovery or existence of such different or additional facts, and such party expressly waives any and all rights that it may have under any applicable law which would limit the effect of such releases to those Claims or causes of action known or suspected to exist at the time of the granting of the release.

ARTICLE 12 - COURT SANCTION, CONDITIONS PRECEDENT AND IMPLEMENTATION

12.1 Application for Sanction Order

If this Plan is approved by the Required Majority, the Applicant shall apply for the Sanction Order on the date set for the hearing for the Sanction Order or such later date as the Court may set.

12.2 Sanction Order

The Sanction Order shall, among other things, declare that:

- (i) this Plan has been approved by the Required Majority entitled to vote at the Meeting in conformity with the CCAA; (ii) the Applicant acted in good faith and has complied with the provisions of the CCAA and the Orders made in the CCAA Proceedings in all respects;
 (iii) the Court is satisfied that the Applicant has not done nor purported to do anything that is not authorized by the CCAA; and (Iv) this Plan and the transactions contemplated by it are fair and reasonable;
- (b) this Plan (including the arrangements and releases set out herein) has been sanctioned and approved pursuant to section 6 of the CCAA and will be binding and effective as herein set out on the Applicant, all Affected Unsecured Creditors, all holders of Equity Claims and all other Persons as provided for in this Plan or in the Sanction Order;
- (c) subject to the performance by the Applicant of its obligations under this Plan, and except to the extent expressly contemplated by this Plan or the Sanction Order, all obligations or agreements to which the Applicant is a party immediately prior to the Implementation Time, will be and shall remain in full force and effect as at the Implementation Date, unamended except as they may have been amended by agreement of the parties subsequent to the Filing Date, and no Person who is a party to any such obligations or agreements shall, following the Implementation Date, accelerate, terminate, rescind, refuse to perform or otherwise repudiate its obligations thereunder, or enforce or exercise any right (including any right of set-off, option, dilution or other remedy) or make any demand under or in respect of any such obligation or agreement, by reason of:
 - (i) any defaults or events of default arising as a result of the insolvency of the Applicant prior to the Implementation Date;
 - (ii) any change of control of the Applicant arising from implementation of this Plan (except in respect of existing, written senior officer and employee employment agreements of Persons who remain senior officers and employees of Jaguar as of the Implementation Date and any payments due under such agreements, which may only be waived by the senior officers and employees who are parties to such agreements);
 - (iii) the fact that the Applicant has sought or obtained relief under the CCAA or that this Plan has been implemented by the Applicant;
 - (Iv) the effect on the Applicant of the completion of any of the transactions contemplated by this Plan;
 - (v) any compromises or arrangements effected pursuant to this Plan; or
 - (vi) any other event(s) which occurred on or prior to the Implementation Date which

DOCSTOR: 2926319\2A DOCSTOR: 2926319\2A

would have entitled any Person to enforce rights and remedies, subject to any express provisions to the contrary in any agreements entered into with the Applicant after the Filing Date.

For greater certainty, nothing in this paragraph 12.2(c) shall waive, compromise or discharge any obligations of the Applicant in respect of any Excluded Claim;

- (d) the commencement or prosecution, whether directly, indirectly, derivatively or otherwise, of any demands, claims, actions, counterclaims, suits, judgment, or other remedy or recovery as described in Section 11.2 hereof shall be permanently enjoined;
- (e) the releases effected by this Plan shall be approved, and declared to be binding and effective as of the Implementation Date upon all Affected Unsecured Creditors, holders of Equity Claims and all other Persons affected by this Plan and shall enure to the benefit of all such Persons;
- (f) from and after the Implementation Date, all Persons with an Affected Unsecured Claim shall be deemed to (i) have consented and agreed to all of the provisions of this Plan as an entirety; and (ii) each Affected Unsecured Creditor shall be deemed to have granted, and executed and delivered to the Applicant all consents, releases, assignments and waivers, statutory or otherwise, required to implement and carry out this Plan in its entirety.

12.3 Conditions to Plan Implementation

The implementation of this Plan shall be conditional upon the fulfillment, satisfaction or waiver (to the extent permitted by Section 12.4 hereof) of the following conditions:

- (a) The Court shall have granted the Sanction Order, the operation and effect of which shall not have been stayed, reversed or amended, and in the event of an appeal or application for leave to appeal, final determination shall have been made by the applicable appellate court;
- (b) No Applicable Law shall have been passed and become effective, the effect of which makes the consummation of this Plan illegal or otherwise prohibited;
- All necessary judicial consents and any other necessary or desirable third party consents, if any, to deliver and implement all matters related to this Plan shall have been obtained;
- (d) All documents necessary to give effect to all material provisions of this Plan (Including the Sanction Order, this Plan, the Share Offering and the Common Share Consolidation and all documents related thereto) shall have been executed and/or delivered by all relevant Persons in form and substance satisfactory to the Applicant and the Majority Consenting Noteholders;
- (e) All required stakeholder, regulatory and Court approvals, consents, waivers and filings shall have been obtained or made, as applicable, on terms satisfactory to the Majority Consenting Noteholders and the Company, each acting reasonably and in good faith;
- (f) All senior officer and employee employment agreements shall have been modified to reflect the revised capital structure of Jaguar following implementation of the Plan, Including, without limitation, to provide that the Implementation of the Plan does not constitute a change of control under such employment agreements, and no change of control payments shall be owing or payable to Jaguar's officers or employees in

connection with the implementation of the Plan;

- (g) The DSU Plan, the RSU Plan and the SAR Plan shall have been addressed in a manner acceptable to Jaguar and the Majority Consenting Noteholders;
- (h) The Articles of Reorganization shall have been filed under the OBCA;
- (i) All material filings under applicable Laws shall have been made and any material regulatory consents or approvals that are required in connection with this Plan shall have been obtained and, in the case of waiting or suspensory periods, such waiting or suspensory periods shall have expired or been terminated;
- (j) The New Common Shares shall have been conditionally approved for listing on the TSX, the TSXV or such other Designated Offshore Securities Market acceptable to the Majority Consenting Noteholders without any vote or approval of the Existing Shareholders, subject only to receipt of customary final documentation;
- (k) All conditions to implementation of this Plan set out in the Support Agreement (which for greater certainty include the conditions set out in sections 9(a), (b) and (c) of the Support Agreement) shall have been satisfied or waived in accordance with their terms and the Support Agreement shall not have been terminated;
- (I) All conditions to implementation of this Plan set out in the Backstop Agreement (which for greater certainty include the conditions set out in sections 7(a), (b) and (c) of the Backstop Agreement) shall have been satisfied or waived in accordance with their terms, and the Backstop Agreement shall not have been terminated;
- (m) The issuance of the Unsecured Creditor Common Shares and Early Consent Shares shall be exempt from registration under the US Securities Act pursuant to the provisions of section 3(a)(10) of the US Securities Act; and
- (n) No insurer under a Director/Officer Insurance Policy or a Jaguar Insurance Policy shall have an unresolved objection, filed in the CCAA Proceedings, to the implementation of this Plan.

12,4 Waiver of Conditions

The Applicant and the Majority Consenting Noteholders may at any time and from time to time waive the fulfillment or satisfaction, in whole or in part, of the conditions set out herein, to the extent and on such terms as such partles may agree to provided however that the conditions set out in Section 12.3(a) cannot be waived and that the conditions set out in Section 12.3(l) can only be waived by the Applicant and the Majority Backstop Parties.

12.5 Implementation Provisions

If the conditions contained in Section 12.3 are not satisfied or waived (to the extent permitted under Section 12.4) by the Outside Date, unless the Applicant and the Majority Consenting Noteholders agree in writing to extend such period, this Plan and the Sanction Order shall cease to have any further force or effect and will not be binding on any Person.

12.6 Monitor's Certificate of Plan Implementation

Upon written notice from the Applicant (or counsel on its behalf) and Goodmans LLP on behalf of the Majority Consenting Noteholders and the Majority Backstop Parties to the Monitor that the conditions to Plan implementation set out in Section 12.3, have been satisfied or waived, the Monitor shall, as soon as

possible following receipt of such written notice, deliver to the Applicant and Goodmans LLP on behalf of the Majority Consenting Noteholders and the Majority Backstop Parties, and file with the Court, a certificate which states that all conditions precedent set out in Section 12.3 have been satisfied or waived and that the Implementation Date has occurred.

ARTICLE 13 – GENERAL

13.1 Waiver of Defaults

Subject to the performance by the Applicant of its obligations under this Plan, and except to the extent expressly contemplated by this Plan or the Sanction Order, no Person who is a party to any obligations or agreements with the Applicant or any Subsidiary shall, following the Implementation Date, accelerate, terminate, rescind, refuse to perform or otherwise repudiate its obligations thereunder, or enforce or exercise any right (including any right of set-off, option, dilution or other remedy) or make any demand under or in respect of any such obligation or agreement, by reason of:

- (a) any defaults or events of default arising as a result of the insolvency of the Applicant prior to the Implementation Date;
- (b) any change of control of the Applicant or any Subsidiary arising from implementation of this Plan (except in respect of existing, written senior officer and employee employment agreements of Persons who remain senior officers and employees of Jaguar as of the implementation Date and any payments due under such agreements, which may only be waived by the senior officers and employees who are parties to such agreements);
- (c) the fact that the Applicant has sought or obtained relief under the CCAA or that this Plan has been implemented by the Applicant;
- (d) the effect on the Applicant or any Subsidiary of the completion of any of the transactions contemplated by this Plan;
- (e) any compromises or arrangements effected pursuant to this Plan; or
- (f) any other event(s) which occurred on or prior to the Implementation Date which would have entitled any Person to enforce rights and remedies, subject to any express provisions to the contrary in any agreements entered into with the Applicant after the Filing Date.

For greater certainty, nothing in this paragraph 13.1 shall walve, compromise or discharge any obligations of the Applicant in respect of any Excluded Claim.

13.2 Deeming Provisions

In this Plan, the deeming provisions are not rebuttable and are conclusive and irrevocable.

13.3 Non-Consummation

The Applicant reserves the right to revoke or withdraw this Plan at any time prior to the Implementation Date, with the consent of the Monitor and the Majority Consenting Noteholders.

If the Implementation Date does not occur on or before the Outside Date (as the same may be extended in accordance with the terms hereof and of the Support Agreement), or If this Plan is otherwise withdrawn in accordance with its terms: (a) this Plan shall be null and void in all respects, and (b) nothing contained in this Plan, and no acts taken in preparation for consummation of this Plan, shall (i) constitute or be

deemed to constitute a waiver or release of any Clalms by or against the Jaguar Group, their respective successors or any other Person; (ii) prejudice in any manner the rights of the Jaguar Group, their respective successors or any other Person in any further proceedings involving the Jaguar Group or their respective successors; or (iii) constitute an admission of any sort by the Jaguar Group, their respective successors or any other Person.

13.4 Modification of Plan

(a) The Applicant may, at any time and from time to time, amend, restate, modify and/or supplement this Plan with the consent of the Monitor and the Majority Consenting Noteholders, provided that: any such amendment, restatement, modification or supplement must be contained in a written document that is filed with the Court and: 1 () 1

- (i) if made prior to or at the Meeting: (A) the Monitor, the Applicant or the Chair (as defined in the Meeting Order) shall communicate the details of any such amendment, restatement, modification and/or supplement to Affected Unsecured Creditors and other Persons present at the Meeting prior to any vote being taken at the Meeting; (B) the Applicant shall provide notice to the service list of any such amendment, restatement, modification and/or supplement and shall file a copy thereof with the Court forthwith and in any event prior to the Court hearing in respect of the Sanction Order; and (C) the Monitor shall post an electronic copy of such amendment, restatement, modification and/or supplement on the Monitor's Website forthwith and in any event prior to the Court hearing in respect of the Sanction Order; and
- (ii) if made following the Meeting: (A) the Applicant shall provide notice to the service list of any such amendment, restatement, modification and/or supplement and shall file a copy thereof with the Court; (B) the Monitor shall post an electronic copy of such amendment, restatement, modification and/or supplement on the Monitor's Website; and (C) such amendment, restatement, modification and/or supplement shall require the approval of the Court following notice to the Affected Unsecured Creditors.
- (b) Notwithstanding Section 13.4(a) hereof, any amendment, restatement, modification or supplement may be made by the Applicant: (i) if prior to the date of the Sanction Order, with the consent of the Monitor and the Majority Consenting Noteholders; and (ii) if after the date of the Sanction Order, with the consent of the Monitor and the Majority Consenting Noteholders and upon approval by the Court, provided in each case that it concerns a matter that, in the opinion of the Applicant, acting reasonably, is of an administrative nature required to better give effect to the implementation of this Plan and the Sanction Order or to cure any errors, omissions or ambiguities and is not materially adverse to the financial or economic interests of the Affected Unsecured Creditors.
- (c) Any amended, restated, modified or supplementary plan or plans of compromise filed with the Court and, if required by this section, approved by the Court, shall, for all purposes, be and be deemed to be a part of and incorporated in this Plan.

13.5 Severability of Plan Provisions

If, prior to the Implementation Time, any term or provision of this Plan is held by the Court to be invalid, vold or unenforceable, at the request of the Applicant, made with the consent of the Majority Consenting Noteholders (acting reasonably), the Court shall have the power to either (a) sever such term or provision from the balance of this Plan and provide the Applicant and the Majority Consenting Noteholders with the option to proceed with the Implementation of the balance of this Plan as of and with effect from the Implementation Time, or (b) alter and interpret such term or provision to make it valid or enforceable to
102

the maximum extent practicable, consistent with the original purpose of the term or provision held to be invalid, void or unenforceable, and such term or provision shall then be applicable as altered or interpreted, provided that the Majority Consenting Noteholders have approved such alteration or interpretation, acting reasonably. Notwithstanding any such holding, alteration or interpretation, and provided that this Plan is implemented, the remainder of the terms and provisions of this Plan shall remain in full force and effect and shall in no way be affected, impaired or invalidated by such holding, alteration or interpretation.

13.6 Preservation of Rights of Action

Except as otherwise provided in this Plan or in the Sanction Order, or in any contract, instrument, release, indenture or other agreement entered into in connection with this Plan, following the Implementation Date, the Applicant will retain and may enforce, sue on, settle, or compromise (or decline to do any of the foregoing) all claims, rights or causes of action, suits and proceedings, whether in law or in equity, whether known or unknown, that the Applicant may hold against any Person or entity without further approval of the Court.

13.7 Responsibilities of Monitor

FTI Consulting Canada Inc. is acting in its capacity as Monitor in the CCAA Proceedings with respect to the Applicant and this Plan and not in its personal or corporate capacity, and will not be responsible or liable for any obligations of the Applicant under the Plan or otherwise.

13.8 Notices

Any notice or other communication to be delivered hereunder must be in writing and refer to this Plan and may, as hereinafter provided, be made or given by personal delivery, ordinary mail, email or by facsimile addressed to the respective Partles as follows:

If to Jaguar or the Subsidiaries, at:

(a) c/o Jaguar Mining Inc. 67 Yonge Street, Suite 1203 Toronto, Ontario M5E 1J8

> Attention: David Petroff Email: david.petroff@jaguarmining.com

with a required copy (which shall not be deemed notice) to:

Norton Rose Fulbright Canada LLP Royal Bank Plaza, South Tower, Suite 3800 200 Bay Street P.O. Box 84 Toronto, Ontario M5J 2Z4

Attention:	Walied Soliman and Evan Cobb
Fax:	(416) 216-3930
Email:	walled.soliman@nortonrosefulbright.com
	evan.cobb@nortonrosefulbright.com

(b)

If to the Ad Hoc Committee of Noteholders:

DOCSTOR: 2926319\2A DOCSTOR: 2926319\2A

Goodmans LLP Suite 3400 333 Bay Street Bay Adelaide Centre Toronto, Ontario M5H 2S7

Attention: Fax: Email Rob Chadwick and Melaney Wagner (416) 979-1234 <u>rchadwick@goodmans.ca</u> <u>mwagner@goodmans.ca</u>

(c) If to the Monitor, at:

FTI Consulting Canada Inc. TD Waterhouse Tower Suite 2010 79 Wellington Street Toronto, Ontario M5K 1G8

Attention:	Greg Watson and Jodi Porepa
Fax:	(416) 649-8101
Email:	Greg.Watson@fticonsulting.com
	Jodi.Porepa@fticonsulting.com

With a required copy (which shall not be deemed notice) to:

Oster, Hoskin & Harcourt LLP Box 50 1 First Canadian Place Toronto, Ontario M5X 1B8

Attention:Marc WassermanFax:(416) 862-6666Email:<u>mwasserman@osler.com</u>

or to such other address as any Party may from time to time notify the others in accordance with this Section 13.8. Any such communication so given or made shall be deemed to have been given or made and to have been received on the day of delivery if delivered, or on the day of faxing or emailing, provided that such day in either event is a Business Day and the communication is so delivered, faxed or emailed before 5:00 p.m. on such day. Otherwise, such communication shall be deemed to have been given and made and to have been received on the next following Business Day.

13.9 Consent of Majority Consenting Noteholders or Majority Backstop Parties

For the purposes of this Plan, any matter requiring the agreement, waiver, consent or approval of the Majority Consenting Noteholders or the Majority Backstop Parties shall be deemed to have been agreed to, waived, consented to or approved by such Majority Consenting Noteholders or Majority Backstop Parties if such matter is agreed to, waived, consented to or approved in writing by Goodmans LLP, provided that Goodmans LLP expressly confirms in writing (which can be by way of e-mail) that it is providing such agreement, consent, waiver or approval on behalf of the Majority Consenting Noteholders or the Majority Backstop Parties, as applicable.

DOCSTOR: 2926319\2A DOCSTOR: 2926319\2A

13.10 Paramountcy

From and after the Implementation Time on the Implementation Date, any conflict between:

- (a) this Plan; and
- (b) the covenants, warranties, representations, terms, conditions, provisions or obligations, expressed or implied, of any contract, mortgage, security agreement, indenture, trust indenture, note, loan agreement, commitment letter, agreement for sale, lease or other agreement, written or oral and any and all amendments or supplements thereto existing between any Person and the Applicant and/or the Subsidiaries as at the Implementation Date,

will be deemed to be governed by the terms, conditions and provisions of this Plan and the Sanction Order, which shall take precedence and priority.

13.11 Further Assurances

Notwithstanding that the transactions and events set out herein will occur and be deemed to occur in the order set out in this Plan without any further act or formality, each of the Persons named or referred to in, or subject to, this Plan will make, do and execute, or cause to be made, done and executed, all such further acts, deeds, agreements, transfers, assurances, instruments or documents as may reasonably be required by any of them to carry out the full intent and meaning of this Plan and to give effect to the transactions contemplated herein.

Schedule "A"

Agreed Excluded Director/Officer Litigation Claims

(i) Claims asserted by Daniel R. Titcomb; Robert J. Lloyd; James M. Roller; William E. Dow; Jeffrey Kirchhoff and/or Brazilian Resources, Inc. (collectively, the "New Hampshire Litigation Plaintiffs") in the proceeding in the United States District Court for the District of New Hampshire bearing Civil Action No. 1:13-cv-00428-JL against Gary E. German, Gilmour Clausen, John Andrews, Richard Falconer, David Petroff and Frederick Hermann (collectively, the "Director/Officer Defendants"); and

(ii) those claims of the New Hampshire Litigation Plaintiffs against the Director/Officer Defendants as allowed in accordance with and subject to the terms of paragraph 3 of the Minutes of Settlement dated February 5, 2014 between the New Hampshire Litigation Plaintiffs and Jaguar Mining Inc.

DOCSTOR: 2926319\2A

1

Schedule "B"

Agreed Excluded Jaguar Litigation Claims

(i) Claims asserted by Daniel R. Titcomb; Robert J. Lloyd; James M. Roller; William E. Dow; Jeffrey Kirchhoff and/or Brazilian Resources, Inc. (collectively, the "New Hampshire Litigation Plaintiffs") in the proceeding in the United States District Court for the District of New Hampshire bearing Civil Action No. 1:13-cv-00428-JL against Jaguar Mining Inc.; and

(ii) those claims of the New Hampshire Litigation Plaintiffs against Jaguar Mining Inc. as allowed in accordance with and subject to the terms of paragraph 4 of the Minutes of Settlement dated February 5, 2014 between the New Hampshire Litigation Plaintiffs and Jaguar Mining Inc.

Schedule "C"

Agreed Excluded Litigation Claimants

Daniel R. Titcomb

Robert J. Lloyd

James M. Roller

William E. Dow

Jeffrey Kirchhoff

Brazilian Resources, Inc.

DOCSTOR: 2926319\2A

1 1

1

මේන් මොස්ම්

108

. Court File No, CV-13-10383-00CL

ONTARIO SUPERIOR COURT OF JUSTICE COMMERCIAL LIST

IN THE MATTER OF THE COMPANIES' CREDITORS ARRANGEMENT ACT, R.S.C. 1985, c. C-36, AS AMENDED

AND IN THE MATTER OF A PLAN OF COMPROMISE OR ARRANGEMENT OF JAGUAR MINING INC.

Applicant

MONITOR'S CERTIFICATE

RECITALS

A. Pursuant to an order of the Court dated December 23, 2013 (the "**Initial Order**"), Jaguar Mining Inc. (the "**Applicant**") filed for and obtained protection from its creditors under the *Companies' Creditors Arrangement Act*, R.S.C. 1985, c. C-36, as amended (the "**CCAA**").

B. Pursuant to the Initial Order, FTI Consulting Canada Inc. was appointed the Monitor of the Applicant (the "Monitor") with the powers, duties and obligations set out in the Initial Order.

C. Also on December 23, 2013, the Applicant sought and the Court granted a Meeting Order and a Claims Procedure Order.

D. The Applicant has filed a Plan of Compromise and Arrangement under the CCAA, dated December 23, 2013 (as amended and restated on January 31, 2014, and as may be further amended, the "Plan"), which Plan has been approved by the required majorities of Affected Unsecured Creditors and sanctioned by the Court; and

E. Unless otherwise indicated herein, initially capitalized terms used herein have the meaning set out in the Plan.

DOCSTOR; 2911021\3B

THE MONITOR HEREBY CERTIFIES that it has received written notice from the Applicant and Goodmans LLP (on behalf of the Majority Consenting Noteholders and the Majority Backstop Parties) that the conditions precedent set out in Section 12.3 of the Plan have been satisfied or waived in accordance with the Plan on February _____, 2014 and that the implementation Date has occurred; and

This Certificate was delivered by the Monitor at ______ [filmE] on February ____, 2014.

FTI CONSULTING CANADA INC., in its capacity as Monitor of Jaguar Mining Inc. and not in its personal or corporate capacity 1-1-0

ų

Ву:

Name: Title:

DOCSTOR; 2911021/3A

5 6 4 T

SCHEDULE "C"

1

л.

;

٦

. . .

Ontarlo Dorporation Number Numèro de la sociátió en Ontarló

1591416

For Ministry Use Daiy A trugage exclusif du ministère

المتقطبينين فسيصيفون للرجارة فطاورهم التراج وجروان ورجا

.

.

» - «الجادية والمرجعة المحافظ الموادي ومن تشاريه التروي التروي الم

.

Form B Businese	1.	î'he Dé(1 19 11 11 11 11 11 11 11 11 11 11 11 11	nie d Inałi	of the long a	e çai Icela	rpota la de	ι∦αη ‡ • /α #φ	: (8) alété	et dui i 7 (Éi	:1 . 9	i na	ם אי	abr	ra:	ነ ዋቸ	'TËS	AN 39) ルと9				l'i);					-	نرغب هاياه	وللتحد
Corporellana Act		_	A		L	1.	R			: N	1	М	1 .			N													
Formula A Lai aur leo socialithe per Actions																							-				•		
	P.	Not	UY#Å	ទៃ ៨៩	inor	nina	lion s	rporal <i>iociali</i> <u>vient</u>	i (de	f c'ha Hi so	ngac olété	by שיק 	lha i Na e	org n m	(411)2 0 <i>0(11)</i>	alloi Ise p	n;(- xa/s	Set c <i>Çila</i> r+m	de k	n 181. 9 740	DOK Ngar	CA Iseli	4719 (ההי) ריייי	41. (Éc	.ETT xire	ren en	8) LET	TRE	9
								_	1				_ .		 							_							
		<u> </u>	 						1,				 	*							_	_							
~		-	ļ				ļ		+			-	-		-*														
	Ë.	L Da	ie ol	<u>}</u> է Լուժ	 סמיט	naik	on/ar	palgá	_). пыйс))) I I I	Date	de l	a co	Miar	l Ufioi	1	da l	i ius	 1/4/1	 						L	I	1	
			003															, mei			- - •	•••	_,		+				
	4 ,		a reg 014			tion	₩4¢	orderi	κi þi	/ the	çour	t oh	446.1 / Le	ាបបាព COL	ግድ ር	y i m idor	ነክቂ እስቂ	ha réa	orgu	** n l96	<i>tio</i> n	,ka							
		r		*****						*****							-	ļņ¢f8,											
,		¦'ar	non	hathr	a d	e (q	oour	ie Dit oo <i>nsi</i>	Ua .	^p antk	axa 1	/An.															90 p	onro	mø
	ច,	fn t Co.	aooo níon	rðar méin	109 16/11	with 14 Pe	iha (ndor	기der Inano	(¢† i« 9 de	iorge réorg	(1)[2]a [#11]9	tion <i>ailei</i>	ihe o N les	filol ște	ea a tula	l iné da li	9 00 (8 50 8 50	pola ciáid	tion isov	1)\$1'4 7 <i>t 1</i> 11	ате ор‼/4	nda 9 di	a (ti) q at	i (ol lege	lowa In el	i) Vivel	nte (
	1	0/1 PL	e r	08	t-c(^`) NSC NY	olidi Fran	idatk ation dona allty,	Cu at te) Ministerie	non seta	-Sh In	iare the	fo bc	n Su Su Su	8,84 511d:	986 ate	4 C	sei loh	ieü 1ma	אתם sn s	a c She	uts Ireș	sar W	iair ill,	ig Wit	hoi hoi	nn	OH.
																												•	
	1																												

1.00				-	117
					нJ
4					۴.
			•		
				•	
				•	
• · ·					
* ·					
These suicies are sui Les présents déluis : exemplaire.	bmlünd under section 188 of iha f sont déposés en vertu de l'ariidie	ueineau Oorpotallons Ac 196 de la Loi eur les sociét 196 de la Loi eur les sociét	f and ars signed in dupli és par adions, <i>lis sòni sig</i>	oste, nés en double	
JAGUAR MINI	ING INC.				
	Name of Ocrooration / D	mominalien acelale de la su	ន/វៀ ត		
₿y/ Par:					
₿y/ Par;			•		
	šignalure i Signature		Deacebilon of Office / Force	allau	
1	Signature		Description of Differ / Fonc	ellon	
	Signature i Signature		Description of Office / Fonc	an a	
	Sløndlure i Slønature		Deacchilon of Driton / Fond		
	Slønnlure i Slønature		Deaception of Office / Force		
	Slønnlure i Slønature		Description of Office / Fond		
1	Slønnlure i Slønature		Deactollon of Dilloo / Fonc		
	Slønnlure i Slønature		Deactollon of Difleo / Fonc		
	ŝlonglure i Signature		Deaception of Dilles / Fonc	Won	
1	ŝlonglure i Signature		Deactollon of Office / Fond	ψοη	
1	šlonglure i Signature		Description of Office / Forc		 -
1	Slønnlure i Slønature		Deaceloilon of Dilloo / Fonc		
1	Slønnlure i Slønature		Deactollon of Diffeo / Fonc		 -
1	êlgnqlure i Signature		Deactollon of Diffeo / Fonc		

. . -. .

• .

.. . ..

.

, `

.

тор 1971 — Т 1971 — Т		
•••	IN THE MATTER OF THE <i>COMPANIES' CREDITORS ARRANGEMENT ACT</i> , R.S.C. 1985, c. C-36, AS AMENDED AND IN THE MATTER OF A PLAN OF COMPROMISE OR ARRANGEMENT OF JAGUAR MINING INC. (Applicant)	Court File No: CV-13-10383-00CL
		ONTARIO SUPERIOR COURT OF JUSTICE (COMMERCIAL LIST) Proceeding commenced at Toronto
		ORDER (PLAN SANCTION)
		Norton Rose Fulbright Canada LLP Royal Bank Plaza, South Tower, Suite 3800 200 Bay Street, P.O. Box 84 Toronto, Ontario M5J 2Z4 CANADA
		Tony Reyes LSUC#: 28218V Tel: 416.216.4825 Email: <u>tony.reyes@nortonrosefulbright.com</u>
		Evan Cobb LSUC#: 55787N Tel: 416.216.1929 Email: <u>evan.cobb@nortonrosefulbright.com</u>
		Fax: 416.216.3930
		Lawyers for the Applicant, Jaguar Mining Inc.
	BecsToR: 396273946	114

i .

Exhibit D

Court File No. CV-13-10383-00CL

This is Exhibitreferred to in the elident of I. DOUGLAS WILLOCK swarn before me, this... \mathfrak{A} FEBRUARY

ONTARIO SUPERIOR COURT OF JUSTICE

ULED ALL ALL ALL AND AND A

IN THE MATTER OF THE COMPANIES' CREDITORS ARRANGEMENT ACT, R.S.C. 1985, c. C-36, AS AMENDED

AND IN THE MATTER OF A PLAN OF COMPROMISE AND ARRANGEMENT OF JAGUAR MINING INC.

AMENDED AND RESTATED PLAN OF COMPROMISE AND ARRANGEMENT

PURSUANT TO THE COMPANIES' CREDITORS ARRANGEMENT ACT

OF JAGUAR MINING INC.

FEBRUARY 5, 2014

RECITALS

- (A) Jaguar Mining Inc. (the "**Applicant**" or "**Jaguar**") is a debtor company (as such term is defined in the *Companies' Creditors Arrangement Act*, R.S.C. 1985, c. C-36, as amended (the "**CCAA**").
- (B) On December 23, 2013, the Honourable Justice Morawetz of the Ontario Superior Court of Justice (Commercial List) (the "**Court**") granted the following Orders pursuant to the CCAA:
 - (i) an Initial Order in respect of the Applicant (as such Order may be amended, restated or varied from time to time, the "**Initial Order**");
 - (ii) a Plan Filing and Meeting Order (as such Order may be amended, restated or varied from time to time, the "Meeting Order") pursuant to which, among other things, the Applicant was authorized to file a plan of compromise and arrangement and to convene a meeting of affected creditors to consider and vote on the plan of compromise and arrangement, as may be amended, restated, modified or supplemented from time to time; and

- (iii) a Claims Procedure Order (as such Order may be amended, restated or varied from time to time, the "**Claims Procedure Order**"), which, among other things, established the procedures by which claims of affected creditors shall be filed in these proceedings.
- (C) This Amended and Restated Plan of Compromise and Arrangement will be filed on February 6, 2014 with the consent of the Majority Consenting Noteholders (as hereinafter defined).
- (D) Mineração Serras Do Oeste Ltda. ("MSOL"), Mineração Turmalina Ltda. ("MTL"), and MCT Mineração Ltda. ("MCT"), each incorporated under the laws of Brazil, are wholly-owned subsidiaries of Jaguar and are not applicants in the CCAA Proceedings.
- (E) The purpose of this Plan is to facilitate the continuation of the business of the Jaguar Group (as hereinafter defined) as a going concern, address certain liabilities of the Applicant, and effect a recapitalization and financing transaction on an expedited basis to provide a stronger financial foundation for the Jaguar Group going forward and additional liquidity to allow the Jaguar Group to continue to work towards its operational and financial goals from and after the Implementation Date in the expectation that all Persons (as hereinafter defined) with an economic interest in the Jaguar Group will derive a greater benefit from the implementation of this Plan than would otherwise result.

NOW THEREFORE the Applicant hereby proposes and presents this Plan under the CCAA.

ARTICLE 1 – INTERPRETATION

1.1 Definitions

In this Plan and the Recitals, unless otherwise stated or unless the subject matter or context otherwise requires:

"4.5% Convertible Note Indenture" means the Indenture dated as of September 15, 2009 among Jaguar, as issuer, The Bank of New York Mellon as trustee and BNY Trust Company of Canada as co-trustee pursuant to which Jaguar issued the 4.5% convertible notes;

***5.5% Convertible Note Indenture**" means the Indenture dated as of February 9, 2011 among Jaguar as issuer, The Bank of New York Mellon as trustee and BNY Trust Company of Canada as co-trustee pursuant to which Jaguar issued the 5.5% convertible notes;

"Accrued Interest Claim" means, with respect to a particular Participating Eligible Investor or Funding Backstop Party, all unpaid interest accrued under the Notes at the applicable rate under the Indentures owing as at the Record Date to such Participating Eligible Investor or Funding Backstop Party;

"Accrued Interest Claims" means the aggregate of all unpaid interest accrued under the Notes at the applicable rate under the Indentures owing as at the Record Date to the Participating Eligible Investors and Funding Backstop Parties;

"Accrued Interest Offering Shares" means 9,044,203 New Common Shares;

"Ad Hoc Committee" means the ad hoc committee of Noteholders represented by the Advisors;

"Administration Charge" has the meaning given to that term in the Initial Order;

"Advisors" means Goodmans LLP, Houlihan Lokey Capital, Inc., Dias Carneiro Advogados, Behre Dolbear & Company (USA), Inc. and Stroock & Stroock & Lavan LLP;

"Affected Creditor Class" has the meaning given to that term in Section 3.1;

"Affected Unsecured Claims" means all Claims against the Applicant that are not Equity Claims;

"Affected Unsecured Creditor" means the holder of an Affected Unsecured Claim in respect of and to the extent of such Affected Unsecured Claim;

"Agreed Excluded Director/Officer Litigation Claims" means any claims against a Director and/or Officer that the Majority Consenting Noteholders and the Applicant have agreed, prior to the Implementation Date, and as set out on Schedule "A" hereto, will constitute Excluded Claims for the purposes of this Plan;

"Agreed Excluded Jaguar Litigation Claims" means any claims against Jaguar that the Majority Consenting Noteholders and the Applicant have agreed, prior to the Implementation Date, and as set out on Schedule "B" hereto, will constitute Excluded Claims for the purposes of this Plan;

"**Agreed Excluded Litigation**" means any proceeding commenced by any Agreed Excluded Litigation Claimant in respect of any Agreed Excluded Litigation Claims, subject to the terms of this Plan;

"Agreed Excluded Litigation Claimants" means any Persons and, if applicable, each of their respective parents, subsidiaries, associated, affiliated and related companies, corporations and Persons, and each of their directors, officers, employees, agents, affiliates, and trustees, that have asserted an Agreed Excluded Director/Officer Litigation Claim and/or an Agreed Excluded Jaguar Litigation Claim, as agreed to by the Majority Consenting Noteholders and the Applicant prior to the Implementation Date and as set out on Schedule "C" hereto;

"Agreed Excluded Litigation Claims" means, collectively, the Agreed Excluded Jaguar Litigation Claims and the Agreed Excluded Director/Officer Litigation Claims;

"Allowed" means, with respect to a Claim, any Claim or any portion thereof that has been finally allowed as a Distribution Claim (as defined in the Claims Procedure Order) for purposes of receiving distributions under this Plan in accordance with the Claims Procedure Order and the CCAA;

"Applicable Law" means, with respect to any Person, property, transaction, event or other matter, any Law relating or applicable to such Person, property, transaction, event or other matter, including, where appropriate, any interpretation of the Law (or any part) by any Person, court or tribunal having jurisdiction over it, or charged with its administration or interpretation;

"Applicant" has the meaning given to that term in Recital A;

"Articles of Reorganization" means the Articles of Reorganization of Jaguar to be filed pursuant to Section 186 of the OBCA and in accordance with Section 7.4(a) hereof, in form and substance satisfactory to Jaguar and the Majority Consenting Noteholders;

"Assumed Backstop Commitment" means, in the event of a Backstop Default/Termination, if any, a Backstop Commitment, or a portion thereof, assumed by an Assuming Backstop Party from a Defaulting Backstop Party, Objecting Backstop Party, Breaching Backstop Party or Non-Delivering Backstop Party, as applicable, in accordance with the terms and conditions of this Plan and the Backstop Agreement;

"Assuming Backstop Party" means, in the event of a Backstop Default/Termination, if any, a Non-Defaulting Backstop Party, Non-Objecting Backstop Party, Non-Breaching/Non-Delivering Backstop Party, or such other party acceptable to the Backstop Parties and Jaguar in each case in accordance with the Backstop Agreement, that executes a Backstop Consent Agreement and that has assumed the obligations (and rights), or a portion thereof, of a Defaulting Backstop Party, Objecting Backstop Party, Breaching Backstop Party or Non-Delivering Backstop Party, as applicable, under the Backstop Agreement, in accordance with the terms and conditions of this Plan and the Backstop Agreement. For greater certainty, any Assuming Backstop Party that has complied with its obligations under this Plan and the Backstop Agreement shall constitute and be treated as a Funding Backstop Party for purposes of this Plan;

"**Backstop Agreement**" means the backstop agreement dated November 13, 2013 (as amended from time to time) between certain Noteholders, Jaguar, MCT, MSOL and MTL, together with any Backstop Consent Agreements executed by other parties from time to time;

"**Backstop Commitment**" means, in respect of each Backstop Party, the commitment set forth on such Backstop Party's signature page to the Backstop Agreement or a Backstop Consent Agreement, as applicable, which commitment may be reduced in accordance with and subject to the terms and conditions of the Backstop Agreement and this Plan;

"Backstop Commitment Reduction Election" has the meaning given to such term in Section 4.1(c);

"Backstop Commitment Shares" means 11,111,111 New Common Shares;

"Backstop Consent Agreement" means an agreement substantially in the form of Schedule B to the Backstop Agreement;

"**Backstop Consideration Commitment**" means, in respect of each Backstop Party, the commitment set forth on such Backstop Party's signature page to the Backstop Agreement or a Backstop Consent Agreement, as applicable, which commitment, for greater certainty, shall not be reduced as a result of a Backstop Commitment Reduction Election;

"Backstop Default/Termination" means any of the following: (a) a breach by a Breaching Backstop Party under section 10(b)(I) or (ii) of the Backstop Agreement in respect of which the Backstop Agreement has been terminated with respect to such Breaching Backstop Party in accordance with its terms; (b) a failure by a Defaulting Backstop Party to meet its obligations in respect of its Backstop Commitment on or before the Backstop Funding Deadline; (c) a failure by a Non-Delivering Backstop Party to deliver an executed Rep Letter to Jaguar by the Election Deadline or if a representation or warranty made in such Rep Letter becomes untrue; and (d) the termination by an Objecting Backstop Party of its obligations under the Backstop Agreement in accordance with section 8(c) thereof;

"Backstop Funding Deadline" has the meaning given to such term in Section 4.1(g);

"**Backstop Parties**" means those Noteholders that have entered into the Backstop Agreement (including a Backstop Consent Agreement), and a "**Backstop Party**" means any one of the Backstop Parties, and their permitted assignees;

"Backstop Payment Amount" has the meaning given to such term in Section 4.1(f);

"Backstop Purchase Obligation" means the obligation of a Backstop Party to purchase Backstopped Shares in accordance with the terms and conditions of the Backstop Agreement and this Plan;

"Backstopped Shares" has the meaning given to such term in Section 4.1(f);

"Beneficial Noteholder" means a beneficial or entitlement holder of Notes holding such Notes in a securities account with a depository, a depository participant or other securities intermediary including, for greater certainty, such depository participant or other securities intermediary only if and to the extent such depository participant or other securities a principal for its own account;

"Bradesco" means Banco Bradesco S.A.;

"**Breaching Backstop Party**" means a Backstop Party that has breached the Backstop Agreement under section 10(b)(i) or (ii) thereof and in respect of whom the Backstop Agreement has been terminated in accordance with its terms;

"Business Day" means any day, other than a Saturday, or a Sunday or a statutory or civic holiday, on which banks are generally open for business in Toronto, Ontario and New York, New York;

"CCAA Proceedings" means the proceedings commenced by the Applicant under the CCAA as contemplated by the Initial Order;

"**CRA Claim**" means the claim as described in the proof of claim, dated January 21, 2014, filed by Canada Revenue Agency in the CCAA Proceedings in the amount of \$5,969.13;

"Charges" has the meaning ascribed thereto in the Initial Order;

"Claim" means:

any right or claim, including any Tax Claim, of any Person that may be asserted or made i. in whole or in part against the Applicant, in any capacity, whether or not asserted or made, in connection with any indebtedness, liability or obligation of any kind whatsoever of the Applicant, and any interest accrued thereon or costs payable in respect thereof, in existence on the Filing Date, or which is based on an event, fact, act or omission which occurred in whole or in part prior to the Filing Date, whether at law or in equity, including by reason of the commission of a tort (intentional or unintentional), by reason of any breach of contract or other agreement (oral or written), by reason of any breach of duty (including, any legal, statutory, equitable or fiduciary duty) or by reason of any equity interest, right of ownership of or title to property or assets or right to a trust or deemed trust (statutory, express, implied, resulting, constructive or otherwise), and together with any security enforcement costs or legal costs associated with any such claim, and whether or not any indebtedness, liability or obligation is reduced to judgment, liquidated, unliquidated, fixed, contingent, matured, unmatured, disputed, undisputed, legal, equitable, secured, unsecured, perfected, unperfected, present or future, known or unknown, by guarantee, warranty, surety or otherwise, and whether or not any right or claim is executory or anticipatory in nature, including any claim arising from or caused by the termination, disclaimer, resiliation, assignment or repudiation by the Applicant of any contract, lease or other agreement, whether written or oral, any claim made or asserted against the Applicant through any affiliate, subsidiary, associated or related person, or any right or ability of any Person to advance a claim for an accounting, reconciliation, contribution, indemnity, restitution or otherwise with respect to any matter, grievance, action (including any class action or proceeding before an administrative tribunal), cause or chose in action, whether existing at present or commenced in the future, and including any other claims that would have been claims provable in bankruptcy had the Applicant become bankrupt on the Filing Date, including for greater certainty any Equity Claim and any claim against the Applicant for indemnification by Director or Officer in respect of a Director/Officer Claim but excluding any such indemnification claims covered by the Directors' Charge (each, a "Pre-filing Claim", and collectively, the "Pre-filing Claims");

ii. any right or claim of any Person against the Applicant in connection with any indebtedness, liability or obligation of any kind whatsoever owed by the Applicant to such Person arising out of the restructuring, disclaimer, resiliation, termination or breach by the Applicant on or after the Filing Date of any contract, lease or other agreement whether written or oral (each, a "Restructuring Period Claim", and collectively, the "Restructuring Period Claims"); and iii. any right or claim of any Person against one or more of the Directors or Officers howsoever arising, whether or not such right or claim is reduced to judgment, liquidated, unliquidated, fixed, contingent, matured, unmatured, disputed, undisputed, legal, equitable, secured, unsecured, perfected, unperfected, present, future, known, or unknown, by guarantee, surety or otherwise, and whether or not such right is executory or anticipatory in nature, including the right or ability of any Person to advance a claim for contribution or indemnity or otherwise with respect to any matter, action, cause or chose in action, whether existing at present or commenced in the future, including any right of contribution or indemnity, for which any Director or Officer is alleged to be by statute or otherwise by law liable to pay in his or her capacity as a Director or Officer (each a "**Director/Officer Claim**", and collectively, the "**Director/Officer Claims**"),

120

in each case other than any Excluded Claim;

"Commitment Reduction Electing Backstopper" has the meaning given to such term in Section 4.1(c);

"Common Share Consolidation" has the meaning given to such term in Section 7.4(a);

"Common Shares" means the common shares in the capital of Jaguar that are duly issued and outstanding at any time;

"Consenting Noteholder" means any Noteholder that has executed the Support Agreement (including a consent agreement substantially in the form of Schedule C thereto), in respect of whom the Support Agreement has not been terminated;

"**Consolidation Number**" means the quotient (to five decimal places) determined by dividing the number of Existing Shares by 1,000,000, which as of the date of this Plan is 86.39636.

"Continuing Other Director/Officer Claims" means Director/Officer Claims against the Other Directors and/or Officers;

"Court" has the meaning given to that term in Recital B;

"Credit Agreement" means the credit agreement made as of December 17, 2012 between Jaguar, as borrower, the Subsidiaries, as guarantors, and Global Resource Fund, as lender.

"**Creditor**" means any Person having a Claim and includes without limitation the transferee or assignee of a Claim transferred and recognized as a Creditor in accordance with the Claims Procedure Order or a trustee, executor, liquidator, receiver, receiver and manager, or other Person acting on behalf of or through such Person;

"Crown" means Her Majesty in right of Canada or a province of Canada;

"Crown Claim" means any Claim of the Crown, for all amounts that were outstanding at the Filing Date and are of a kind that could be subject to a demand under:

- i. subsection 224(1.2) of the ITA;
- any provision of the Canada Pension Plan or of the Employment Insurance Act that refers to subsection 224(1.2) of the ITA and provides for the collection of a contribution, as defined in the Canada Pension Plan, or an employee's premium, or employer's premium, as defined in the Employment Insurance Act, and of any related interest, penalties or other amounts;

- iii. any provision of provincial legislation that has a similar purpose to subsection 224(1.2) of the ITA, or that refers to that subsection, to the extent that it provides for the collection of a sum, and of any related interest, penalties or other amounts, where the sum:
 - a. has been withheld or deducted by a person from a payment to another person and is in respect of a tax similar in nature to the income tax imposed on individuals under the ITA; or
 - b. is of the same nature as a contribution under the Canada Pension Plan if the province is a "province providing a comprehensive pension plan" as defined in subsection 3(1) of the Canada Pension Plan and the provincial legislation establishes a "provincial pension plan" as defined in that subsection.

"**Defaulting Backstop Party**" means a Backstop Party that has failed to meet its obligations in respect of its Backstop Commitment on or before the Backstop Funding Deadline;

"Designated Offshore Securities Market" has the meaning given to that term in Rule 902 of Regulation S.

"Direct Registration System Advice" means, if applicable, a statement delivered by the Transfer Agent or any such Person's agent to any Person entitled to receive New Common Shares pursuant to the Plan indicating the number of New Common Shares registered in the name of or as directed by the applicable Person in a direct registration account administered by the Transfer Agent in which those Persons entitled to receive New Common Shares pursuant to the Plan will hold such New Common Shares in registered form and including, if applicable, a securities law legend;

"Director" means anyone who is or was, or may be deemed to be or have been, whether by statute, operation of law or otherwise, a director or *de facto* director of Jaguar;

"Director Defence Costs Indemnity Claim" means any existing or future right of any current director (as at the date of this Plan) of Jaguar who is a defendant to any Agreed Excluded Director/Officer Litigation Claims against Jaguar for indemnification of reasonable defence costs incurred by such current director of Jaguar (whether or not a director of Jaguar at the time such claim for indemnification is made) in connection with defending against such Agreed Excluded Director/Officer Litigation Claims solely to the extent that such defence costs are not covered by insurance and for which such Director or Officer of Jaguar is entitled to be indemnified by Jaguar;

"Director/Officer Claim" has the meaning given to that term in the definition of Claim;

"Director/Officer Indemnity Claim" means any existing or future right of any Director or Officer of Jaguar against Jaguar that arose or arises as a result of (i) any Person filing a Proof of Claim (as defined in the Claims Procedure Order) in respect of a Director/Officer Claim in respect of such Director or Officer of Jaguar or (ii) any Agreed Excluded Litigation Claims and/or any Agreed Excluded Litigation, in each case for which such Director or Officer of Jaguar is entitled to be indemnified by Jaguar, other than a Director Defence Costs Indemnity Claim;

"Director/Officer Insurance Policy" means any insurance policy pursuant to which any Director or Officer is insured, in his or her capacity as a Director or Officer;

"Directors' Charge" has the meaning given to that term in the Initial Order;

"Disputed Distribution Claim" means an Affected Unsecured Claim (including a contingent Affected Unsecured Claim which may crystallize upon the occurrence of an event or events occurring after the Filing Date) or such portion thereof which has not been allowed as a Distribution Claim (as defined in the Claims Procedure Order), which is validly disputed for distribution purposes in accordance with the Claims

Procedure Order and which remains subject to adjudication for distribution purposes in accordance with the Claims Procedure Order;

"Disputed Distribution Claims Reserve" means the reserve, if any, to be established by the Applicant on the Implementation Date, which shall be comprised of the Unsecured Creditor Common Shares that would have been delivered in respect of Disputed Distribution Claims if such Disputed Distribution Claims had been Allowed Claims as of such date;

"**Disputed Voting Claim**" means an Affected Unsecured Claim (including a contingent Affected Unsecured Claim which may crystallize upon the occurrence of an event or events occurring after the Filing Date) or such portion thereof which has not been allowed as a Voting Claim, which is validly disputed for voting purposes in accordance with the Claims Procedure Order and which remains subject to adjudication for voting purposes in accordance with the Claims Procedure Order;

"**Distribution Claim**" means the amount of the Affected Unsecured Claim of an Affected Unsecured Creditor against the Applicant as finally accepted and determined for distribution purposes in accordance with this Claims Procedure Order and the CCAA;

"Distribution Record Date" means the Business Day immediately before the Implementation Date;

"DSU Plan" means the Deferred Share Unit Plan for non-executive directors adopted in November of 2008 by Jaguar, as amended from time to time;

"**DSU/RSU/SAR Notice**" means a notice delivered by Goodmans to Jaguar prior to the date scheduled for the hearing of the motion for the Sanction Order, if, in satisfaction of Section 12.3(g) hereof, Jaguar and the Majority Consenting Noteholders have agreed to terminate the DSU Plan, the RSU Plan, and/or the SAR Plan;

"DTC" means The Depository Trust Company, or any successor thereof;

"Early Consent Deadline" means November 26, 2013 (or such other date as the Applicant, the Monitor and the Majority Consenting Noteholders may agree);

"Early Consent Shares" means 5,000,000 New Common Shares;

"Early Consenting Noteholder" means any Noteholder that has executed the Support Agreement (including a consent agreement substantially in the form of Schedule C thereto) on or before the Early Consent Deadline and in respect of whom the Support Agreement has not been terminated;

"Election Deadline" means 5:00 p.m. on the second Business Day before the Meeting (or such other time or date as the Applicant and the Majority Consenting Noteholders may agree);

"Election Form" has the meaning given to that term in Section 4.1(b);

"Electing Eligible Investor" means an Eligible Investor who has completed and submitted an Election Form on or prior to the Election Deadline to participate in the Share Offering in accordance with the Meeting Order, provided that an Electing Eligible Investor that irrevocably elects under Section 4.1(b) to participate in the Share Offering and subscribes for such number of Offering Shares that is less than such Eligible Investor's Pro Rata Share of all Offering Shares offered pursuant to the Share Offering shall be deemed to be an Electing Eligible Investor only in respect of such lesser amount, and shall not be treated as an Electing Eligible Investor in respect of the balance;

"Electing Eligible Investor Funding Amount" has the meaning given to that term in Section 4.1(d);

"Electing Eligible Investor Funding Deadline" has the meaning given to that term in Section 4.1(e);

"Eligible Investor" means a person that: (i) is a Noteholder as at the Subscription Record Date; and (ii) has delivered an executed Rep Letter to Jaguar on or before the Election Deadline and the information set forth in such Rep Letter is true and correct as of the Implementation Date, and such person's permitted assignees;

"Eligible Voting Creditors" means Affected Unsecured Creditors holding Voting Claims or Disputed Voting Claims;

"Employee Priority Claims" means the following claims of Jaguar's employees and former employees:

- i. claims equal to the amounts that such employees and former employees would have been qualified to receive under paragraph 136(I)(d) of the *Bankruptcy and Insolvency Act* (Canada) if Jaguar had become bankrupt on the Filing Date; and
- ii. claims for wages, salaries, commissions or compensation for services rendered by them after the Filing Date and on or before the date of the Sanction Order, together with, in the case of travelling salespersons, disbursements properly incurred by them in and about Jaguar's business during the same period.

"Equity Claim" has the meaning set forth in section 2(1) of the CCAA;

"Escrow Agent" means an independent third party escrow agent agreed to by Jaguar and the Majority Backstop Parties, in each case acting reasonably;

"Escrow Agreement" means the escrow agreement entered into by the Escrow Agent, Jaguar and the applicable Participating Eligible Investors and Funding Backstop Parties in connection with the Share Offering;

"Excluded Claim" means

- i. any claims secured by any of the Charges;
- ii. any Section 5.1(2) Director/Officer Claims;
- iii. any claims that cannot be compromised pursuant to subsection 19(2) of the CCAA, provided that no claims that have been or may be asserted by any Agreed Excluded Litigation Claimant shall constitute claims that cannot be compromised pursuant to subsection 19(2) of the CCAA for purposes of this Plan;
- iv. any claims of the Subsidiaries against the Applicant;
- v. any Secured Claims;
- vi. any Employee Priority Claims against the Applicant;
- vii. any Crown Claims against the Applicant;
- viii. the Trustees' claims under Section 6.07 of the 4.5% Convertible Note Indenture and the 5.5% Convertible Note Indenture, if any;
 - ix. any Post-Filing Claims;

- x. any claims of Persons who, at the Filing Date, are senior officers or employees of the Applicant, in respect of their employment arrangements or any termination of such arrangements;
- xi. the Renvest Claim;
- xii. the Agreed Excluded Director/Officer Litigation Claims;
- xiii. the Agreed Excluded Jaguar Litigation Claims; and
- xiv. the CRA Claim.

"Excluded Creditor" means a Person who has an Excluded Claim, but only in respect of and to the extent of such Excluded Claim;

"Existing Equity Holders" means, collectively, the Existing Shareholders and, as context requires, the Registered Holders or beneficial holders of Existing Share Options and the Registered Holders or beneficial holders of such;

"Existing Shareholders" means, as context requires, Registered Holders or beneficial holders of the Existing Shares, in their capacities as such;

"Existing Share Options" means all rights, options, warrants and other securities (other than the Notes) convertible or exchangeable into equity securities of Jaguar;

"Existing Shares" means all common shares of Jaguar that are issued and outstanding at the applicable time prior to the Implementation Time;

"Filing Date" means December 23, 2013;

"Funding Backstop Party" means a Backstop Party (i) in respect of whom the Backstop Agreement has not been terminated and (ii) unless such Backstop Party's Backstop Commitment has been reduced to zero in accordance with the Backstop Agreement and this Plan, who has deposited in escrow with the Escrow Agent either (a) its Backstop Payment Amount in full in cash; or (b) a qualified letter of credit in the full amount of its Backstop Payment Amount, in each case by the Backstop Funding Deadline and in accordance with the Backstop Agreement and Section 4.1(g) of this Plan;

"Governmental Entity" means any government, regulatory authority, governmental department, agency, commission, bureau, official, minister, Crown corporation, court, board, tribunal or dispute settlement panel or other law, rule or regulation-making organization or entity: (a) having or purporting to have jurisdiction on behalf of any nation, province, territory or state or any other geographic or political subdivision of any of them; or (b) exercising, or entitled or purporting to exercise any administrative, executive, judicial, legislative, policy, regulatory or taxing authority or power;

"Implementation Date" means the Business Day on which this Plan becomes effective, which shall be the Business Day on which the Monitor has filed with the Court the certificate contemplated in Section 12.6 hereof, or such other date as the Applicant, the Monitor and the Majority Consenting Noteholders may agree;

"Implementation Time" means 12:01 a.m. on the Implementation Date (or such other time as the Applicant, the Monitor and the Majority Consenting Noteholders may agree);

"Indentures" means the 4.5% Convertible Note Indenture and the 5.5% Convertible Note Indenture;

"Initial Order" has the meaning given to that term in Recital B;

"ITA" means the Income Tax Act, R.S.C. 1985, c.1 (5th Supp.);

"Itaú BBA" means Banco Itaú BBA S.A.;

"Jaguar Group" means, collectively, Jaguar, MSOL, MCT, MTL.;

"Jaguar Insurance Policy" means any insurance policy pursuant to which Jaguar is insured and any Director or Officer is insured, in his or her capacity as a Director or Officer;

"Law" means any law, statute, order, decree, consent decree, judgment, rule regulation, ordinance or other pronouncement having the effect of law whether in Canada, the United States, Brazil or any other country, or any domestic or foreign state, county, province, city or other political subdivision or of any Governmental Entity;

"Letter of Transmittal" means a letter of transmittal to be used by Registered Holders of Existing Shares to obtain replacement share certificates reflecting the Common Share Consolidation;

"**Majority Backstop Parties**" means the Backstop Parties (other than Defaulting Backstop Parties) having at least 66 ^{2/3} % of the aggregate Backstop Commitment of the Backstop Parties (other than Defaulting Backstop Parties) at the time that a consent, approval, waiver or agreement is sought pursuant to the terms of this Plan;

"Majority Consenting Noteholders" means Consenting Noteholders holding at least a majority of the aggregate principal amount of all Notes held by all Consenting Noteholders at the time that a consent, approval, waiver or agreement is sought pursuant to the terms of this Plan;

"MCT" has the meaning given to that term in Recital C;

"MSOL" has the meaning given to that term in Recital C;

"MTL" has the meaning given to that term in Recital C;

"Meeting" means a meeting of the Affected Unsecured Creditors called for the purpose of considering and voting in respect of this Plan;

"Monitor" means FTI Consulting Canada Inc., in its capacity as Court-appointed Monitor of Jaguar in the CCAA Proceedings;

"Named Directors and Officers" means the current directors and officers of Jaguar and such other directors and officers as agreed to by the Majority Consenting Noteholders prior to the Meeting;

"New Board" means the board of directors in place from and after the Implementation Date, the composition and size of which shall be satisfactory to the Majority Backstop Parties, subject to applicable Law;

"New Common Shares" means the 110,111,111 Common Shares to be issued by Jaguar on the Implementation Date in accordance with the steps set out in Section 7.4;

"Non-Breaching/Non-Delivering Backstop Parties" means those Backstop Parties that are neither Breaching Backstop Parties nor Non-Delivering Backstop Parties;

"Non-Defaulting Backstop Parties" means those Backstop Parties that are not Defaulting Backstop Parties;

"**Non-Delivering Backstop Party**" means a Backstop Party (who is not otherwise an Objecting Backstop Party) that has not delivered an executed Rep Letter to Jaguar by the Election Deadline or for whom a representation or warranty made in such Rep Letter becomes untrue;

"Non-Objecting Backstop Parties" means those Backstop Parties that are not Objecting Backstop Parties;

"Non-Released Director/Officer Claims" means Director/Officer Claims against the Directors and Officers of Jaguar in respect of which such Director or Officer has been adjudged by the express terms of a judgment rendered on a final determination on the merits to have committed fraud or wilful misconduct, but excluding any claims that have been or may be asserted by any Agreed Excluded Litigation Claimants;

"Noteholder Released Claim" means the matters that are subject to release and discharge pursuant to Section 11.1(c);

"Noteholder Released Party" has the meaning given to that term in Section 11.1(c);

"Noteholder Voting Record Date" means December 19, 2013;

"Noteholders" means, as the context requires, the Registered Holders or beneficial holders of the Notes, in their capacities as such;

"Noteholders Allowed Claim" means all principal amounts outstanding and all accrued interest under the Notes as at the applicable record date under this Plan as determined in accordance with the Claims Procedure Order for purposes of voting on, and receiving distributions under, this Plan;

"Noteholder's Allowed Claim" means, in respect of a particular Noteholder, all principal amounts outstanding and accrued interest under the Notes owing to such Noteholder as at the applicable record date under this Plan as determined in accordance with the Claims Procedure Order for purposes of voting on, and receiving distributions under, this Plan;

"Notes" means, collectively, the notes issued by Jaguar under and pursuant to the Indentures;

"**Objecting Backstop Party**" means a Backstop Party that has terminated its obligations under the Backstop Agreement in accordance with section 8(c) thereof;

"Offering Shares" means the 70,955,797 New Common Shares to be issued by Jaguar pursuant to the Share Offering;

"Offered Shares" means, collectively, the Offering Shares (including the Backstopped Shares), the Accrued Interest Offering Shares, and the Backstop Commitment Shares;

"Officer" means anyone who is or was, or may be deemed to be or have been, whether by statute, operation of law or otherwise, an officer or *de facto* officer of Jaguar;

"Order" means any order of the Court in the CCAA Proceedings;

"Other Directors and/or Officers" means any Directors and/or Officers other than the Named Directors and Officers;

"Outside Date" means February 28, 2014 (or such other date as the Applicant and the Majority Consenting Noteholders may agree);

"Participant Holder" has the meaning ascribed thereto in the Meeting Order;

"Participating Eligible Investor" has the meaning given to that term in Section 4.1(h);

"Participating Eligible Investor Shares" has the meaning given to that term in Section 4.1(h);

"**Party**" means a party to the Support Agreement and/or to the Backstop Agreement, and any reference to a Party includes its successors and permitted assigns; and "**Parties**" means every Party;

"**Person**" is to be broadly interpreted and includes any individual, firm, corporation, limited or unlimited liability company, general or limited partnership, association, trust, unincorporated organization, joint venture, Government Entity or any agency, officer or instrumentality thereof or any other entity, wherever situate or domiciled, and whether or not having legal status;

"**Plan**" means this Amended and Restated Plan of Compromise and Arrangement and any amendments, modifications or supplements hereto made in accordance with the terms hereof or made at the direction of the Court in the Sanction Order or otherwise with the consent of Jaguar and the Majority Consenting Noteholders, each acting reasonably;

"Plan Resolution" means the resolution of the Affected Unsecured Creditors relating to this Plan considered at the Meeting;

"**Post-Filing Claim**" means any claims against the Applicant that arose from the provision of authorized goods and services provided or otherwise incurred on or after the Filing Date in the ordinary course of business, but specifically excluding any Restructuring Period Claim;

"Pre-filing Claim" has the meaning given to that term in the definition of Claim;

"Pro Rata Share" means:

- (a) in respect of Unsecured Creditor Common Shares, the percentage that an Affected Unsecured Creditor's Allowed Affected Unsecured Claim calculated as at the Record Date bears to the aggregate of all Allowed Affected Unsecured Claims calculated as at the Record Date and all Disputed Distribution Claims calculated as at the Record Date;
- (b) in respect of the Early Consent Shares, the percentage that an Early Consenting Noteholder's Noteholder's Allowed Claim calculated as at the Record Date bears to the aggregate of all Early Consenting Noteholders' Noteholder's Allowed Claims calculated as at the Record Date;
- in respect of the Subscription Privilege, the percentage that an Eligible Investor's Noteholder's Allowed Claim calculated as at the Record Date bears to the Noteholders Allowed Claim calculated as at the Record Date, subject to adjustment pursuant to Section 5.2(c) hereof;
- (d) in respect of the Accrued Interest Offering Shares, the percentage that a Participating Eligible Investor's Accrued Interest Claim or a Funding Backstop Party's Accrued Interest Claim (without duplication), as applicable, bears to the aggregate of all Accrued Interest Claims;
- (e) in respect of the Backstop Commitment Shares, the percentage that a Funding Backstop Party's Backstop Consideration Commitment bears to the aggregate of all Funding Backstop Parties' Backstop Consideration Commitments; and
- (f) in respect of the Backstopped Shares, the percentage that a Backstop Party's Backstop Commitment bears to the aggregate of all Backstop Commitments.

"Record Date" means December 31, 2013;

"**Registered Holder**" means (i) in respect of the Notes, the holder of such Notes as recorded on the books and records of the Trustees, (ii) in respect of the Existing Shares, the holder of such Existing Shares as recorded on the share register maintained by the Transfer Agent, and (iii) in respect of the Existing Share Options, the holder of such Existing Share Options as recorded on the books and records of Jaguar;

"**Regulation S**" means Regulation S as promulgated by the US Securities Commission under the US Securities Act;

"**Released Claims**" means the matters that are subject to release and discharge pursuant to Section 11.1(a) and (b) hereof;

"Released Party" has the meaning given to that term in Section 11.1(b);

"**Renvest Claim**" means any claim for amounts owing by the Applicant to Global Resource Fund, pursuant to the Credit Agreement or pursuant to any Credit Document (as such term is defined in the Credit Agreement).

"**Rep Letter**" means a letter from a Noteholder, or an Assuming Backstop Party who is not a Noteholder, or an Affected Unsecured Creditor with an Allowed Affected Unsecured Claim who is not a Noteholder, if applicable in accordance with Section 5.2(c) hereof, to Jaguar containing representations and warranties relating to such Person's eligibility to acquire the Offering Shares (including the Backstopped Shares), Accrued Interest Offering Shares, or Backstop Commitment Shares under US Securities Laws, in a form acceptable to such Person and Jaguar, each acting reasonably;

"Required Majority" means a majority in number of Affected Unsecured Creditors representing at least two thirds in value of the Voting Claims of Affected Unsecured Creditors who are entitled to vote at the Meeting in accordance with the Meeting Order and who are present and voting in person or by proxy on the Plan Resolution at the Meeting;

"Restructuring Period Claim" has the meaning given to that term in the definition of Claim;

"Rights" means the rights issued pursuant to the Shareholder Rights Plan;

"**RSU Plan**" means the restricted share unit plan for senior officers, employees and consultants adopted in November of 2008 by Jaguar, as amended from time to time;

"SAR Plan" means the Third<u>Fourth</u> Amended and Restated Share Appreciation Rights Plan of Jaguar, effective as of December 8<u>October 4</u>, 20103;

"**Sanction Order**" means the Order of the Court sanctioning and approving this Plan pursuant to section 6(1) of the CCAA, which shall include such terms as may be necessary or appropriate to (i) give effect to this Plan, in form and substance satisfactory to the Applicant and the Majority Consenting Noteholders, each acting reasonably, and (ii) allow Jaguar to rely on the exemption from registration set forth in section 3(a)(10) of the US Securities Act;

"Section 5.1(2) Director/Officer Claim" means any claim against any Director and/or Officer that is not permitted to be compromised pursuant to section 5.1(2) of the CCAA, but only to the extent not so permitted, provided that any Director/Officer Claim that qualifies as a Non-Released Director/Officer Claim shall not constitute a Section 5.1(2) Director/Officer Claim for the purposes of Section 11.1(a) hereof; and provided further that no claims that have been or may be asserted by any Agreed Excluded Litigation Claimant shall constitute Section 5.1(2) Director/Officer Claims for the purposes of this Plan;

"Secured Claims" means that portion of a Claim that is (i) secured by security validly charging or encumbering property or assets of the Applicant (including statutory and possessory liens that create security interests) but only up to the value of such collateral, and (ii) duly and properly perfected in accordance with the relevant legislation in the appropriate jurisdiction as of the Filing Date;

"Share Offering" means the offering by Jaguar of Offering Shares at the Subscription Price in accordance with this Plan;

"Shareholder Rights Plan" means the Shareholder Rights Plan Agreement dated May 2, 2013 between Jaguar Mining Inc. and Computershare Investor Services Inc. as Rights Agent;

"Solicitation/Election Agent" means Globic Advisors Inc., or any successor solicitation or election agent;

"Stock Option Plan" means the stock option plan of Jaguar in effect as of the Filing Date;

"Subscription Price" means \$0.7047 per Offering Share;

"Subscription Privilege" means the right of an Eligible Investor to participate in the Share Offering by electing, in accordance with the provisions of this Plan, to subscribe for and purchase from Jaguar up to its Pro Rata Share of Offering Shares under the Share Offering;

"Subscription Record Date" means December 19, 2013;

"Subsidiaries" means, collectively, MTL, MSOL and MCT, and "Subsidiary" means any one of the Subsidiaries;

"Support Agreement" means the Support Agreement made November 13, 2013 (as amended from time to time) between Jaguar, the Subsidiaries and the Noteholders party thereto, together with any consent agreements executed by other Noteholders from time to time, substantially in the form of Schedule C thereto;

"Tax" or "Taxes" means any and all taxes, duties, fees, premiums, assessments, imposts, levies and other charges of any kind whatsoever, including all interest, penalties, fines, additions to tax or other additional amounts in respect thereof, and including those levied on, or measured by, or referred to as, income, gross receipts, profits, capital, transfer, land transfer, sales, goods and services, harmonized sales, use, value-added, excise, stamp, withholding, business, franchising, property, development, occupancy, employer health, payroll, employment, health, social services, education and social security taxes, all surtaxes, all customs duties and import and export taxes, countervail and anti-dumping, all licence, franchise and registration fees and all employment insurance, health insurance and Canada, Quebec and other government pension plan premiums or contributions;

"Tax Claim" means any Claim against the Applicant for any Taxes in respect of any taxation year or period;

"Transfer Agent" means Computershare Investor Services Inc.;

"Trustees" means The Bank of New York Mellon, as trustee, and BNY Trust Company of Canada, as co-trustee, under each of the Indentures;

"TSX" means Toronto Stock Exchange;

"TSXV" means TSX Venture Exchange;

"Undeliverable Distribution" has the meaning given to that term in Section 8.3;

"Unsecured Creditor Common Shares" means 14,000,000 New Common Shares;

"US Dollars" or "US\$" means the lawful currency of the United States of America;

***US Securities Act**" means the *United States Securities Act of 1933*, as amended from time to time, and the rules and regulations promulgated thereunder, or any successor statute;

"US Securities Commission" means the United States Securities and Exchange Commission;

"US Securities Laws" means, collectively, the *Sarbanes-Oxley Act of 2002* ("Sarbanes-Oxley"), the US Securities Act, as amended, the *United States Securities Exchange Act of 1934*, as amended, the rules and regulations of the US Securities Commission, the auditing principles, rules, standards and practices applicable to auditors of "issuers" (as defined in Sarbanes-Oxley) promulgated or approved by the Public Company Accounting Oversight Board and, as applicable, the rules of the New York Stock Exchange;

"Voting Claim" means the amount of the Affected Unsecured Claim of an Affected Unsecured Creditor against the Applicant as finally accepted and determined for purposes of voting at the Meeting, in accordance with the provisions of the Claims Procedure Order and the CCAA; and

"Voting Deadline" means 10 a.m. on the Business Day prior to the Meeting.

1.2 Certain Rules of Interpretation

For the purposes of this Plan:

- (a) Unless otherwise expressly provided herein, any reference in this Plan to an instrument, agreement or an Order or an existing document or exhibit filed or to be filed means such instrument, agreement, Order, document or exhibit as it may have been or may be amended, modified, or supplemented in accordance with its terms;
- (b) The division of this Plan into articles and sections are for convenience of reference only and do not affect the construction or interpretation of this Plan, nor are the descriptive headings of articles and sections intended as complete or accurate descriptions of the content thereof;
- (c) The use of words in the singular or plural, or with a particular gender, including a definition, shall not limit the scope or exclude the application of any provision of this Plan to such Person (or Persons) or circumstances as the context otherwise permits;
- (d) The words "includes" and "including" and similar terms of inclusion shall not, unless expressly modified by the words "only" or "solely", be construed as terms of limitation, but rather shall mean "includes but is not limited to" and "including but not limited to", so that references to included matters shall be regarded as illustrative without being either characterizing or exhaustive;
- (e) Unless otherwise specified, all references to time herein and in any document issued pursuant hereto mean local time in Toronto, Ontario and any reference to an event occurring on a Business Day shall mean prior to 5:00 p.m. on such Business Day;
- (f) Unless otherwise specified, time periods within or following which any payment is to be made or act is to be done shall be calculated by excluding the day on which the period commences and including the day on which the period ends;
- (g) Unless otherwise provided, any reference to a statute or other enactment of parliament, a legislature or other Governmental Entity includes all regulations made thereunder, all

amendments to or re-enactments of such statute or regulations in force from time to time, and, if applicable, any statute or regulation that supplements or supersedes such statute or regulation;

- (h) References to a specific Recital, Article or Section shall, unless something in the subject matter or context is inconsistent therewith, be construed as references to that specific Recital, Article or Section of this Plan, whereas the terms "this Plan", "hereof", "herein", "hereto", "hereunder" and similar expressions shall be deemed to refer generally to this Plan and not to any particular Recital, Article, Section or other portion of this Plan and include any documents supplemental hereto; and
- (i) The word "or" is not exclusive.

1.3 Governing Law

This Plan shall be governed by and construed in accordance with the laws of Ontario and the federal laws of Canada applicable therein. All questions as to the interpretation or application of this Plan and all proceedings taken in connection with this Plan and its provisions shall be subject to the jurisdiction of the Court.

1.4 Currency

Unless otherwise stated, all references in this Plan to sums of money are expressed in, and all payments provided for herein shall be made in, US Dollars.

1.5 Date for Any Action

If the date on which any action is required to be taken hereunder by a Person is not a Business Day, such action shall be required to be taken on the next succeeding day which is a Business Day.

1.6 Time

Time shall be of the essence in this Plan.

ARTICLE 2- PURPOSE AND EFFECT OF THIS PLAN

2.1 Purpose

The purpose of this Plan is to facilitate the continuation of the business of the Jaguar Group as a going concern, address certain liabilities of the Applicant, and effect a recapitalization and financing transaction on an expedited basis to provide a stronger financial foundation for the Jaguar Group going forward and additional liquidity to allow the Jaguar Group to continue to work towards its operational and financial goals from and after the Implementation Date in the expectation that all Persons with an economic interest in the Jaguar Group will derive a greater benefit from the implementation of this Plan than would otherwise result.

2.2 Effectiveness

Subject to the satisfaction, completion or waiver (to the extent permitted pursuant to Section 12.4) of the conditions precedent set out herein, this Plan will become effective in the sequence described in Section 7.4 from and after the Implementation Time and shall be binding on and enure to the benefit of the Jaguar Group, the Affected Unsecured Creditors, all Existing Equity Holders, all holders of Equity Claims, the Released Parties, the Noteholder Released Parties and all other Persons as provided for herein, or subject to, this Plan and their respective successors and assigns and their respective heirs, executors, administrators and other legal representatives, successors and assigns.

2.3 Persons Not Affected

For greater certainty, except as provided in Sections 11.1(a)(iii), 11.1(b)(i), 11.2, 12.2(c) and 13.1, this Plan does not affect the holders of Excluded Claims to the extent of those Excluded Claims. Nothing in this Plan shall affect the Jaguar Group's rights and defences, both legal and equitable, with respect to any Excluded Claims, including, but not limited to, all rights with respect to legal and equitable defences or entitlements to set-offs or recoupments against such Excluded Claims. Nothing herein shall constitute a waiver of any right of either the Monitor or the Applicant to dispute the quantum of an Excluded Claim.

ARTICLE 3– CLASSIFICATION, VOTING CLAIMS AND RELATED MATTERS

3.1 Classes

For the purposes of considering and voting on the Plan Resolution, there shall be one class of stakeholders, consisting of Affected Unsecured Creditors (the "Affected Creditor Class").

3.2 Meeting

- (a) The Meeting shall be held in accordance with this Plan, the Meeting Order and any further Order in the CCAA Proceedings. Subject to the terms of any further Order in the CCAA Proceedings, the only Persons entitled to notice of, to attend or to speak at the Meeting are the Eligible Voting Creditors (or their respective duly appointed proxyholders), representatives of the Monitor, the Applicant, the Consenting Noteholders, all such parties' financial and legal advisors, the Chair (as defined in the Meeting Order), the Secretary (as defined in the Meeting Order) and the Scrutineers (as defined in the Meeting Order). Any other person may be admitted to the Meeting only by invitation of the Applicant or the Chair.
- (b) For the purposes of voting at the Meeting, each Affected Unsecured Creditor (including a Beneficial Noteholder with respect to its Noteholder's Allowed Claim) shall be entitled to one vote as a member of the Affected Creditor Class.
- (c) For the purposes of voting at the Meeting, the Voting Claim of any Beneficial Noteholder shall be deemed to be equal to its Noteholder's Allowed Claim as at the Noteholder Voting Record Date. Registered Holders of Notes, in their capacities as such, will not be entitled to vote at the Meeting.

3.3 Required Majority

In order to be approved, this Plan must receive the affirmative vote of the Required Majority of the Affected Creditor Class.

3.4 Excluded Claims

Excluded Creditors shall not be entitled to vote or (except as otherwise expressly stated in the Meeting Order) attend in respect of their Excluded Claims at any meeting to consider and approve this Plan.

3.5 Existing Equity Holders and Holders of Equity Claims

Existing Equity Holders and holders of Equity Claims shall not be entitled to attend or vote in respect of their Equity Claims at any meeting to consider and approve this Plan.

3.6 Crown Claims

All Crown Claims in respect of all amounts that were outstanding at the Filing Date shall be paid in full to the Crown within six months of the Sanction Order, as required by subsection 6(3) of the CCAA.

3.7 Payments to Employees

Immediately after the date of the Sanction Order, the Applicant will pay in full all Employee Priority Claims, if any, to its employees and former employees.

ARTICLE 4 – ELECTIONS AND SHARE OFFERING

4.1 Participation In Share Offering

- (a) Each Noteholder that is an Eligible Investor shall be entitled to participate in the Share Offering.
- (b) Pursuant to and in accordance with the Meeting Order, there shall be delivered an election form (an "Election Form") to each Participant Holder of the Notes, as of the Subscription Record Date, together with instructions to deliver such Election Form (or copies thereof) to the applicable Beneficial Noteholders to the extent such Participant Holder is not also the Beneficial Noteholder of such Notes. Each Eligible Investor shall have the right, but not the obligation, to irrevocably elect to exercise its Subscription Privilege, with such subscription to be conditioned upon the implementation of this Plan and effective on the Implementation Date in accordance with Section 7.4. In order to exercise its Subscription Privilege, such Eligible Investor shall return, or cause to be returned, the duly executed Election Form (including a Rep Letter) in accordance with the Meeting Order, so that it is received by the Solicitation/Election Agent on or before the Election Deadline.
- (c) An Electing Eligible Investor that is also a Backstop Party may elect, in accordance with the Election Form, to have its Backstop Commitment reduced by the total funds that such Electing Eligible Investor deposits into escrow on or before the Electing Eligible Investor Funding Deadline in respect of Offering Shares that such Electing Eligible Investor subscribes for pursuant to the exercise of all or part of its Subscription Privilege, provided that such Backstop Commitment shall not be reduced below zero (the "Backstop Commitment Reduction Election", with a Backstop Party so electing being a "Commitment Reduction Electing Backstopper").
- (d) Following the issuance of the Sanction Order, but in any event by 5:00 p.m. on the tenth Business Day prior to the expected Implementation Date, Jaguar shall inform each Electing Eligible Investor of (i) the expected Implementation Date, (ii) the number of Offering Shares that, subject to compliance with the procedures described in this Plan, will be acquired by such Electing Eligible Investor on the Implementation Date pursuant to the Subscription Privilege; and (iii) the amount of funds (in cash) required to be deposited in escrow with the Escrow Agent by such Electing Eligible Investor to purchase such Offering Shares pursuant to the Share Offering (the "Electing Eligible Investor Funding Amount") by the Electing Eligible Investor Funding Deadline.
- (e) Each Electing Eligible Investor must deposit its Electing Eligible Investor Funding Amount in escrow with the Escrow Agent so that it is received by the Escrow Agent by no later than 11:00 a.m. on the seventh Business Day prior to the expected Implementation Date (the "Electing Eligible Investor Funding Deadline"). If an Electing Eligible Investor deposits less than the full amount of its Electing Eligible Investor Funding Amount by the Electing Eligible Investor Funding Deadline, then (i) the funds so deposited by such

Electing Eligible Investor shall be returned to such Electing Eligible Investor within five Business Days following the Electing Eligible Investor Funding Deadline; and (ii) such Eligible Investor shall be deemed to have ceased, as of the Electing Eligible Investor Funding Deadline, to be an Electing Eligible Investor and its subscription for Offering Shares pursuant to the Subscription Privilege and right to receive Accrued Interest Offering Shares shall be null and void.

- (f) As soon as practicable but in any event no later than 11:00 a.m. one Business Day after the Electing Eligible Investor Funding Deadline, Jaguar shall inform each Backstop Party (other than a Backstop Party in respect of whom the Backstop Agreement has been terminated) of (i) the total number of Offering Shares not validly subscribed for pursuant to the Subscription Privilege (the "Backstopped Shares"), (ii) the number of Backstopped Shares to be acquired by such Backstop Party pursuant to its Backstop Commitment, based upon its Pro Rata Share of the Backstopped Shares, and (iii) the amount of funds (by way of cash or a letter of credit) required to be deposited in escrow with the Escrow Agent by such party to purchase such Backstopped Shares (the "Backstop Payment Amount") by the Backstop Funding Deadline.
- (g) Each Backstop Party (other than a Backstop Party in respect of whom the Backstop Agreement has been terminated) shall deliver to the Escrow Agent and the Escrow Agent shall have received, not later than 2:00 p.m. (Toronto time) on the day that is five Business Days prior to the expected Implementation Date (the "**Backstop Funding Deadline**"), either:
 - (i) cash in an amount equal to the full amount of such Backstop Party's Backstop Payment Amount; or
 - (ii) a letter of credit, in form and substance reasonably satisfactory to Jaguar, having a face amount equal to such Backstop Party's Backstop Payment Amount, and issued by a financial institution having an equity market capitalization of at least \$10,000,000,000 and a credit rating of at least A+ from Standard & Poor's or A1 from Moody's,
- in each case: (1) to be held in escrow in accordance with the Escrow Agreement until all conditions to the Share Offering have been satisfied or waived in accordance with the Backstop Agreement and with irrevocable instructions to use such cash or letter of credit, as applicable, to the extent required to enable such Backstop Party to comply with its Backstop Purchase Obligation; and (2) provided for greater certainty that, if a Backstop Party (A) has exercised all or part of its Subscription Privilege and has paid its Electing Eligible Investor Funding Amount on or before the Electing Eligible Investor Funding Deadline, and (B) is a Commitment Reduction Electing Backstop Party shall not be required to deliver cash or a letter of credit to the Escrow Agent.
- (h) An Electing Eligible Investor who complies with Section 4.1(e) (the "Participating Eligible Investor") shall participate in the Share Offering and shall be deemed to have subscribed for Offering Shares in an amount equal to the Electing Eligible Investor Funding Amount deposited in escrow with the Escrow Agent by that Participating Eligible Investor in accordance with Section 4.1(e) divided by the Subscription Price (the "Participating Eligible Investor Shares").
- (i) Each Funding Backstop Party shall be deemed to have subscribed for its Pro Rata Share of the Backstopped Shares.

- (j) On or prior to the Implementation Date, Jaguar shall inform: (i) each Participating Eligible Investor of the number of Accrued Interest Offering Shares to be allocated to such Participating Eligible Investor in accordance with section 5.1(b); and (ii) each Funding Backstop Party of the number of Accrued Interest Offering Shares and the number of Backstop Commitment Shares to be allocated to such Funding Backstop Party in accordance with section 5.1(b).
- In the event of a Backstop Default/Termination, provided that the Backstop Agreement (k) remains in full force and effect with respect to other Backstop Parties thereafter, Jaguar shall, in accordance with the Backstop Agreement, provide the applicable Backstop Parties, or such other parties acceptable to the Backstop Parties and Jaguar in accordance with the Backstop Agreement that will execute a Backstop Consent Agreement, with an opportunity to assume the obligations (and rights) of a Defaulting Backstop Party, Objecting Backstop Party, Breaching Backstop Party or Non-Delivering Backstop Party, as applicable, in each case in accordance with and subject to the terms and conditions of this Plan and the Backstop Agreement. Any Assuming Backstop Party shall comply with its obligations in connection with its Assumed Backstop Commitment and shall be entitled to receive the applicable Offered Shares under this Plan in connection with such Assumed Backstop Commitment, subject to such Assuming Backstop Party having complied with its obligations under this Plan and the Backstop Agreement and such other terms and conditions under this Plan and the Backstop Agreement, For greater certainty, any Assuming Backstop Party that has complied with its obligations under this Plan and the Backstop Agreement shall constitute and be treated as a Funding Backstop Party for purposes of this Plan.

ARTICLE 5 – TREATMENT OF CLAIMS

5.1 Treatment of Noteholders

- (a) For the purposes of distributions under this Plan, the Distribution Claim of any Beneficial Noteholder shall be deemed to be equal to its Noteholder's Allowed Claim.
- (b) On the Implementation Date and in accordance with the steps and sequence as set forth in this Plan, each Noteholder shall and shall be deemed to irrevocably and finally exchange its Notes for the following consideration which shall and shall be deemed to be received in full and final settlement of its Notes and its Noteholder's Allowed Claim:
 - (i) its Pro Rata Share of the Unsecured Creditor Common Shares;
 - (ii) its Pro Rata Share of the Early Consent Shares, if such Noteholder is an Early Consenting Noteholder;
 - (iii) its Pro Rata Share of Accrued Interest Offering Shares if such Noteholder is a Participating Eligible Investor and/or a Funding Backstop Party, provided that in no event shall a Participating Eligible Investor or a Funding Backstop Party receive a greater number of Accrued Interest Offering Shares than Offering Shares (including Backstopped Shares, as applicable) received by such person. Any Accrued Interest Offering Shares remaining after the allocation of the Accrued Interest Offering Shares to Participating Eligible Investors and Funding Backstop Parties pursuant to the immediately preceding sentence shall be reallocated among those Participating Eligible Investors and/or Funding Backstop Parties who have received less Accrued Interest Offering Shares than Offering Shares (including Backstopped Shares, as applicable) on a *pro rata* basis based on Accrued Interest Claims of such Participating Eligible Investors and/or Funding Backstop Parties (calculated as at the Record Date); and

- (iv) its Pro Rata Share of the Backstop Commitment Shares, if such Noteholder is a Funding Backstop Party.
- (c) On the Implementation Date and in accordance with the steps and sequence as set forth in this Plan, each Participating Eligible Investor shall receive its Participating Eligible Investor Shares and each Funding Backstop Party shall receive its Pro Rata Share of the Backstopped Shares.
- (d) After giving effect to the terms of this Section 5.1, the obligations of Jaguar with respect to the Notes of each Noteholder shall, and shall be deemed to, have been irrevocably and finally extinguished and each Noteholder shall have no further right, title or interest in or to the Notes or its Noteholder's Allowed Claim.

5.2 Treatment of Affected Unsecured Creditors Other Than Noteholders

- (a) On the Implementation Date and in accordance with the steps and sequence as set forth in this Plan, each Affected Unsecured Creditor (except for a Noteholder in respect of its Noteholder's Allowed Claim, which shall be dealt with in accordance with Section 5.1) shall receive its Pro Rata Share of the Unsecured Creditor Common Shares and shall be deemed to irrevocably and finally exchange its Affected Unsecured Claim for its Pro Rata Share of the Unsecured Creditor Common Shares, which shall and shall be deemed to be received in full and final settlement of its Affected Unsecured Claim.
- (b) After giving effect to the terms of this Section 5.2, the obligations of Jaguar with respect to such Affected Unsecured Creditor's Affected Unsecured Claim shall, and shall be deemed to, have been irrevocably and finally extinguished and such Affected Unsecured Creditor shall have no further right, title or interest in or to the Affected Unsecured Claim.
- (c) With the consent of the Monitor and the Majority Backstop Parties, an Affected Unsecured Creditor with an Allowed Affected Unsecured Claim who is not a Noteholder may be entitled to participate in the Share Offering for its Pro Rata Share of the Offering Shares (calculated as if the Affected Unsecured Creditor's Allowed Affected Unsecured Claim was a Noteholder's Allowed Claim); provided that any such Affected Unsecured Creditor completes and submits an Election Form and Rep Letter on or prior to the Election Deadline and complies with all of the obligations of a Participating Eligible Investor in accordance with the terms and conditions of the Plan, including without limitation Section 4.1(e) hereof, in which case, such Affected Unsecured Creditor shall be treated as an Eligible Investor for the purpose of the Offering Shares and each Eligible Investor's Subscription Privilege will be adjusted accordingly.

5.3 Treatment of Existing Equity Holders

- (a) Each Existing Shareholder shall retain its Existing Shares subject to the Common Share Consolidation pursuant to Section 7.4(a) and in accordance with the steps and sequences set forth herein.
- (b) Pursuant to this Plan and in accordance with the steps and sequences set forth herein, all Existing Share Options, Rights and the Shareholder Rights Plan shall be cancelled and shall be deemed to be cancelled without the need for any repayment of capital thereof or any other liability, payment or compensation therefor and for greater certainty, no holders of Existing Share Options or Rights shall be entitled to receive any interest, dividends, premium or other payment in connection therewith.
5.4 Equity Claims

All Equity Claims shall be fully, finally and irrevocably and forever compromised, released, discharged, cancelled and barred on the Implementation Date. Holders of Equity Claims shall not receive any consideration or distributions under this Plan and shall not be entitled to vote on this Plan at the Meeting. Notwithstanding the foregoing, Existing Shareholders shall be entitled to continue to hold their Existing Shares in accordance with the terms of this Plan, subject to the Common Share Consolidation.

5.5 Claims of the Trustees

The Trustees' claims under Section 6.07 of the 4.5% Convertible Note Indenture and the 5.5% Convertible Note Indenture shall be unaffected by this Plan.

5.6 Application of Plan Distributions

- (a) All amounts paid or payable hereunder on account of the Noteholders Allowed Claim (including, for greater certainty, any securities received hereunder) shall be applied as follows: (i) first, in respect of the principal amount of the obligations to which such Noteholders Allowed Claim relate, and (ii) second, if such principal amounts have been fully repaid, in respect of any accrued but unpaid interest on such obligations.
- (b) In the event that a Funding Backstop Party is not a Noteholder, such Funding Backstop Party shall receive its Backstop Commitment Shares as a fee.

ARTICLE 6 – MEETING

6.1 Meeting

The Meeting to consider and vote on this Plan shall be conducted in accordance with the terms of the Claims Procedure Order and the Meeting Order.

6.2 Acceptance of Plan

If this Plan is approved by the Required Majority entitled to vote at the Meeting, then this Plan shall be deemed to have been agreed to, accepted and approved by the Affected Unsecured Creditors and shall be binding upon all Affected Unsecured Creditors, if the Sanction Order is granted and the conditions described in Section 12.3 hereof have been satisfied or waived, as applicable.

ARTICLE 7 – IMPLEMENTATION

7.1 Administration Charge

On the Implementation Date, all outstanding, invoiced obligations, liabilities, fees and disbursements secured by the Administration Charge shall be fully paid by the Applicant. Upon receipt by the Monitor of confirmation from each of the beneficiaries of the Administration Charge that payments of the amounts secured by the Administration Charge have been made, the Monitor shall file a certificate with the Court confirming same and thereafter, the Administration Charge shall be and be deemed to be discharged from the assets of the Applicant, without the need for any other formality.

7.2 Corporate Authorizations

The adoption, execution, delivery, implementation and consummation of all matters contemplated under this Plan involving corporate action of any members of the Jaguar Group will occur and be effective as of the Implementation Date (or such other date as Jaguar and the Majority Consenting Noteholders may agree), and will be authorized and approved under this Plan and by the Court, where appropriate, as part of the Sanction Order, in all respects and for all purposes without any requirement of further action by shareholders, directors or officers of the Jaguar Group. All necessary approvals to take actions shall be deemed to have been obtained from the directors or the shareholders of the Jaguar Group, as applicable, including the deemed passing by any class of shareholders of any resolution or special resolution and no shareholders' agreement or agreement between a shareholder and another Person limiting in any way the right to vote shares held by such shareholder or shareholders with respect to any of the steps contemplated by this Plan shall be deemed to be effective and no such agreement shall have any force or effect.

7.3 Fractional Interests

No certificates representing fractional Common Shares shall be allocated under this Plan, and fractional share interests shall not entitle the owner thereof to vote or to any rights of a shareholder of Jaguar. Any legal, equitable, contractual and any other rights or claims (whether actual or contingent, and whether or not previously asserted) of any Person with respect to fractional Common Shares pursuant to this Plan shall be rounded down to the nearest whole number without compensation therefor.

7.4 Implementation Date Transactions

Commencing at the Implementation Time, the following events or transactions will occur, or be deemed to have occurred and be taken and effected, in the following order in five minute increments and at the times set out in this Section 7.4 (or in such other manner or order or at such other time or times as Jaguar and the Majority Consenting Noteholders may agree, acting reasonably), without any further act or formality required on the part of any Person, except as may be expressly provided herein:

- (a) Articles of Reorganization shall be filed under the OBCA to amend the articles of Jaguar to effect a consolidation (the "**Common Share Consolidation**") of the issued and outstanding Common Shares on the basis of one post-consolidation Common Share for each Consolidation Number of Common Shares outstanding immediately prior to the Common Share Consolidation. Any fractional interests in the consolidated Common Shares will, without any further act or formality, be cancelled without payment of any consideration therefor. Following the completion of such consolidation, the stated capital of the Common Shares shall be equal to the stated capital of the Common Shares immediately prior to consolidation.
- (b) The following shall occur concurrently:
 - the Rights and the Shareholder Rights Plan shall be cancelled and shall be deemed to be cancelled without the need for any repayment of capital thereof or any other compensation therefor and shall cease to be of any further force or effect;
 - (ii) any and all Existing Share Options and the Stock Option Plan shall be cancelled and shall be deemed to be cancelled without the need for any repayment of capital thereof or any other compensation therefor and shall cease to be of any further force or effect;
 - (iii) if the DSU/RSU/SAR Notice is delivered, the DSU Plan, the RSU Plan and/or the SAR Plan, as set out in the DSU/RSU/SAR Notice shall be cancelled and shall be deemed to be cancelled without the need for any repayment of capital thereof or any other compensation therefor and shall cease to be of any further force or effect; and

- (iv) all Equity Claims shall be fully, finally, irrevocably and forever compromised, released, discharged, cancelled and barred without any consideration or distributions therefor.
- (c) In exchange for, and in full and final settlement of, the Noteholders Allowed Claim as at the Implementation Date, Jaguar shall issue:
 - (i) to each Noteholder its Pro Rata Share of Unsecured Creditor Common Shares;
 - to each Early Consenting Noteholder its Pro Rata Share of the Early Consent Shares;
 - (iii) to each Participating Eligible Investor and Funding Backstop Party the number of Accrued Interest Offering Shares such Participating Eligible Investor or Funding Backstop Party is entitled to receive in accordance with Section 5.1(b); and
 - (iv) to each Funding Backstop Party, its Pro Rata Share of the Backstop Commitment Shares,
- which New Common Shares shall be distributed in the manner described in Section 8.2 hereof. Upon issuance of these New Common Shares, the Noteholders Allowed Claim shall and shall be deemed to be irrevocably and finally extinguished and such Noteholder shall have no further right, title or interest in and to the Notes or its Noteholder's Allowed Claim.
- (d) The Notes and the Indentures will not entitle any Noteholder to any compensation or participation other than as expressly provided for in this Plan and shall be cancelled and will thereupon be null and void, and the obligations of the Applicant thereunder or in any way related thereto shall be satisfied and discharged, except to the extent expressly set forth in section 6.07 of the Indentures, which section shall remain in effect until two months following the Implementation Date or such later date agreed to by the Applicant, the Monitor, the Trustees and the Majority Consenting Noteholders.
- (e) In exchange for, and in full and final settlement of, its Affected Unsecured Claim, Jaguar shall issue to each Affected Unsecured Creditor, other than the Noteholders, its Pro Rata Share of the Unsecured Creditor Common Shares;
- (f) The following shall occur concurrently:
 - (i) Jaguar shall issue to each Participating Eligible Investor its Participating Eligible Investor Shares in accordance with Section 5.1(c) hereof in consideration for its Electing Eligible Investor Funding Amount, which Participating Eligible Investor Shares shall be distributed in the manner described in Section 8.2 hereof; and
 - (ii) Jaguar shall issue to each Funding Backstop Party the number of Backstopped Shares such Funding Backstop Party is entitled to receive in accordance with Section 5.1(c) hereof in consideration for such Funding Backstop Party's Backstop Payment Amount, which Backstopped Shares shall be distributed in the manner described in Section 8.2 hereof.
- (g) The releases and injunctions referred to in Section 11 shall become effective.
- (h) The directors of Jaguar immediately prior to the Implementation Time shall be deemed to have resigned and the New Board shall be deemed to have been appointed.

- (i) The Escrow Agent shall be deemed to be holding the Electing Eligible Investor Funding Amounts and the Backstop Payment Amounts for Jaguar and shall release from escrow such amounts to Jaguar in accordance with the Escrow Agreement.
- (j) Jaguar shall pay: (i) all of the reasonable fees and expenses of the Advisors for services rendered to the Ad Hoc Committee up to and including the Implementation Date, (ii) the reasonable accrued and unpaid third party expenses of any of the Consenting Noteholders up to an amount agreed to by the Majority Backstop Parties; (iii) the fees and expenses of Jaguar's financial advisors in connection with the transactions contemplated under this Plan pursuant to their engagement letter, as amended, with Jaguar, subject to a maximum amount agreed to by the Majority Backstop Parties, (iv) the reasonable fees and expenses of Jaguar's Canadian and U.S. legal advisors and legal advisor to the special committee of the board of directors of Jaguar, and (v) amounts owing to the Trustees under Section 6.07 of the 4.5% Convertible Note Indenture and the 5.5% Convertible Note Indenture

ARTICLE 8 – ISSUANCE AND DISTRIBUTION OF NEW COMMON SHARES

8.1 Issuance of New Common Shares

All New Common Shares issued and outstanding as part of the implementation of this Plan shall be deemed to be issued and outstanding as fully-paid and non-assessable. The amount added to the stated capital of the Common Shares as a result of the issuance of New Common Shares in accordance with this Plan shall be equal to the fair market value of the consideration received by Jaguar for the issuance of such New Common Shares.

8.2 Delivery of New Common Shares

- (a) Jaguar shall use its commercially reasonable best efforts to cause the delivery of the New Common Shares to be distributed under this Plan no later than the second Business Day following the Implementation Date (or such other date as Jaguar and the Majority Consenting Noteholders may agree).
- (b) The Notes are held by DTC (as sole Registered Holder) through its nominee company CEDE & Co. DTC will surrender, or will cause the surrender of, the certificates, if any, representing the Notes to the Trustees in exchange for New Common Shares as contemplated in this Plan.
- (c) The delivery of Unsecured Creditor Common Shares to Noteholders in exchange for the Notes will be made through the facilities of DTC to Participant Holders who, in turn will make delivery of the Unsecured Creditor Common Shares to the Beneficial Noteholders pursuant to standing instructions and customary practices of DTC. If for any reason the New Common Shares are not DTC eligible, then the delivery of the Unsecured Creditor Common Shares shall be made (at the election of Jaguar with the consent of the Monitor and the Majority Consenting Noteholders) either (i) by delivery of a Direct Registration System Advice to each Noteholder or (ii) by delivery of a share certificate to each Noteholder, in either case based on registration instructions received by, or on behalf of, the Monitor from Participant Holders in such manner as the Monitor determines reasonable in the circumstances.
- (d) The delivery of Early Consent Shares to Early Consenting Noteholders will be made (at the election of Jaguar with the consent of the Monitor and the Majority Consenting Noteholders) either: (i) by delivery of a Direct Registration System Advice to each Early Consenting Noteholder; or (ii) by delivery of a share certificate to each Early Consenting

Noteholder, in any case based on registration and delivery instructions contained in the Rep Letter.

- (e) The delivery of Offering Shares, Backstopped Shares, Backstop Commitment Shares and Accrued Interest Offering Shares to the Participating Eligible Investors and the Funding Backstop Parties will be made (at the election of Jaguar with the consent of the Monitor and the Majority Consenting Noteholders) either (i) by delivery of a Direct Registration System Advice to each Participating Eligible Investor and Funding Backstop Party or (ii) by delivery of a share certificate to each Participating Eligible Investor and Funding Backstop Party, in either case based on registration and delivery instructions contained in the Election Forms in the case of Participating Eligible Investors and in the Rep Letter in the case of Funding Backstop Parties.
- (f) The delivery of New Common Shares to Affected Unsecured Creditors (other than Noteholders) in consideration for their Affected Unsecured Claims will be made (at the election of Jaguar with the consent of the Monitor and the Majority Consenting Noteholders) either (i) by delivery of a Direct Registration System Advice to each of the Affected Unsecured Creditors (other than Noteholders) or (ii) by delivery of a share certificate to each of the Affected Unsecured Creditors (other than Noteholders), in either case based on registration and delivery instructions received by the Monitor pursuant to the Claims Procedure Order and the Meeting Order.
- (g) Jaguar, the Monitor and the Trustees will have no liability or obligation in respect of all deliveries from DTC, or its nominee, to Participant Holders or from Participant Holders to Beneficial Noteholders.
- (h) Upon receipt of and in accordance with written instructions from the Monitor, the Trustees shall instruct DTC to, and DTC shall: (i) establish an escrow position representing the respective positions of the Noteholders as of the Implementation Date for the purpose of making distributions to the Noteholders on and after the Implementation Date; and (ii) block any further trading in the Notes, effective as of the close of business on the Distribution Record Date, all in accordance with the customary practices and procedures of DTC.
- Unless a securities law legend is not required by US Securities Laws, the Direct Registration System Advices and share certificates delivered pursuant to this Section 8.2 shall have legends affixed thereon in substantially the form provided for in the Rep Letter.

8.3 Undeliverable Distributions

If any distribution of New Common Shares is undeliverable (that is for greater certainty that cannot be properly registered or delivered to the intended recipient because of inadequate or incorrect registration or delivery information or otherwise) (an **"Undeliverable Distribution**") it shall be delivered to the Escrow Agent, which shall hold such Undeliverable Distribution in escrow, and administered in accordance with this Section 8.3. No further distributions in respect of an Undeliverable Distribution shall be made unless and until the Escrow Agent is notified by the applicable Person of its current address and/or registration information, as applicable, at which time the Escrow Agent shall make such distributions to such Person. All claims for Undeliverable Distributions must be made on or before the date that is the 365th day following the Implementation Date, after which the right to receive distributions under this Plan in respect of such an Undeliverable Distribution shall be fully, finally, irrevocably and forever compromised, released, discharged, cancelled and barred without any compensation therefor, notwithstanding any federal, provincial, or state laws to the contrary, and any New Common Shares that are the subject of such Undeliverable Distribution shall be cancelled.

ARTICLE 9 – RELEASE OF FUNDS FROM ESCROW

9.1 Release of Funds from Escrow

The Escrow Agent shall release any Electing Eligible Investor Funding Amounts and Backstop Payment Amounts, or portions thereof, as follows and in accordance with the terms of the Escrow Agreement:

- (a) If an Electing Eligible Investor deposits less than the full amount of its Electing Eligible Investor Funding Amount by the Electing Eligible Investor Funding Deadline, such party shall cease to be an Electing Eligible Investor and the Escrow Agent shall return such funds so deposited by such Electing Eligible Investor to such Electing Eligible Investor in accordance with Section 4.1(e) hereof.
- (b) On the Implementation Date, the Escrow Agent shall release from escrow to Jaguar, at the applicable time, the applicable Electing Eligible Investor Funding Amounts and Backstop Payment Amounts pursuant to and in accordance with Section 7.4 hereof.
- (c) If this Plan is terminated for any reason or not implemented in accordance with the terms hereof by the Outside Date, the Escrow Agent shall as soon as practicable return all Electing Eligible Investor Funding Amounts and Backstop Payment Amounts to the applicable Participating Eligible Investors and Funding Backstop Parties.
- (d) If any Electing Eligible Investor or Funding Backstop Party provides to the Escrow Agent more than its applicable Electing Eligible Investor Funding Amount or Backstop Payment Amount under this Plan, the Escrow Agent shall as soon as practicable return any excess funds to such Electing Eligible Investor or Funding Backstop Party.

ARTICLE 10 – PROCEDURE FOR DISTRIBUTIONS REGARDING DISPUTED DISTRIBUTION CLAIMS

10.1 No Distribution Pending Allowance

An Affected Unsecured Creditor holding a Disputed Distribution Claim will not be entitled to receive a distribution under the Plan in respect of such Disputed Distribution Claim or any portion thereof unless and until, and then only to the extent that, such Disputed Distribution Claim becomes an Allowed Affected Unsecured Claim.

10.2 Distributions After Disputed Distribution Claims Resolved

(a) <u>Unsecured Creditor Common Shares in relation to a Disputed Distribution Claim of an</u> <u>Affected Unsecured Creditor will be, on or prior to the Implementation Date, either:</u>

(i) issued by the Applicant and held by the Applicant, in a segregated account; or

(ii) authorized by the Applicant's board of directors for issuance by the Applicant.

- (a) Distributions of Unsecured Creditor Common Shares in relation to a Disputed Distribution Claim of an Affected Unsecured Creditor will be held by the Applicant, in a segregated account constitutingwhich in either case shall constitute the Disputed Distribution Claims Reserve, for the benefit of the Affected Unsecured Creditors with Allowed Affected Unsecured Claims until the final determination of the Disputed Distribution Claim in accordance with the Claims Procedure Order and this Plan.
- (b) To the extent that any Disputed Distribution Claim becomes an Allowed Affected Unsecured Claim in accordance with this Plan, the Applicant shall distribute to the holder

106510R: 2926319\2A

of such Allowed Affected Unsecured Claim, that number of Unsecured Creditor Common Shares from the Disputed Distribution Claims Reserve equal to such Affected Unsecured Creditor's Pro Rata Share of Unsecured Creditor Common Shares.

(c) On the date that all Disputed Distribution Claims have been finally resolved in accordance with the Claims Procedure Order and any required distributions contemplated in section (b) have been made, if (i) the aggregate number of Unsecured Creditor Common Shares remaining in the Disputed Distribution Claims Reserve is less than 14,000, the Applicant shall cancel those Unsecured Creditor Common Shares; or (ii) the aggregate number of Unsecured Creditor Common Shares remaining in the Disputed Distribution Claims Reserve is equal to or greater than 14,000, the Applicant shall distribute such Unsecured Creditor Common Shares to the Affected Unsecured Creditors with Allowed Affected Unsecured Claims such that after giving effect to such distributions each such Affected Unsecured Creditor has received its applicable Pro Rata Share of such Unsecured Creditor Common Shares.

ARTICLE 11– RELEASES

11.1 Release

- (a) On the Implementation Date, the following shall be fully, finally, irrevocably and forever compromised, released, discharged, cancelled and barred:
 - (i) all Affected Unsecured Claims;
 - (ii) all Equity Claims;
 - all Director/Officer Claims other than Continuing Other Director/Officer Claims (iii) and Non-Released Director/Officer Claims and also (for greater certainty) excluding Section 5.1(2) Director/Officer Claims and any Agreed Excluded Director/Officer Litigation Claims; provided that any Section 5.1(2) Director/Officer Claims and any Agreed Excluded Director/Officer Litigation Claims shall be limited to recovery from any insurance proceeds payable in respect of such Section 5.1(2) Director/Officer Claims or Agreed Excluded Director/Officer Litigation Claims, as applicable, pursuant to the Director/Officer Insurance Policies, and any Persons with any such Section 5.1(2) Director/Officer Claims or Agreed Excluded Director/Officer Litigation Claims shall have no right to, and shall not, make any claim or seek any recoveries from any Person (including Jaguar, any of its Subsidiaries or any Director or Officer), other than enforcing such Person's rights to be paid from the proceeds of a Director/Officer Insurance Policy by the applicable insurer(s); provided that nothing in this Plan shall prejudice, compromise, release or otherwise affect any right or defence of any insurer in respect of a Director/Officer Insurance Policy or any insured in respect of a Section 5.1(2) Director/Officer Claim or Agreed Excluded Director/Officer Litigation Claim; and
 - (iv) all Director/Officer Indemnity Claims.
- (b) On the Implementation Date, the Applicant, the Subsidiaries, and each of their respective financial advisors, legal counsel and agents, the Monitor, legal counsel to the Monitor, and legal counsel to the special committee of the board of directors of Jaguar (collectively, the "Released Parties") shall be released and discharged from any and all demands, claims, actions, causes of action, counterclaims, suits, debts, sums of money, accounts, covenants, damages, judgments, expenses, executions, liens and other recoveries on account of any indebtedness, liability, obligation, demand or cause of action of whatever

nature that any Person (including any Person who may claim contribution or indemnification against or from them) may be entitled to assert, whether known or unknown, matured or unmatured, direct, indirect or derivative, foreseen or unforeseen, existing or hereafter arising, based in whole or in part on any act or omission, transaction, dealing or other occurrence existing or taking place on or prior to the Implementation Date that are in any way relating to, arising out of or in connection with (i) the restructuring, disclaimer, resiliation, breach or termination of any contract, lease, agreement or other arrangement, whether written or oral; (ii) the business and affairs of the Applicant or the Subsidiaries; (iii) the Notes; (iv) the Indentures; (v) the Existing Shares; (vi) the Existing Share Options; (vii) the Shareholder Rights Plan; (viii) Equity Claims; (ix) the Support Agreement; (x) the Backstop Agreement; (xi) this Plan; or (xii) the CCAA Proceedings; provided, however, that nothing in this Section 11.1 will release or discharge:

- the Applicant or any of the Subsidiaries from or in respect of (x) any Excluded (i) Claim, (y) its obligation to Affected Unsecured Creditors under this Plan or under any Order, or (z) its obligations under the Backstop Agreement or the Support Agreement; provided that any Agreed Excluded Jaguar Litigation Claims shall be limited to recovery from any insurance proceeds payable in respect of such Agreed Excluded Jaguar Litigation Claims pursuant to the Jaguar Insurance Policies, and any Persons with any such Agreed Excluded Jaguar Litigation Claims against the Applicant shall have no right to, and shall not, make any claim or seek any recoveries from any Person (including Jaguar, any of its Subsidiaries or any Director or Officer), other than enforcing such Person's rights to be paid from the proceeds of a Jaguar Insurance Policy by the applicable insurer(s); provided further that nothing in this Plan shall prejudice, compromise, release or otherwise affect any right or defence of any insurer in respect of a Jaguar Insurance Policy or any insured in respect of an Agreed Excluded Jaguar Litigation Claim; or
- (ii) a Released Party if the Released Party is adjudged by the express terms of a judgment rendered on a final determination on the merits to have committed fraud or wilful misconduct.
- At the Implementation Time, each of the Noteholders, the Ad Hoc Committee, the (C) Trustees, and each of their respective present and former shareholders, officers, directors, and the Advisors and the Trustees' counsel (collectively, the "Noteholder Released Parties") will be released and discharged from any and all demands, claims, actions, causes of action, counterclaims, suits, debts, sums of money, accounts, covenants, damages, judgments, expenses, executions, liens and other recoveries on account of any indebtedness, liability, obligation, demand or cause of action of whatever nature that any Person (including any Person who may claim contribution or indemnification against or from them) may be entitled to assert whether known or unknown, matured or unmatured, direct, indirect or derivative, foreseen or unforeseen, existing or hereafter arising, based in whole or in part on any act or omission, transaction, dealing or other occurrence existing or taking place on or prior to the Implementation Date that are in any way relating to, arising out of or in connection with: (i) the Notes; (ii) the Indentures; (iii) the Existing Shares; (iv) the Existing Share Options; (v) the Shareholder Rights Plan; (vi) Equity Claims; (vii) the Support Agreement; (viii) the Backstop Agreement; (ix) this Plan; or (x) the CCAA Proceedings, and any other matters or actions related directly or indirectly to the foregoing; provided that nothing in this Section 11.1(c) will release or discharge a Noteholder Released Party in respect of their obligations under this Plan, the Backstop Agreement, the Support Agreement, any Election Form and provided further that nothing in this Section 11.1(c) will release or discharge a Noteholder Released Party if the Noteholder Released Party is adjudged by the express terms of a

judgment rendered on a final determination on the merits to have committed fraud or wilful misconduct.

11.2 Injunctions

All Persons (regardless of whether or not such Persons are Affected Unsecured Creditors) are permanently and forever barred, estopped, stayed and enjoined, on and after the Implementation Time, with respect to any and all Released Claims or Noteholder Released Claims, from (i) commencing, conducting or continuing in any manner, directly or indirectly, any action, suits, demands or other proceedings of any nature or kind whatsoever (including, without limitation, any proceeding in a judicial, arbitral, administrative or other forum) against the Released Parties, the Named Directors and Officers and the Noteholder Released Parties; (ii) enforcing, levving, attaching, collecting or otherwise recovering or enforcing by any manner or means, directly or indirectly, any judgment, award, decree or order against the Released Parties, the Named Directors and Officers and Noteholder Released Parties or their property; (iii) commencing, conducting or continuing in any manner, directly or indirectly, any action, suits or demands, including without limitation, by way of contribution or indemnity or other relief, in common law, or in equity, or for breach of trust or breach of fiduciary duty or under the provisions of any statute or regulation, or other proceedings of any nature or kind whatsoever (including, without limitation, any proceeding in a judicial, arbitral, administrative or other forum) against any Person who makes such a claim or might reasonably be expected to make such a claim, in any manner or forum, against one or more of the Released Parties, the Named Directors and Officers and Noteholder Released Parties; (iv) creating, perfecting, asserting or otherwise enforcing, directly or indirectly, any lien or encumbrance of any kind against the Released Parties, the Named Directors and Officers and Noteholder Released Parties or their property; or (v) taking any actions to interfere with the implementation or consummation of this Plan; provided, however, that the foregoing shall not apply to the enforcement of any obligations under this Plan. For greater certainty, the provisions of this Section 11.2 shall apply to Section 5.1(2) Director/Officer Claims, Agreed Excluded Director/Officer Litigation Claims and Agreed Excluded Jaguar Litigation Claims in the same manner as Released Claims, except to the extent that the rights of a holder of such Section 5.1(2) Director/Officer Claims, Agreed Excluded Director/Officer Litigation Claims and/or Agreed Excluded Jaguar Litigation Claims to enforce such claims against an insurer in respect of a Directors/Officer Insurance Policy and/or a Jaguar Insurance Policy, as applicable, are expressly preserved pursuant to Section 11.1(a)(iii) and/or Section 11.1(b)(i) hereof.

11.3 Timing of Releases and Injunctions

All releases and injunctions set forth in this Article 11 shall become effective on the Implementation Date at the time or times and in the manner set forth in Section 7.4 hereof.

11.4 Knowledge of Claims

Each Person to which Section 11.1 hereof applies shall be deemed to have granted the releases set forth in Section 11.1 notwithstanding that it may hereafter discover facts in addition to, or different from, those which it now knows or believes to be true, and without regard to the subsequent discovery or existence of such different or additional facts, and such party expressly waives any and all rights that it may have under any applicable law which would limit the effect of such releases to those Claims or causes of action known or suspected to exist at the time of the granting of the release.

ARTICLE 12 – COURT SANCTION, CONDITIONS PRECEDENT AND IMPLEMENTATION

12.1 Application for Sanction Order

If this Plan is approved by the Required Majority, the Applicant shall apply for the Sanction Order on the date set for the hearing for the Sanction Order or such later date as the Court may set.

12.2 Sanction Order

The Sanction Order shall, among other things, declare that:

- (i) this Plan has been approved by the Required Majority entitled to vote at the Meeting in conformity with the CCAA; (ii) the Applicant acted in good faith and has complied with the provisions of the CCAA and the Orders made in the CCAA Proceedings in all respects;
 (iii) the Court is satisfied that the Applicant has not done nor purported to do anything that is not authorized by the CCAA; and (iv) this Plan and the transactions contemplated by it are fair and reasonable;
- (b) this Plan (including the arrangements and releases set out herein) has been sanctioned and approved pursuant to section 6 of the CCAA and will be binding and effective as herein set out on the Applicant, all Affected Unsecured Creditors, all holders of Equity Claims and all other Persons as provided for in this Plan or in the Sanction Order;
- (c) subject to the performance by the Applicant of its obligations under this Plan, and except to the extent expressly contemplated by this Plan or the Sanction Order, all obligations or agreements to which the Applicant is a party immediately prior to the Implementation Time, will be and shall remain in full force and effect as at the Implementation Date, unamended except as they may have been amended by agreement of the parties subsequent to the Filing Date, and no Person who is a party to any such obligations or agreements shall, following the Implementation Date, accelerate, terminate, rescind, refuse to perform or otherwise repudiate its obligations thereunder, or enforce or exercise any right (including any right of set-off, option, dilution or other remedy) or make any demand under or in respect of any such obligation or agreement, by reason of:
 - (i) any defaults or events of default arising as a result of the insolvency of the Applicant prior to the Implementation Date;
 - any change of control of the Applicant arising from implementation of this Plan (except in respect of existing, written senior officer and employee employment agreements of Persons who remain senior officers and employees of Jaguar as of the Implementation Date and any payments due under such agreements, which may only be waived by the senior officers and employees who are parties to such agreements);
 - (iii) the fact that the Applicant has sought or obtained relief under the CCAA or that this Plan has been implemented by the Applicant;
 - (iv) the effect on the Applicant of the completion of any of the transactions contemplated by this Plan;
 - (v) any compromises or arrangements effected pursuant to this Plan; or
 - (vi) any other event(s) which occurred on or prior to the Implementation Date which would have entitled any Person to enforce rights and remedies, subject to any express provisions to the contrary in any agreements entered into with the Applicant after the Filing Date.

For greater certainty, nothing in this paragraph 12.2(c) shall waive, compromise or discharge any obligations of the Applicant in respect of any Excluded Claim;

- 147
- (d) the commencement or prosecution, whether directly, indirectly, derivatively or otherwise, of any demands, claims, actions, counterclaims, suits, judgment, or other remedy or recovery as described in Section 11.2 hereof shall be permanently enjoined;
- the releases effected by this Plan shall be approved, and declared to be binding and effective as of the Implementation Date upon all Affected Unsecured Creditors, holders of Equity Claims and all other Persons affected by this Plan and shall enure to the benefit of all such Persons;
- (f) from and after the Implementation Date, all Persons with an Affected Unsecured Claim shall be deemed to (i) have consented and agreed to all of the provisions of this Plan as an entirety; and (ii) each Affected Unsecured Creditor shall be deemed to have granted, and executed and delivered to the Applicant all consents, releases, assignments and waivers, statutory or otherwise, required to implement and carry out this Plan in its entirety.

12.3 Conditions to Plan Implementation

The implementation of this Plan shall be conditional upon the fulfillment, satisfaction or waiver (to the extent permitted by Section 12.4 hereof) of the following conditions:

- (a) The Court shall have granted the Sanction Order, the operation and effect of which shall not have been stayed, reversed or amended, and in the event of an appeal or application for leave to appeal, final determination shall have been made by the applicable appellate court;
- (b) No Applicable Law shall have been passed and become effective, the effect of which makes the consummation of this Plan illegal or otherwise prohibited;
- (c) All necessary judicial consents and any other necessary or desirable third party consents, if any, to deliver and implement all matters related to this Plan shall have been obtained;
- (d) All documents necessary to give effect to all material provisions of this Plan (including the Sanction Order, this Plan, the Share Offering and the Common Share Consolidation and all documents related thereto) shall have been executed and/or delivered by all relevant Persons in form and substance satisfactory to the Applicant and the Majority Consenting Noteholders;
- (e) All required stakeholder, regulatory and Court approvals, consents, waivers and filings shall have been obtained or made, as applicable, on terms satisfactory to the Majority Consenting Noteholders and the Company, each acting reasonably and in good faith;
- (f) All senior officer and employee employment agreements shall have been modified to reflect the revised capital structure of Jaguar following implementation of the Plan, including, without limitation, to provide that the implementation of the Plan does not constitute a change of control under such employment agreements, and no change of control payments shall be owing or payable to Jaguar's officers or employees in connection with the implementation of the Plan;
- (g) The DSU Plan, the RSU Plan and the SAR Plan shall have been addressed in a manner acceptable to Jaguar and the Majority Consenting Noteholders;
- (h) The Articles of Reorganization shall have been filed under the OBCA;

- (i) All material filings under applicable Laws shall have been made and any material regulatory consents or approvals that are required in connection with this Plan shall have been obtained and, in the case of waiting or suspensory periods, such waiting or suspensory periods shall have expired or been terminated;
- (j) The New Common Shares shall have been conditionally approved for listing on the TSX, the TSXV or such other Designated Offshore Securities Market acceptable to the Majority Consenting Noteholders without any vote or approval of the Existing Shareholders, subject only to receipt of customary final documentation;
- (k) All conditions to implementation of this Plan set out in the Support Agreement (which for greater certainty include the conditions set out in sections 9(a), (b) and (c) of the Support Agreement) shall have been satisfied or waived in accordance with their terms and the Support Agreement shall not have been terminated;
- (I) All conditions to implementation of this Plan set out in the Backstop Agreement (which for greater certainty include the conditions set out in sections 7(a), (b) and (c) of the Backstop Agreement) shall have been satisfied or waived in accordance with their terms, and the Backstop Agreement shall not have been terminated;
- (m) The issuance of the Unsecured Creditor Common Shares and Early Consent Shares shall be exempt from registration under the US Securities Act pursuant to the provisions of section 3(a)(10) of the US Securities Act; and
- (n) No insurer under a Director/Officer Insurance Policy or a Jaguar Insurance Policy shall have an unresolved objection, filed in the CCAA Proceedings, to the implementation of this Plan.

12.4 Waiver of Conditions

The Applicant and the Majority Consenting Noteholders may at any time and from time to time waive the fulfillment or satisfaction, in whole or in part, of the conditions set out herein, to the extent and on such terms as such parties may agree to provided however that the conditions set out in Section 12.3(a) cannot be waived and that the conditions set out in Section 12.3(l) can only be waived by the Applicant and the Majority Backstop Parties.

12.5 Implementation Provisions

If the conditions contained in Section 12.3 are not satisfied or waived (to the extent permitted under Section 12.4) by the Outside Date, unless the Applicant and the Majority Consenting Noteholders agree in writing to extend such period, this Plan and the Sanction Order shall cease to have any further force or effect and will not be binding on any Person.

12.6 Monitor's Certificate of Plan Implementation

Upon written notice from the Applicant (or counsel on its behalf) and Goodmans LLP on behalf of the Majority Consenting Noteholders and the Majority Backstop Parties to the Monitor that the conditions to Plan implementation set out in Section 12.3, have been satisfied or waived, the Monitor shall, as soon as possible following receipt of such written notice, deliver to the Applicant and Goodmans LLP on behalf of the Majority Consenting Noteholders and the Majority Backstop Parties, and file with the Court, a certificate which states that all conditions precedent set out in Section 12.3 have been satisfied or waived and that the Implementation Date has occurred.

ARTICLE 13 – GENERAL

13.1 Waiver of Defaults

- Subject to the performance by the Applicant of its obligations under this Plan, and except to the extent expressly contemplated by this Plan or the Sanction Order, no Person who is a party to any obligations or agreements with the Applicant or any Subsidiary shall, following the Implementation Date, accelerate, terminate, rescind, refuse to perform or otherwise repudiate its obligations thereunder, or enforce or exercise any right (including any right of set-off, option, dilution or other remedy) or make any demand under or in respect of any such obligation or agreement, by reason of:
 - (a) any defaults or events of default arising as a result of the insolvency of the Applicant prior to the Implementation Date;
 - (b) any change of control of the Applicant or any Subsidiary arising from implementation of this Plan (except in respect of existing, written senior officer and employee employment agreements of Persons who remain senior officers and employees of Jaguar as of the Implementation Date and any payments due under such agreements, which may only be waived by the senior officers and employees who are parties to such agreements);
 - (c) the fact that the Applicant has sought or obtained relief under the CCAA or that this Plan has been implemented by the Applicant;
 - (d) the effect on the Applicant or any Subsidiary of the completion of any of the transactions contemplated by this Plan;
 - (e) any compromises or arrangements effected pursuant to this Plan; or
 - (f) any other event(s) which occurred on or prior to the Implementation Date which would have entitled any Person to enforce rights and remedies, subject to any express provisions to the contrary in any agreements entered into with the Applicant after the Filing Date.
 - For greater certainty, nothing in this paragraph 13.1 shall waive, compromise or discharge any obligations of the Applicant in respect of any Excluded Claim.

13.2 Deeming Provisions

In this Plan, the deeming provisions are not rebuttable and are conclusive and irrevocable.

13.3 Non-Consummation

The Applicant reserves the right to revoke or withdraw this Plan at any time prior to the Implementation Date, with the consent of the Monitor and the Majority Consenting Noteholders.

If the Implementation Date does not occur on or before the Outside Date (as the same may be extended in accordance with the terms hereof and of the Support Agreement), or if this Plan is otherwise withdrawn in accordance with its terms: (a) this Plan shall be null and void in all respects, and (b) nothing contained in this Plan, and no acts taken in preparation for consummation of this Plan, shall (i) constitute or be deemed to constitute a waiver or release of any Claims by or against the Jaguar Group, their respective successors or any other Person; (ii) prejudice in any manner the rights of the Jaguar Group, their respective successors; or (iii) constitute an admission of any sort by the Jaguar Group, their respective successors or any other Person.

13.4 Modification of Plan

- (a) The Applicant may, at any time and from time to time, amend, restate, modify and/or supplement this Plan with the consent of the Monitor and the Majority Consenting Noteholders, provided that: any such amendment, restatement, modification or supplement must be contained in a written document that is filed with the Court and:
 - (i) if made prior to or at the Meeting: (A) the Monitor, the Applicant or the Chair (as defined in the Meeting Order) shall communicate the details of any such amendment, restatement, modification and/or supplement to Affected Unsecured Creditors and other Persons present at the Meeting prior to any vote being taken at the Meeting; (B) the Applicant shall provide notice to the service list of any such amendment, restatement, modification and/or supplement and shall file a copy thereof with the Court forthwith and in any event prior to the Court hearing in respect of the Sanction Order; and (C) the Monitor shall post an electronic copy of such amendment, restatement, modification and/or supplement on the Monitor's Website forthwith and in any event prior to the Court hearing in respect of the Sanction Order; and
 - (ii) if made following the Meeting: (A) the Applicant shall provide notice to the service list of any such amendment, restatement, modification and/or supplement and shall file a copy thereof with the Court; (B) the Monitor shall post an electronic copy of such amendment, restatement, modification and/or supplement on the Monitor's Website; and (C) such amendment, restatement, modification and/or supplement shall require the approval of the Court following notice to the Affected Unsecured Creditors.
- (b) Notwithstanding Section 13.4(a) hereof, any amendment, restatement, modification or supplement may be made by the Applicant: (i) if prior to the date of the Sanction Order, with the consent of the Monitor and the Majority Consenting Noteholders; and (ii) if after the date of the Sanction Order, with the consent of the Monitor and the Majority Consenting Noteholders and upon approval by the Court, provided in each case that it concerns a matter that, in the opinion of the Applicant, acting reasonably, is of an administrative nature required to better give effect to the implementation of this Plan and the Sanction Order or to cure any errors, omissions or ambiguities and is not materially adverse to the financial or economic interests of the Affected Unsecured Creditors.
- (c) Any amended, restated, modified or supplementary plan or plans of compromise filed with the Court and, if required by this section, approved by the Court, shall, for all purposes, be and be deemed to be a part of and incorporated in this Plan.

13.5 Severability of Plan Provisions

If, prior to the Implementation Time, any term or provision of this Plan is held by the Court to be invalid, void or unenforceable, at the request of the Applicant, made with the consent of the Majority Consenting Noteholders (acting reasonably), the Court shall have the power to either (a) sever such term or provision from the balance of this Plan and provide the Applicant and the Majority Consenting Noteholders with the implementation of the balance of this Plan as of and with effect from the Implementation Time, or (b) alter and interpret such term or provision to make it valid or enforceable to the maximum extent practicable, consistent with the original purpose of the term or provision held to be invalid, void or unenforceable, and such term or provision shall then be applicable as altered or interpretation, acting reasonably. Notwithstanding any such holding, alteration or interpretation, and provided that this Plan is implemented, the remainder of the terms and provisions of this Plan shall remain in full force and effect and shall in no way be affected, impaired or invalidated by such holding, alteration or interpretation.

13.6 Preservation of Rights of Action

Except as otherwise provided in this Plan or in the Sanction Order, or in any contract, instrument, release, indenture or other agreement entered into in connection with this Plan, following the Implementation Date, the Applicant will retain and may enforce, sue on, settle, or compromise (or decline to do any of the foregoing) all claims, rights or causes of action, suits and proceedings, whether in law or in equity, whether known or unknown, that the Applicant may hold against any Person or entity without further approval of the Court.

13.7 Responsibilities of Monitor

FTI Consulting Canada Inc. is acting in its capacity as Monitor in the CCAA Proceedings with respect to the Applicant and this Plan and not in its personal or corporate capacity, and will not be responsible or liable for any obligations of the Applicant under the Plan or otherwise.

13.8 Notices

Any notice or other communication to be delivered hereunder must be in writing and refer to this Plan and may, as hereinafter provided, be made or given by personal delivery, ordinary mail, email or by facsimile addressed to the respective Parties as follows:

If to Jaguar or the Subsidiaries, at:

 (a) c/o Jaguar Mining Inc.
 67 Yonge Street, Suite 1203 Toronto, Ontario M5E 1J8

> Attention: David Petroff Email: david.petroff@jaguarmining.com

with a required copy (which shall not be deemed notice) to:

Norton Rose Fulbright Canada LLP Royal Bank Plaza, South Tower, Suite 3800 200 Bay Street P.O. Box 84 Toronto, Ontario M5J 2Z4

Attention:Walied Soliman and Evan CobbFax:(416) 216-3930Email:walied.soliman@nortonrosefulbright.comevan.cobb@nortonrosefulbright.com

(b) If to the Ad Hoc Committee of Noteholders:

Goodmans LLP Suite 3400 333 Bay Street Bay Adelaide Centre Toronto, Ontario M5H 2S7

 Attention:
 Rob Chadwick and Melaney Wagner

 Fax:
 (416) 979-1234

 Email
 rchadwick@goodmans.ca mwagner@goodmans.ca

(c) If to the Monitor, at:

FTI Consulting Canada Inc. TD Waterhouse Tower Suite 2010 79 Wellington Street Toronto, Ontario M5K 1G8

Attention:	Greg Watson and Jodi Porepa
Fax:	(416) 649-8101
Email:	Greg.Watson@fticonsulting.com
	Jodi.Porepa@fticonsulting.com

With a required copy (which shall not be deemed notice) to:

Osler, Hoskin & Harcourt LLP Box 50 1 First Canadian Place Toronto, Ontario M5X 1B8

Attention:Marc WassermanFax:(416) 862-6666Email:mwasserman@osler.com

or to such other address as any Party may from time to time notify the others in accordance with this Section 13.8. Any such communication so given or made shall be deemed to have been given or made and to have been received on the day of delivery if delivered, or on the day of faxing or emailing, provided that such day in either event is a Business Day and the communication is so delivered, faxed or emailed before 5:00 p.m. on such day. Otherwise, such communication shall be deemed to have been given and made and to have been received on the next following Business Day.

13.9 Consent of Majority Consenting Noteholders or Majority Backstop Parties

For the purposes of this Plan, any matter requiring the agreement, waiver, consent or approval of the Majority Consenting Noteholders or the Majority Backstop Parties shall be deemed to have been agreed to, waived, consented to or approved by such Majority Consenting Noteholders or Majority Backstop Parties if such matter is agreed to, waived, consented to or approved in writing by Goodmans LLP, provided that Goodmans LLP expressly confirms in writing (which can be by way of e-mail) that it is providing such agreement, consent, waiver or approval on behalf of the Majority Consenting Noteholders or the Majority Backstop Parties, as applicable.

13.10 Paramountcy

From and after the Implementation Time on the Implementation Date, any conflict between:

- (a) this Plan; and
- (b) the covenants, warranties, representations, terms, conditions, provisions or obligations, expressed or implied, of any contract, mortgage, security agreement, indenture, trust indenture, note, loan agreement, commitment letter, agreement for sale, lease or other agreement, written or oral and any and all amendments or supplements thereto existing between any Person and the Applicant and/or the Subsidiaries as at the Implementation Date,
- will be deemed to be governed by the terms, conditions and provisions of this Plan and the Sanction Order, which shall take precedence and priority.

13.11 Further Assurances

Notwithstanding that the transactions and events set out herein will occur and be deemed to occur in the order set out in this Plan without any further act or formality, each of the Persons named or referred to in, or subject to, this Plan will make, do and execute, or cause to be made, done and executed, all such further acts, deeds, agreements, transfers, assurances, instruments or documents as may reasonably be required by any of them to carry out the full intent and meaning of this Plan and to give effect to the transactions contemplated herein.

Schedule "A"

Agreed Excluded Director/Officer Litigation Claims

(i) Claims asserted by Daniel R. Titcomb; Robert J. Lloyd; James M. Roller; William E. Dow; Jeffrey Kirchhoff and/or Brazilian Resources, Inc. (collectively, the "New Hampshire Litigation Plaintiffs") in the proceeding in the United States District Court for the District of New Hampshire bearing Clvil Action No. 1:13-cv-00428-JL against Gary E. German, Gilmour Clausen, John Andrews, Richard Falconer, David Petroff and Frederick Hermann (collectively, the "Director/Officer Defendants"); and

(ii) those claims of the New Hampshire Litigation Plaintiffs against the Director/Officer Defendants as allowed in accordance with and subject to the terms of paragraph 3 of the Minutes of Settlement dated February 5, 2014 between the New Hampshire Litigation Plaintiffs and Jaguar Mining Inc.

Schedule "B"

and the second state of th

na variationa anti-

Agreed Excluded Jaguar Litigation Claims

(i) Claims asserted by Daniel R. Titcomb; Robert J. Lloyd; James M. Roller; William E. Dow; Jeffrey Kirchhoff and/or Brazilian Resources, Inc. (collectively, the "New Hampshire Litigation Plaintiffs") in the proceeding in the United States District Court for the District of New Hampshire bearing Civil Action No. 1:13-cv-00428-JL against Jaguar Mining Inc.; and

(ii) those claims of the New Hampshire Litigation Plaintiffs against Jaguar Mining Inc. as allowed in accordance with and subject to the terms of paragraph 4 of the Minutes of Settlement dated February 5, 2014 between the New Hampshire Litigation Plaintiffs and Jaguar Mining Inc.

... 집안 그는 것 같아요. 한 것

Sector Co

62351863.

10.00

Schedule "C"

NA SEE E CARA

Agreed Excluded Litigation Claimants

Daniel R. Titcomb

Robert J. Lloyd

James M. Roller

William E. Dow

Jeffrey Kirchhoff

Brazilian Resources, Inc.

that bases

Stepping L

(Astronical) (States and Added)

IN THE MATTER OF THE COMPANIES' CREDITORS ARRANGEMENT ACT, R.S.C. 1985, c. C-36, AS AMENDED	Court File No: CV-13-10383-00CL
AND IN THE MATTER OF A PLAN OF COMPROMISE OR ARRANGEMENT OF JAGUAR MINING INC. (Applicant)	
	ONTARIO SUPERIOR COURT OF JUSTICE
	Proceeding commenced at Toronto
	AFFIDAVIT OF T. DOUGLAS WILLOCK (Sworn February 27, 2014)
	Norton Rose Fulbright Canada LLP Royal Bank Plaza, South Tower, Suite 3800 200 Bay Street, P.O. Box 84 Toronto, Ontario M5J 2Z4 CANADA Tony Reyes LSUC#: 28218V Tel: 416.216.4825 Email: <u>tony.reyes@nortonrosefulbright.com</u>
	Evan Cobb LSUC#: 55787N Tel: 416.216.1929 Email: <u>evan.cobb@nortonrosefulbright.com</u>
	Fax: 416.216.3930
	Lawyers for the Applicant, Jaguar Mining Inc.
	156 P

DOCSTOR: 2944324/1B

A



Court File No. CV-13-10383-00CL

ONTARIO SUPERIOR COURT OF JUSTICE COMMERCIAL LIST

))

)

THE HONOURABLE REGIONAL SENIOR

JUSTICE MORAWETZ

THURSDAY, THE 27TH

DAY OF FEBRUARY, 2014

IN THE MATTER OF THE COMPANIES' CREDITORS ARRANGEMENT ACT, R.S.C. 1985, c. C-36, AS AMENDED

AND IN THE MATTER OF A PLAN OF COMPROMISE OR ARRANGEMENT OF JAGUAR MINING INC.

Applicant

ORDER (Stay Extension)

THIS MOTION, made by the Applicant for an order, inter alia:

- (a) that the time for service of the Notice of Motion, Motion Record and the Fourth Report of the Monitor, FTI Consulting Canada Inc., (the "Monitor") dated February 27, 2014 (the "Fourth Report") is abridged and validated so that this Motion is properly returnable today and dispensing with further service thereof; and
- (b) approving an extension of the Stay Period, as defined in paragraph 14 of the Initial Order, issued by this Court on December 23, 2013, (as amended, the "Initial Order") to and including March 10, 2014; and
- (c) approving a certain amendments to the Applicant's Amended and Restated Plan of Compromise and Arrangement, dated February 5, 2014 (the "**Plan**"),

was heard this day at 330 University Avenue, Toronto, Ontario.

ON READING the Affidavit of T. Douglas Willock, sworn February 27, 2014, the exhibits thereto, (the **"Willock Affidavit**") and the Fourth Report, and on hearing submissions of counsel for the Applicant, the Monitor, the Ad Hoc Committee (as defined in the Willock Affidavit), Global Resource Fund, and such other counsel present, no one else appearing although duly served as appears from the affidavit of service of Evan Cobb sworn on February 27, 2014,

Service

1. THIS COURT ORDERS that the time for service of the Notice of Motion, Motion Record and Fourth Report is hereby abridged and validated so that this Motion is properly returnable today and hereby dispenses with further service thereof.

Stay Extension

2. THIS COURT ORDERS that the Stay Period, as defined in paragraph 14 of the Initial Order, be and is hereby extended up to and including March 10, 2014.

Plan Amendment

3. THIS COURT ORDERS that the Amendments (as defined in the Willock Affidavit) are hereby approved in accordance with Section 13.4(b) of the Plan.

Court File No: CV-13-10383-00CL	ONTARIO SUPERIOR COURT OF JUSTICE COMMERCIAL LIST Proceeding commenced at Toronto	ORDER	Norton Rose Fulbright Canada LLP Royal Bank Plaza, South Tower, Suite 3800 200 Bay Street, P.O. Box 84 Toronto, Ontario M5J 224 CANADA Toronto, Ontario M5J 224 CANADA Tony Reyes LSUC#: 28218V Tel: 416.216.4825 Email: <u>tony.reyes@nortonrosefulbright.com</u> Fail: <u>evan Cobb LSUC#: 55787N</u> Tel: 416.216.1929 Email: <u>evan.cobb@nortonrosefulbright.com</u> Fax: 416.216.3930 Lawyers for the Applicant, Jaguar Mining Inc.	9
IN THE MATTER OF THE <i>COMPANIES' CREDITORS ARRANGEMENT ACT</i> , R.S.C. 1985, c. C-36, AS AMENDED AND IN THE MATTER OF A PLAN OF COMPROMISE OR ARRANGEMENT OF JAGUAR MINING INC. (Applicant)			DOCSTOR: 29448732	

1.7814

100

(Weissel March 2012

나라 관람을 알

1.1

, t<u>erinanin'ny 6-56556555566666</u>6

<u>- 1996</u>

Court File No: CV-13-10383-00CL	ONTARIO SUPERIOR COURT OF JUSTICE COMMERCIAL LIST Proceeding commenced at Toronto	MOTION RECORD	Norton Rose Fulbright Canada LLP Royal Bank Plaza, South Tower, Suite 3800 200 Bay Street, P.O. Box 84 Toronto, Ontario M5J 2Z4 CANADA Toronto, Ontario M5J 2Z4 CANADA Tory Reyes LS#: 28218V Tel: 416.216.4825 Email: tony.reyes@nortonrosefulbright.com Fer 416.216.1929 Evan Cobb LS#: 55787N Tel: 416.216.1929 Email: evan.cobb@nortonrosefulbright.com Fax: 416.216.3930 Fax: 416.216.3930 Lawyers for the Applicant, Jaguar Mining Inc.
IN THE MATTER OF THE COMPANIES' CREDITORS ARRANGEMENT ACT, R.S.C. 1985, c. C-36, AS AMENDED AND IN THE MATTER OF A PLAN OF COMPROMISE OR ARRANGEMENT OF JAGUAR MINING INC. Applicant			